The Formation of Nominal Derivatives in the Arabic Language

With a View to Computational Linguistics

Abdallah Tayseer Alshdaifat

Ph.D. Thesis

2014

The Formation of Nominal Derivatives in the Arabic Language With a View to Computational Linguistics

Abdallah Tayseer Alshdaifat

School of Humanities, Languages and Social Sciences University of Salford, Salford, UK

Submitted in Partial Fulfilment of the Requirements of the Degree of Doctor of Philosophy, October 2014

The	table	of	contents
-----	-------	----	----------

List of tables	vii
List of figures	ix
Acknowledgements	X
Declaration	xii
Chapter One	
1.1 Introduction	1
1.2 The Arabic language and its linguistic tradition	1
1.3 Research problem	5
1.4 Research questions	7
1.5 Research motivation and contribution	
1.6 Transliteration	9
1.7 Thesis structure	
Chapter Two	
2.1 Introduction	
2.2 Morphology	
2.3 Nonconcatenative morphology	
2.4 Autosegmental phonology	
2.5 Prosodic phonology	
2.6 Theoretical approaches to Arabic morphology	
2.6.1 The root-based approach	
2.6.2 The stem-based approach	
2.7 Verb classifications	
2.7.1 The philosophical classification	
2.7.2 The semantic classification	
2.7.3 The syntactic classification	
2.8 The literature of computational linguistics on Arabic morphology	
Chapter Three	
3.1 Introduction	
3.2 Background to the study	
3.3 Data collection	
3.4 Our classification of Arabic verbs	
3.5 Multi-level analysis of the nominal derivatives	
3.6 Computational application for the nominal derivatives	
Chapter Four	

4.1 Introduction	46
4.2 Arabic verb classes	48
4.2.1 Verbs of the state of the body (class 01)	48
4.2.2 Verbs of body parts (Class 02)	49
4.2.3 Verbs of diseases (Class 03)	50
4.2.4 Verbs of social and personal behaviour (Class 04)	51
4.2.5: Verbs of emotions (class 05)	54
4.2.6 Verbs of colouring (class 06)	57
4.2.7 Verbs of bodily qualities (class 07)	58
4.2.8 Verbs of motion (class 08)	59
4.2.9 Verbs of swimming (class 09)	61
4.2.10 Verbs of location/place (class 10)	62
4.2.11 Verbs of violence and abuse (class 11)	63
4.2.12 Verbs of mental process (class 12)	65
4.2.13 Verbs of financial transaction (class 13)	66
4.2.14 Verbs of agriculture (class 14)	69
4.2.15 Verbs of desire and request (class 15)	71
4.2.16 Verbs of intention (class 16)	72
4.2.17 Verbs of combining and constructing (class 17)	73
4.2.18 Verbs of sending and carrying (class 18)	75
4.2.19 Verbs of separating and disassembling (class 19)	76
4.2.20 Verbs of removing (class 20)	78
4.2.21 Verbs of bending (class 21)	79
4.2.22 Verbs of decorating and transcribing (class 22)	80
4.2.23 Verbs of measurement (class 23)	81
4.2.24 Verbs of quantity and size (class 24)	81
4.2.25 Verbs of stability (class 25)	82
4.2.26 Verbs of creation (class 26)	84
4.2.27 Verbs of preparing (class 27)	85
4.2.28 Verbs of ingesting (class 28)	86
4.2.29 Verbs of the five senses (class 29)	87
4.2.30 Verbs of ruling and government (Class 30)	88
4.2.30 Verbs of the development of life (class 31)	89
4.2.32 Verbs of uttering (class 32)	90
4.2.33: Verbs of accepting (class 33)	91

4.2.34 Verbs of refusing and disobedience (class 34)	92
4.2.35 Verbs of preventing and prohibition (class 35)	93
4.2.36 Verbs of occurrence and progressing (class 36)	95
4.2.37 Verbs of appearance (class 37)	96
4.2.38 Verbs of disappearing and ending (class 38)	97
4.2.39 Verbs of bodily movements (class 39)	98
4.2.40 Verbs of taking and giving (class 40)	99
4.2.41 Verbs of human sounds (class 41)	101
4.2.42 Verbs of sounds made by animals (class 42)	102
4.2.43 Verbs of bodily care (class 43)	103
4.2.44 Verbs of winning (class 44)	104
Chapter Five	106
5.1 Introduction	106
5.2 Morphological and prosodic analysis of the nominal derivatives	106
5.2.1 Morphological and prosodic analysis of the active participle	106
5.2.2 Morphological and prosodic analysis of the passive participle	109
5.2.3 Morphological and prosodic analysis of the form of exaggeration	112
5.2.4 Morphological and prosodic analysis of the instrumental noun	117
5.2.5 Morphological and prosodic analysis of the qualificative adjective	123
5.2.6 Morphological and prosodic analysis of the locative noun	129
5.3 Semantic analysis of the nominal derivatives	132
5.3.1 Semantic features of the active participle	133
5.3.2 Semantic features of the passive participle	135
5.3.3 Semantic features of the form of exaggeration	137
5.3.4 Semantic features of the Instrumental Noun	139
5.3.5 Semantic features of the qualificative adjective	140
5.3.6 Semantic features of the locative noun	142
5.4 Semantic features of the verb classes and their compatibility with the nominal derivatives .	144
Chapter Six	214
6.1 Introduction	214
6.2 Computational morphology	214
6.2.1 Morphological analysis	217
6.2.2 Morphological generation	218
6.2.2.1 Undergeneration problem	219
6.2.2.2 Overgeneration problem	220

6.3 Overgeneration in the formation of nominal derivatives	
6.4 Our solution to the overgeneration problem of nominal derivatives	
6.5 The computational application of nominal derivatives	
6.6 Evaluation	
Conclusion	
Recommendations	
References	

List of tables

Table 1: Arabic characters and transliteration symbols	9
Table 2: Form I active participle	14
Table 3: The ten derived verb forms in Arabic	15
Table 4: Examples of various vocalic melodies applied to the root k-t-b	22
Table 5: English examples of Kenny's verb classification	29
Table 6: Aspectual features of Vendler's classes	
Table 7: Definitions of nominal derivatives	
Table 8: The verb Forms and their active and passive participles	41
Table 9: The semantic features of the object entities and the situation entities	42
Table 10: Arabic verb classes and their Levin and Vendler's equivalents	46
Table 11: Form I verbs and their participles	107
Table 12: The different verb types with their active participles	107
Table 13: Passive participles derived from Form I verbs	110
Table 14: Passive participles derived from different verb types	110
Table 15: Patterns of the form of exaggeration	112
Table 16: The different verb types with their forms of exaggeration	112
Table 17: Form I verbs and their instrumental noun patterns	117
Table 18: The different verb types with their instrumental nouns	118
Table 19: The different verb types with their qualificative adjectives	124
Table 20: Examples of locative nouns and their input verbs	129
Table 21: The different verb types with their locative nouns	130
Table 22: Semantic features of active participle	134
Table 23: The semantic features of the passive participle	136
Table 24: The semantic features of the verb of exaggeration	138
Table 25: The semantic features of the instrumental noun	139
Table 26: The semantic features of the qualitative adjective	141
Table 27: The semantic features of the locative noun	143

Table 28: The output data of the computational application	.225
Table 29: The evaluation data	.229
Table 30: Confusion matrix elements	.229
Table 31: The values of the confusion matrix elements	.230
Table 32: The nominal derivatives of the verbal forms	.236
Table 33: Form I stem templates	.238

List of figures

Figure 1: Helbig's ontology of entities	32
Figure 2: Semantic feature compatability between the nominal derivative and the verb classes	. 132
Figure 3: The semantic feature compatability of deriving the active participle	. 135
Figure 4: The semantic feature compatability of deriving the passive participle	. 137
Figure 5: The semantic feature compatability of deriving the form of exaggeration	. 139
Figure 6: The semantic feature compatability of the instrumental noun	. 140
Figure 7: The semantic feature compatability of the qualificative adjective	. 142
Figure 8: The semantic feature compatability of deriving the locative noun	. 143
Figure 9: The interface of the computational application	. 226
Figure 10: Two stages of derivation	
Figure 11: Nominal derivatives compatibilities	
Figure 12: The numbers of valid nominal derivatives	

Acknowledgements

First and foremost I would like to thank my supervisor Janet Watson for her insightful feedback and valuable methodological advice that helped me accomplish this thesis. Being her student has been an invaluable and enriching experience. I have gained from her a lot of research knowledge at every stage of the study. In addition, I am grateful to my second supervisor Domenyk Eades for his help and guidance.

Special thanks to University of Jordan for awarding me a full scholarship.

I would like to express my gratitude to Majdi Sawalha, for the valuable knowledge in computational morphology he has shared with me.

A big thank you to Abdullah Alfaifi, for his generous assistance in designing the computational application.

I would like to thank my dear friends Anas Alhunety and Basil Almashaqba for their help and encouragement.

Dedication

I dedicate this thesis with love to my dear parents.

Declaration

I declare that this thesis was the result of my own work. No portion of the work covered in this thesis has been submitted in support of any application for another degree or qualification at this or any other university or institution of higher learning.

Abstract

This study investigates the formation of nominal derivatives in Arabic by providing a multi-level analysis in the light of state-of-the-art theories and approaches in modern linguistics. Six types of nominal derivative are described and analyzed: the active participle, the passive participle, the form of exaggeration, the instrumental noun, the qualificative adjective, and the locative noun. The study considers the Form I verb stem as input for forming the six types of nominal derivative (the output). A multi-level approach is employed, involving semantics, syntax, morphology and prosodic phonology.

The study establishes an Arabic verb classification in which 980 Form I verbs are divided into 44 classes from which nominal derivatives are derived. Verbs are allocated to classes according to their semantic features and syntactic behaviour. Semantically, the verbs in each class share related meanings and semantic functions. In addition, semantic relations such as synonyms, antonyms, polysemy and hyponym are taken into consideration. Syntactically, the verbs in each class share syntactic behaviour in terms of their transitivity and syntactic frames (in which all the verbs of a class can be used alternatively).

Morphologically, the word formation of nominal derivatives is analyzed and described using the stem-based approach where the input stem is a Form I verb and the output stem is a nominal derivative. Prosodically, the analysis describes the word formation processes and prosodic structures of the nominal derivatives.

The valid nominal derivatives are derived from each of the 44 verb classes based on the compatibility between their semantic features and the semantic features of each class. The derivation of an invalid nominal derivative causes an overgeneration problem that is involved in computational linguistic applications. This study has determined that the overgeneration problem is found only in three types of nominal derivative: the instrumental noun, the qualificative adjective, and the locative noun, where there is no compatibility (agreement) between the semantic features of a given verb and the semantic features of a nominal derivative. Only four or five valid nominal derivatives can be derived from a given verb, not all six types.

Chapter One

Introduction

1.1 Introduction

This chapter begins with a description of the Arabic linguistic tradition, from its place within Semitic language family, to its development and rapid spread, its status over the centuries, the traditional grammatical schools and prominent scholars within this tradition. The chapter then turns to the research topic, discussing the research problem, research questions, research motivation and contribution, and thesis structure.

1.2 The Arabic language and its linguistic tradition

The Arabic language belongs to the Semitic language family. All languages within this family descend from Proto-Semitic. The family has two main branches: the east Semitic languages that include the extinct languages, Eblaite and Akkadian; and western Semitic languages, divided into southern Semitic – Ethio-Semitic and Modern South Arabian – and central Semitic – Arabic, Aramaic, Canaanite, Phoenician, Hebrew and Ugaritic. Arabic is the most widespread of the living Semitic languages (Hetzron 1992: 412-413; cf. Watson 2002: 1).

At the end of the sixth century CE, a rapid development of Arabic occurred through the emergence of Islam, when Arabic became the official language of the new Islamic state. As in other religions, the prophets are associated with miracles, in this case the miracle of Mohammad, the prophet of Islam, and the Qur'an that is the sacred book of Islam. In Islamic belief, the Qur'an is the speech of Allah, conveyed through the Archangel Gabriel to the Prophet Mohammad. Muslims consider the Qur'an to be a literary and linguistic miracle.

Within a hundred years, the Islamic state had spread outside the Arabian Peninsula to reach the Levant in the north, Iraq, Khuzistan and Persia in the east, and North Africa and Spain in the west. Native Arabs from the Arabian Peninsula then settled in the conquered countries. Over subsequent centuries there was further expansion of the borders of the state, through three continents: in Europe, Spain and Portugal (Andalusia); in Africa, the central African countries; and in Asia, Turkestan, Pakistan, Indonesia, Malaysia, and the west of China. The non-Arab Muslims seized on learning the Arabic language for their prayers and religious rituals.

According to Watson (2002: 6) "the rise and expansion of Islam was not only a religious and hence cultural conquest, but also a linguistic conquest, and within a few hundred years Arabic became both the official and the vernacular language of all Islamicized countries". However, with the demise of the Islamic Ottoman Empire in 1916, several nations regained their original national tongue, such as the Turks, Kurds, Berbers, and Persians.

Currently, Arabic is one of the six official languages of the United Nations; it is also the official national language of twenty-two countries in the Middle East and North Africa, from the Arabian Gulf countries in the east to North African countries in the west. At the same time, Arabic is the liturgical and religious language of approximately one billion Muslims around the world (Chon & Arzt 2005: 246).

Arabic grammar has a long-standing linguistic tradition. Since the eighth century CE, Arabic has received continuous attention from researchers. Medieval Arab grammarians classify Arabic grammar into two main categories: *şarf* and *naḥw*, which are the closest Arabic terms to morphology and syntax respectively. However, the boundaries between them are distinct from their counterparts in Western linguistics. In Arabic, the category of *şarf* (morphology) covers many aspects of derivational morphology (e.g. the forms of the verb, verbal and nominal derivatives) as well as inflectional morphology (e.g. tense/aspect paradigms, number marking in nouns, number-gender marking in adjectives), but does not encompass grammatical case and mood which are considered under *naḥw* (syntax) (Ryding 2005).

The first linguistic treatises on Arabic to appear were written by two influential scholars: Al-Khalīl ibn Ahmad (d. 791 CE) and Sibawaih (d. 793 CE). Al-Khalīl ibn Ahmad produced the first Arabic dictionary *Mu^c jam Al-cayn* (1967) (ed. by Abdallah Darwīsh), while Sibawaih (1966) (ed. by Abd al-Salām Hārūn) is considered to be the father of Arabic grammar; his work *Al-Kitāb* ('The Book') provides a comprehensive descriptive analysis of Arabic. Sibawaih deals mainly with syntactic issues, but also discusses the structure of Arabic words through the description of morphological and phonological structures which determine the surface form of the word.

Thereafter, research activities in the discipline of Arabic grammar appeared in the form of grammatical schools. Al-Başrah and al-Kūfah were the most famous schools of medieval Arabic grammatical analysis (Daif 1983). Some of these scholars have enriched the literature of Arabic morphology by devoting portions of their works to describing the morphological aspects (Az-Zajjāji 1984; Al-Jurjāni 1984; Al-Istarabādi 1983). Al-Māzini (d. 863) was the first Arab grammarian to devote a whole work, *Kitab At-Taṣrīf*, to morphological issues. The grammarian Ibn Jinnī (d. 1002) commented on Al-Māzini's work in his book *Al-Munṣif* (1960), which is considered to be one of the most valuable books on Arabic morphology.

The rapid expansion of the Arab-Islamic Empire during the seventh and eighth centuries CE meant that by the Abbasid era classical Arab linguistic study had reached maturity. Although the Arabs of the Arabian Peninsula were a minority in this sprawling multiethnic and multi-lingual civilization, the supremacy of Arabian tribes in governance and administration and the central position of the Qur'an in Muslim liturgy meant that Arabic became the lingua franca. The incorporation of non-Arabs into the cultural life of the times aroused significant interest in the study of Arabic linguistics, particularly on the part of the Persians (cf. Daif 1983). Thus, medieval grammarians were encouraged to provide their representation of the internal structure of Arabic words.

Furthermore, the Holy Qur'an is believed to be a literary miracle in Islam, and therefore the linguistic mistakes made by non-Arabs who converted to Islam (such as incorrect assignment of case and mood markings) were criticised by native Arabic speakers. Sibawaih was born c. 750 in Persia. His father converted to Islam and became a client of the Arab tribe of the Banī l-Hāri<u>t</u> ibn Ka^cb. His mother tongue was Persian and he never completely lost his Persian accent in Arabic. On arriving in Basra his original intention was to study Islamic law, but when he was ridiculed by people for the grammatical mistakes he made in Arabic he decided to study Arabic grammar instead (Versteegh & Versteegh, 1997).

The traditional study of morphology is known in Arabic linguistics as *Al-Mīzān Aṣ-Ṣarfī* الميزان الصّر في 'the balance of morphology/the morphological measure'. In theory, it seems highly systematic since it organizes the relationship between the root and templatic patterns by representing Arabic words within a limited number of patterns, considered to be templates to which consonants and vowels are associated.

The rules of this traditional account *Al-Mīzān Aṣ-Ṣarfī* الميزان الصرفي 'the balance of morphology/the morphological measure' were set down early on by the Arab grammarians Sibawaih (1966) (ed. by Abd al-Salām Hārūn) and Ibn Jinnī (1960) (ed. by Ibarhem Mustafa). They abstracted the Arabic consonants (f, c, and l) (فعل) to distinguish the different morphological processes that a word form may undergo. Thus for the most frequently attested triliteral roots, (f = 0) represents the first radical of the root, $(c \in 1)$ represents the second radical of the root, and (l = 0) represents the third radical of the root. Furthermore, they used the pattern (f, c, l, and l) (i = 1) to represent quadriliteral roots, where the third letter (l = 0) is duplicated. Augmented radicals and/or vocalisms are then inserted within the morphological pattern.

Sibawaih (1966) dealt sporadically with many morphological issues related to derivation; he discussed the morphological patterns of verbs and nouns in some detail, while also considering the functional meaning of these patterns. In addition, he discussed the origins of derivation in Arabic, considering the verbal noun to be the origin from which to derive the verb. This issue was later debated by the Basran and Kūfan schools of medieval Arabic grammatical analysis, which debated whether the verbal noun or the verb itself was the input of derivatives. The Basran grammarians considered the verbal noun to be the input of derivation, whereas the Kūfan grammarians considered the verb to be the input of derivation (Al-Anbāri 1961).

Within modern Arabic literature on Arabic grammar, a number of books have appeared which deal with particular areas of Arabic morphology. These books draw on the traditional literature of the medieval Arab grammarians. Modern Arab grammarians (Al-Hamalāwi 1957; Hasan 1969; Al-Rājhi 1973; Al-Sayyid 1998; Al-Halawāni 1978; Nahir 2001) attempt to provide an authoritative description of Arabic morphology, organizing and disambiguating morphological topics that were scattered in traditional references.

1.3 Research problem

This study aims to investigate nominal derivative formation in Arabic by providing a comprehensive analysis of nominal derivatives in view of state-of-the-art theories and approaches in modern linguistics. A multi-level approach is employed, involving semantics, syntax, morphology and prosodic phonology. Semantically, Arabic verbs¹ are classified in order to determine variation in deriving their valid nominal derivatives. Furthermore, the semantic compatibility (agreement) between the verbs and the nominal derivatives is taken into consideration in order to determine the validity of deriving nominal derivatives from a given verb. Morphologically, root-based and word- or stembased approaches to word formation will be reviewed in the light of variation in the linguistic data. Phonologically, the prosodic structures of the nominal derivatives will be represented to determine realizations of each nominal derivative. In this study, the nominal derivatives (six types) in Arabic are categorized as: the active participle (agent noun), the passive participle, the form of exaggeration, the instrumental noun, the qualificative adjective, and the locative noun.

The Semitic languages, to which Arabic belongs, essentially have highly nonconcatenative (non-agglutinative) morphological systems. This means that word formation is not created through the concatenation (i.e. linking together) of morphemes, as is the case in largely concatenative morphological systems such as English. Semitic languages employ nonconcatenative word-formation processes including template change, infixation and gemination to create further stems. The stems in Arabic are governed by canonical patterns (templates) that include vowels and consonants, conveying particular functional meanings as well as grammatical and syntactic information. This feature has attracted the attention of scholars of modern linguistics. Varying theoretical frameworks have been proposed to

^{1.} The verbs are dealt with as the input of the derivation process, while the nominal derivatives are dealt with as the output, i.e the nominal derivatives are derived from the verbs.

approach nonconcatenative morphology. Within the literature, Arabic is treated as a prime example, alongside Hebrew, of a nonconcatenative morphological system.

The current study aims to explore the characteristics of formation of the nominal derivatives (derived nouns) in Arabic by providing a new comprehensive analysis of nominal derivatives. It focuses on the six nominal derivatives mentioned above derived from Form I verbs.² The formation of each type of nominal derivative will be analyzed from its base (the input) to its final form (the output).

Recent literature on Semitic morphologies has focused on Hebrew and Arabic. The literature on Arabic morphology is mainly dedicated to verbal morphology and the formation of broken plurals, despite the extensive use of nominal derivatives in the language. For this reason, the current study selects nominal derivatives as the linguistic data which will undergo a multi-level analysis that deals with different aims and various data from the literature.

The study employs a multi-level approach involving semantics, morphology and prosodic phonology. At the semantic level, it establishes 44 semantic classes of Form I Arabic verbs (the base form), from which the nominal derivatives are derived. The study proposes that the semantics shows why some nominal derivatives cannot be formed from some verbs while they can be formed from other verbs. For example, a locative noun and an instrumental noun cannot be derived from verbs of emotion, verbs of colouring, or verbs of bodily qualities. Addressing this question through considering the semantic classification of verbs will contribute to filling a gap in the literature.

Dichy and Farghaly (2007) emphasized that there is a problem resulting from considering the morphological patterns (including the nominal derivative patterns) as applicable to any verb to generate word forms. According to Dichy and Farghaly (2007) and to the best of the author's knowledge, no attempt has hitherto been made to provide linguistic solutions to determine which patterns can be applied to particular verbs. Therefore, the current study

^{2.} Form I verb is the base form of the verbal system in Arabic, from which other verbal and nominal forms are derived.

aims to address this problem based on the semantic classification of Arabic verbs, and then explore the interaction between the semantic features of a class of verbs and the semantic features of each nominal derivative type in order to determine eventually which nominal derivatives can be validly derived from a given verb, as well as which nominal derivatives cannot so be derived.

At the morphological level, the word-formation processes of the nominal derivatives will be described and analyzed in view of appropriate assumptions presented in the literature. Morphologically, the current study aims to review the two main approaches in Semitic templatic morphology: one approach that supports a root-based analysis, and one that supports a word- or stem-based analysis, in an attempt to determine whether a wordformation derivative is root-based, stem-based, or both. At the prosodic phonological level, the current study intends to analyze and describe the prosodic structures of the nominal derivatives by providing a prosodic representation which determines realizations of each nominal derivative.

1.4 Research questions

The study aims to address the following main research question and sub-questions:

- What are the characteristics of the formation of nominal derivatives in the Arabic language semantically, morphologically and phonologically (prosodically)?
- What are the semantic restrictions of the formation of valid nominal derivatives? Why can nominal derivatives be formed from some verbs but not from others? From a computational linguistic perspective, the question is, how to prevent overgeneration in deriving nominal derivatives?
- What are the morphological characteristics of nominal derivatives in terms of morphological processes, and the input of their formation (is it the root, the stem, or both?)?
- What is the prosodic structure of the nominal derivatives?

1.5 Research motivation and contribution

The motivations behind this research initially came from the field of computational linguistics. Computationally, nominal derivatives (the output) can be generated from given verbs (the input); however, the problem is that some potential nominal derivatives are not valid semantically and not in use. This problem is described as overgeneration. Here, the question that arises is how we can avoid overgeneration. I hypothesize that the solution may lie in the semantics and syntax of verb classes.

This study involves the semantics-morphophonology interface. In the literature, much work has addressed the interface between morphology and phonology; however the semantic-morphology interface has not been examined in depth within the literature on Semitic languages. The study proposes that the semantics shows why some nominal derivatives cannot be formed from some verbs while they can from other verbs. Addressing this question through considering the semantic classification of verbs will contribute to filling a gap in the literature.

In terms of the data, the literature on Arabic morphology focuses principally on verbal morphology and the formation of broken plurals, despite extensive use of nominal derivatives. Therefore, the current study selects nominal derivatives as linguistic data for analysis. It conducts a comprehensive approach that deals with various aspects (semantic, morphological and prosodic) and different data from the literature. In this study, six different types of nominal derivatives are examined. This variety of data will help to address the question as to whether derivation is from the root or the stem.

The findings of this study will be particularly beneficial to morphologists and prosodic phonologists. Furthermore, by seeking to address the overgeneration problem, it will interest people concerned with computational linguistics and corpus linguistics, where it could be useful in developing wordNet databases and automatic generators.

1.6 Transliteration

In the current study, Arabic characters are represented by the transliteration system (phonetic symbols) given in Table 1:

Arabic characters	Transliteration
ç	Э
ب	b
ت	t
ث	<u>t</u>
ج	j
ζ	ķ
Ċ	x
د	d
ć	₫
ر	r
ز	Z
س	S
ش	Š
ص	Ş
ض	d
ط	ţ
ظ	<i>Ż</i>
ع	с
غ	ġ
ف	f
ق	q
ك	k
ل	l
م	т
ن ن	n
ھ	h
و	W
ي	У

Table 1: Arabic characters and transliteration symbols

The short vowels in the Arabic language are represented as a, i, u, while the long vowels are represented as \bar{a} , \bar{i} , \bar{u} .

1.7 Thesis structure

Chapter One presents a background to the Arabic language and its linguistic tradition, covering historical aspects of Arabic within Semitic languages, the development of Arabic and its rapid spread, the status of Arabic over the centuries, and traditional linguistic schools and prominent scholars. The chapter then discusses the topic of the present study, the research problem, research questions, research motivation and contribution, and the overall structure of the thesis.

Chapter Two examines the literature on Arabic phonology and morphology, as well as other literature relevant to the study. It starts with a discussion of nonconcatenative templatic morphology, the templatic structure in Arabic, and the phenomena that characterize nonconcatenative templatic morphology, including infixation, gemination, circumfixation, and melodic overwriting. Thereafter, the chapter deals with two theoretical frameworks which have been applied to Arabic: autosegmental theory, and prosodic theory. The chapter argues for different approaches to word-formation in Arabic, including the root-based approach and the word- or stem-based approach. Additionally, three types of verb classification, philosophical, syntactic, and semantic, are presented. The chapter ends with a review of the computational literature on Arabic morphology that includes morphological analysis and morphological generation.

Chapter Three introduces the research methodology. It starts with demonstrating our criteria for classifying Arabic verbs, based on semantic descriptions, transitivity and syntactic frames. The selected data comprise 980 verbs (Form I) divided into 44 classes. Thereafter, the chapter demonstrates the multi-level analysis of nominal derivative formation that involves the semantic restrictions/constraints of driving the nominal derivative from its input (Form I verb), the morphological formation of the nominal derivatives. This chapter ends with the computational application of this research where a computational system of the nominal derivatives will be implemented.

Chapter Four presents our Arabic verb classification. 980 Arabic verbs are presented in 44 classes. Each class includes a semantic description, their relation to Levin's classes and Vendler's classes, transitivity tests, class members, the syntactic frame of transitives and intransitives, the nature of the subject and the object, and examples of the syntactic frame of each listed verb.

Chapter Five provides a multi-level analysis (semantics, morphology, prosodic phonology) of the six types of nominal derivatives. It starts with analyzing and describing the morphological and prosodic structure of the six selected types of the nominal derivatives, namely: the active participle (agent noun), the passive participle, the form of exaggeration, the instrumental noun, the qualificative adjective, and the locative noun. The chapter provides a semantic analysis of the nominal derivatives in order to determine semantic restrictions/constraints on the formation of nominal derivatives based on the semantic compatibility between the semantic features of the verb classes and the semantic features of the nominal derivatives.

Chapter Six deals with the formation of the nominal derivatives from a computational linguistic perspective. It discusses the morphological analysis and morphological generation. Morphological generation problems are taken into consideration, specifically the overgeneration problem. The chapter then deals with overgeneration in the formation of nominal derivatives, as well as our solution to the overgeneration problem. The chapter ends with an evaluation of the performance and accuracy of our model.

The study ends with a conclusion that summarizes the stages of the research. The conclusion presents the research findings throughout the study within the semantic, morphological and prosodic phonology levels of analysis, and ends with recommendations for future research.

Chapter Two

Theoretical background and research context

2.1 Introduction

This chapter presents the theoretical background and research context to the study. It starts with a discussion of nonconcatenative templatic morphology, templatic structure in Arabic, and the characteristic features of Arabic nonconcatenative templatic morphology, including infixation, gemination, circumfixation, and melodic overwriting. Thereafter, the chapter deals with two theoretical frameworks which have been applied to Arabic: autosegmental theory and prosodic theory. The chapter considers different approaches to word-formation in Arabic, including the root-based approach and the word- or stem-based approach. Three types of verb classification are presented: philosophical, grammatical, and semantic. The chapter ends with a discussion of some computational literature on Arabic morphology that includes morphological analysis and morphological generation.

2.2 Morphology

Morphology deals with word structure and word formation processes. In terms of word formation, morphology is categorized into derivational morphology and inflectional morphology. The distinction between these two types is attributed to the function of affixes involved in word formation. Essentially, derivational affixes produce a new word: a word that differs either in semantics or in word class. For example, adding the affix *-al* to the noun *information* in English creates the adjective *informational*. Affixation of *-ship* to *friend* creates an abstract noun with differing semantics from the stem. In contrast, affixation of inflectional affixes does not create a new word, but gives a different form of the same word, for example adding *-s* to the noun *home* gives the plural form *homes*, and adding *-ed* to the verb *walk* gives the past tense form of the verb. Arabic morphology traditionally covers derivational morphology (e.g. the ten³ forms of the verb, verbal and nominal derivatives) as well as inflectional morphology (e.g. verbal paradigms), but it does not encompass grammatical case and mood which are considered under *nal*_iw (syntax) (Ryding 2005).

^{3.} There are actually 15, but only 10 are in common use today.

2.3 Nonconcatenative morphology

The morphological structure of languages of the world can be categorized into concatenative and nonconcatenative. Concatenative word formation, which is widespread, is based on prefixation and suffixation. In concatenative morphology, a series of continuous morphemes (morphemes linked to one another in a linear manner) can form a new word. For instance, the English word *unacceptable* comprises three morphemes: the negative prefix *un*-, the stem *accept* and the adjectival suffix *-able*. By contrast, nonconcatenative morphology involves infixation and/or prosodic template change. In the case of Arabic, phenomena that should be highlighted in this context include templatic structure, infixation, gemination (reduplication), and circumfixation.

Watson (2002) defines the boundaries between different categories in the application of morphology to Arabic. Watson (2002: 132) assumes that "Arabic has two morphological levels. Level one, which affects the stem of the word predominantly, can be said to correspond roughly to the nonconcatenative (or infixal) morphology; and level two, which does not affect the stem of the word, works predominantly by adding affixes to the beginning and end of the word stem. As a working hypothesis, level-one morphology is roughly equivalent to derivational morphology, and level-two morphology to inflectional morphology. However, some level-two affixes change the class membership of the word from adjective to noun, or from noun to adjective and therefore appear to be part of the derivational morphology".

Inflectional morphology in Arabic most commonly uses prefixation or suffixation, while derivational morphology characteristically involves infixation and/or templatic change. For example, the inflectional suffix -*at* \rightarrow can be added to the stem *qātili* 'killer' to obtain the feminine form *qātilat* قاتله' female killer'. In the derivational morphology, the active participle *qātil* 'killer' of the Form I verb *qatal* 'to kill' is derived through a different prosodic template.

In the templatic structure of Arabic, the structure of a word (verb and noun) is dependent on a pattern (template) and a root. In Arabic, each stem⁴ is governed by a canonical templatic pattern that includes vowels and consonants, conveying particular functional meanings, grammatical and syntactic information, and an indication of word class (grammatical category). In other words, the templatic pattern comprises three components: a series of discontinuous consonants, a series of discontinuous vowels, and a templatic pattern. For instance, the templatic pattern $C_1VVC_2VC_3$ together with the vocalic melody a-i- represents the active participle in Arabic, and can be applied to any valid root to derive the active participle of Form I verbs, as shown in Table 2.

Root	Active participle		
'to write' کتب <i>k-t-b</i>	$k\bar{a}tib$ کاتب $C_1VVC_2VC_3$ 'writer'		
'to play' لعب <i>l- °-b</i> لعب	$l\bar{a}^{c}ib$ لاعب $C_1VVC_2VC_3$ 'player'		
'to kill' قتل <i>q-t-l</i>	$q\bar{a}til$ قاتل $C_1VVC_2VC_3$ 'killer'		
'to read' قرأ <i>°-q-r</i>	$q\bar{a}ri^{\circ}$ قارِئ $C_{1}VVC_{2}VC_{3}$ 'reader'		

Table 2: Form I active participle

At the derivational level, there are two types of templatic pattern: verbal stem patterns and nominal stem patterns. Verbal stem patterns in Arabic are more restricted than the nominal stems (Watson 2002: 133). There are ten frequently encountered verb forms (see Table 3 below), where Form I is the base form from which the other nine forms (Forms II–X) are derived. According to Watson (2006b: 432), "Forms II, III, and IV are derived from Form I by extension of the stem; Forms V and VI are derived by prefixation of *ta*- to Forms II and III, respectively. Forms VII, IX, and X involve various types of prefixation, and Form VIII is derived from Form I by infixation of */t/* after the leftmost root consonant. No consonantal root in Modern Standard Arabic has all ten verb forms, and a few verbs have one or more derived forms but lack the basic form".

^{4.} The stem is the basic form of the word before adding inflectional affixes.

Form	Root	Pattern	Example	Gloss
Form I	ك ت ب <i>k-t-b</i>	C1aC2aC3	کتب katab	'to write'
Form II	ك ت ب <i>k-t-b</i>	C1aC2C2aC3	كتّب kattab	'to dictate'
Form III	ك ت ب <i>k-t-b</i>	C1aaC2aC3	كاتَب kātab	'to correspond with'
Form IV	جـ ل س <i>j-l-s</i>	°aC1C2aC3	أجلَس ajlas°	'to seat'
Form V	ع <i>L</i> م ^c - <i>l</i> - <i>m</i>	taC1C2C2aC3	تعلّم ta ^c allam تعلّم	'to learn'
Form VI	ك ت ب <i>k-t-b</i>	$taC_1aaC_2aC_3$	تكاتب takātab	'to correspond'
Form VII	ك ت ب <i>k-t-b</i>	$^{\circ}inC_{1}aC_{2}aC_{3}$	^o inkatab انکتب	'to subscribe'
Form VIII	ك ت ب <i>k-t-b</i>	${}^{\circ}iC_{1}taC_{2}aC_{3}$	[»] iktatab اکتّتب	'to register'
Form IX	ה-m-r דאנ	${}^{\circ}iC_{1}C_{2}aC_{3}aC_{3}$	[»] iḥmarar احمر	'to become red'
Form X	÷ כ ج x-r-j	$^{\circ}istaC_{1}C_{2}aC_{3}$	^ə istaxraj	'to extract'
			استخرج	

Table 3: The ten derived verb forms in Arabic

* C_1 , C_2 , and C_3 represent the first, second, and third consonantal root letters respectively.

Infixation, gemination, melodic overwriting, and templatic change are phenomena that characterize nonconcatenative morphology. Infixation is a very common morphological process in Arabic, in which an infix is inserted between the root or word-stem letters. The infix is a dependent bound morpheme; it can consist of vowels or consonants. For example, Form VIII is derived from Form I by infixing t after its first root consonant, such as *°iktatab* 'to register' that is derived from Form I verb *katab* ' λ 'to write'.

Gemination involves reduplication of a root consonant. Form I might involve gemination when the second and third root consonants are the same, as in the verb *dall dall* 'to indicate'. The formation of the Form II verb is based on the gemination of the medial Form I consonant, for example, *qattal* قتل (Form II) is formed by duplicating the *-t-* of the Form I verb *qatal* 'to kill'. Orthographically, the geminated consonant is represented using a symbol called *shadda* which appears above the geminated consonant.

Melodic overwriting is a morphological process in which the vocalic melody is overwritten with another vocalic melody to form a new stem. More specifically, this kind of morphological process occurs only at the stem level. For example, the passive of Form I verbs is formed by overwriting the vocalic melody *a-a*, *a-u*, or *a-i* with vocalic melody *u-i*, as in the passive verb *fuhim* $\stackrel{\text{def}}{\stackrel{\text{def}}{=}}$ 'it is understood', whose active form is *faham* $\stackrel{\text{def}}{\stackrel{\text{def}}{=}}$ 'he understood'.

Templatic change involves associating the root consonants with a different template, and often also melodic overwriting. For example, the active participle $q\bar{a}ri^2$ قارى 'reader' in Arabic is formed by associating the consonants of the Form I verb $qara^2$ (to read' with the template $C_1\bar{a}C_2iC_3$ (CVVCVC) and overwriting the vocalic melody -*a*- with the melody $-\bar{a}-i$ -.

2.4 Autosegmental phonology

In generative phonological theory, there are two contrasting approaches: segmental and autosegmental. Within the segmental approach, the representation of morphological and phonological operations is based on the notion of linearity: i.e. a linear string of elements that interact with each other in one-to-one behaviour according to morphological and syntactic rules. This theory has been applied to languages whose morphology is based on the linear concatenation of morphemes through prefixation or suffixation. For example, the English word *unhelpfully* is formed by concatenation of four lexical morphemes *un - help - ful - ly*. Concatenative theory does not, however, serve languages which exhibit significant nonconcatenative (nonlinear) morphology, such as Arabic, Hebrew, Amharic, Syriac, Dakota, Tagalog, Terena, Tiv and Ulwa. To exemplify the difference between concatenative and nonconcatenative morphologies, the English active participle word *writer* is formed by suffixing the morpheme *-er* to the verb *write*, whereas the Arabic active participle word $k\bar{a}tib \rightarrow 2\bar{k}$ (writer' is formed from the verb *katab* 'write' ψ by templatic change and melodic overwriting.

The first linguistic framework adopted to describe nonconcatenative (nonlinear) morphologies was produced by John Goldsmith in his PhD dissertation in 1976. Goldsmith (1976) developed autosegmental theory, which was applied to tone languages (tonal

phenomena) in which the phonemic elements cannot be represented or analyzed by a linear approach. Since then, the autosegmental framework has been extended to handle other features that involve multiple segments. Building on the autosegmental framework, McCarthy (1979, 1981) developed a templatic approach to Semitic languages (McCarthy's works will be discussed later in this chapter).

Within autosegmental theory, the phonological structure of a word comprises several tiers, each tier consisting of a linear arrangement of components; these tiers are connected to each other by means of association lines that show how the tiers are coarticulated (Crystal 2008: 45-46). In this theory, each linguistic feature can be represented on its own separate autosegmental tier that allows it to act independently in different morphophonological operations. Features are associated to tiers through Association lines. These association lines vertically link segments that are placed on separate autosegmental tiers.

According to Kager, van der Hulst et al. (1999), the *Well-Formedness Condition* is universal. They claim that the *Well-Formedness Condition* cross-linguistically has the following form: "(a) all vowels are associated to at least one autosegment, (b) all autosegments are associated to at least one vowel, (c) association lines may not cross, and (d) association of unassociated autosegments precedes drawing lines from associated autosegments" (Kager et al. 1999: 5).

2.5 Prosodic phonology

The term prosody is used to indicate phonological phenomena which are related with the phonological elements at a level larger than an individual sound, such as syllables and syllabification, word stress and intonation (Plag 2002). The theory of prosodic structure was developed by Selkirk (1980) and Nespor & Vogel (1982,1986). McCarthy and Prince (1990a, 1990b, 1993, 1995, 1996, 1999) then recognised that prosodic structure could usefully be applied to analyses of morphology, and presented a prosodic theory that benefits from autosegmental phonology and templatic morphology; the notion of the prosodic template has also been adapted here. They suppose that words are identified not as sequences of consonants and vowels, but consist of four prosodic units: mora, syllable, foot, phonological word. In the view of Ussishkin (2005), the advantage of prosodic theory over templatic theory is that: "under prosodic morphology, templates are no longer viewed

as extra-theoretical structures that the language happens to make use of under certain phonological or morphological circumstances. Rather, their existence is motivated by the fact that their prosodic make-up is independently necessary" (Ussishkin 2005: 172).

The principles of the prosodic units were inspired by Selkirk (1980). In this theory, McCarthy and Prince (1986, 1990ab, 1993a, 1995, 1999, 2004) established a prosodic representation that is based on a hierarchy within a metrical tree comprising the phonological word, the foot, the syllable, and the mora. These four prosodic units are symbolized as (PW), (F), (σ), and (μ) respectively (McCarthy and Prince 1990; cf. Selkirk 1980). For example, the prosodic representation of the Arabic word *tannuur* \vec{x} clay oven' (*CVCCVVC*) is:



In the prosodic hierarchy, each prosodic unit occupies a separate level within the prosodicphonological representation. Here, the hierarchy is read from the upper level to the base, i.e. the phonological word (PW) is located in the upper level above the foot level (F). Below the foot level is the syllable level (σ) that is above the mora level (μ) (McCarthy and Prince 1990). Watson (2002: 129) and others argue for minimality in the prosodic hierarchy where each higher unit includes minimally one unit from the units that come below; therefore, the phonological word (PW) contains at least a single foot (F), the foot contains at least a single syllable (σ), and the syllable contains at least a single mora (μ). Potentially, the phonological word (PW) can match a syntactic or a function word, as in: *tannūr* 'clay oven' given above, or involve the concatenation of a syntactic and one or more function words, as in San'ani Arabic: *galliš* [said3m-to-2fs] 'he said to you f.s.'.

Languages differ as to whether they allow sub-minimal or degenerate feet, that whether syllables that are left over at the beginning or end of a phonological word after the foot parse surface as feet. Therefore, degenerate feet are "[s]ub-minimal feet which survive to

the surface" (Watson 2002: 88). Degenerate feet can receive stress in prosodic strong positions. Thus in a language in which stress is assigned to the right-most foot, the final monomoraic syllable in a word of the structure *CVCCVC* will be assigned word stress in a language that allows degenerate feet, but not in a language which does not permit degenerate feet. Most documented dialects of Arabic, including Modern Standard Arabic, have a strong prohibition on degenerate feet, and thus monomoraic syllables left over at the end of the foot parse remain unparsed.

The determination of syllabification patterns plays a major role in prosodic analysis. In this context, it is worthy of mention that standard Arabic has two salient features in the architecture of syllables. First, no syllable can begin with two consonants. Second, no syllable can begin with a vowel (cf. Ryding 2005: 35). As explained by McCarthy and Prince (1990a, 1990b), Arabic has three core syllables (*CV*, *CVV*, *CVC*): the light syllable which consists of a consonant followed by a short vowel, represented by *CV* such as $ka \leq in katab$ in katab in consist = core in maktab in maktab = core in more in more in more in the heavy syllable*CV*takes one mora (µ), whereas heavy syllables take two moras (µµ), as shown below:



The first mora in all syllables refers to a vowel (u, a, i), while the second mora in the heavy syllable refers to the second part of a long vowel or diphthong *CVV*, or the coda consonant *CVC*. The last consonant of the *CVC* syllable is regarded as extrametrical if this syllable comes as the final syllabe in a word (word-final position) (cf. Watson 2002: 57). McCarthy and Prince (1990a) argue for two further types of syllable, called super-heavy. These syllables are *CVVC* and *CVCC* and are limited to word-final position. Both can be regarded as a heavy syllable followed by an additional consonant (*CVV-C* and *CVC-C*). This consonant is therefore an extra-prosodic consonant that does not interact in the prosody of the word. The prosodic representation of word-final syllables in Arabic is illustrated below:



In Arabic, the final consonant that occurs in a final heavy syllable CV < C> is considered extrametrical, while the final consonant that occurs in final superheavy syllables such as CVV < C> and CVC < C> is considered extrasyllabic (Watson 2002: 58, 90). According to Watson (2002: 58) the extrametrical consonants are incorporated into "the syllable node of the final syllable", while the extrasyllabic consonants are "not incorporated into the adjacent syllable at any stage in the derivation".

2.6 Theoretical approaches to Arabic morphology

In modern theoretical literature dealing with phonology and morphology, various approaches to nonconcatenative templatic morphology have been proposed. Alongside Hebrew, Arabic is treated as a prime example of a nonconcatenative morphological system which requires special theoretical consideration. Many languages of the world exhibit predominantly concatenative morphologies, whereby morphemes are linked linearly, as in the English word *un-happi-ly*. Languages such as Arabic and Hebrew present a challenge because most of their derivational morphology and some of their inflectional morphology does not consist of strings of independently identifiable morphemes. As we have seen, the agent noun in English *writer* is formed by attaching the suffix *-er* to the verb *write*, whereas the Arabic equivalent $k\bar{a}tib$ is formed by inserting a series of vowels ($-\bar{a} - i -$) within the triliteral root (*k-t-b*).

In Arabic, a particular sequence of vowels may signify a particular grammatical category. For example, the root k-t-b with a-a vowels signifies the perfective active katab 'he wrote', the root k-t-b with u-i vowels signifies the perfective passive kutib 'it m. was written', the root k-t-b with a-u vowels signifies the imperfective active aktub 'I write', and the root k-t-b with u-a vowels signifies the imperfective passive (y)uktab 'it m. was written' Accordingly, it is necessary to make sure that the series of consonants and vowels appear in the correct order (Spencer 1991).

The modern morphological analysis of Semitic languages today is categorized into two opposing approaches: the root-based approach, e.g. McCarthy (1981); Prunet, Béland et al. (2000); Davis and Zawaydeh (2001); and the word- or stem-based approach, e.g. Heath (1987); Darden (1992); Bat-El (1994, 2001); McOmber (1995); Ratcliffe (1998); Gafos (1999); Ussishkin (1999, 2000, 2003, 2005).

2.6.1 The root-based approach

Proponents of the root-based approach argue that the consonantal root is the input (source), from which semantically related words are derived by combining the root consonants (often three) with a canonical templatic pattern that includes consonant positions and vocalic affixes. Within this approach, the root is regarded as a morpheme. The notion of the root may involve cognitive reality where native speakers rely on the root in forming words within their mental lexicon.

Within the domain of autosegmental phonology (Goldsmith 1976), the earliest influential attempt to address Semitic structure within a formal linguistic framework was carried out by McCarthy (1979, 1981), who presented a root-based approach to the types of nonconcatenative morphology exhibited by Arabic and Hebrew. Before McCarthy's work (1979), there had been no elegant theoretical framework to describe and analyze the word formation processes of Semitic languages in the modern linguistic literature (Katamba 1993).

According to McCarthy, the templatic representation of a stem is composed of three separate levels called autosegmental tiers: the root tier (consonantal tier), the template tier (CV skeleton), and the vocalization tier (vocalic melody). The morphemic tier representation of the stem *fatah* is shown below:



According to McCarthy (1979, 1981), the three autosegmental tiers (root, template and vocalic melody) are considered independent morphemes. The consonantal root is a base morphological unit, conveying the core lexical meaning. In Arabic, the root normally consists of three consonants, and less frequently of two or four consonants. From a root-based perspective, the majority of Arabic words (nouns and verbs) are derived from triliteral consonantal roots, uncommonly from two- or four-consonantal roots. For instance, the consonantal root *d*-*r*-*s* \dots has the basic lexical meaning of 'studying', from which these words are derived: *darsun* $\tilde{c}_{\tilde{c}}$ 'lesson', *mudarris* $\tilde{c}_{\tilde{c}}$ 'teacher', *dirāsah* $\tilde{c}_{\tilde{c}}$ 'study', *madrasah* $\tilde{c}_{\tilde{c}}$ a school', and *dāris* $\tilde{c}_{\tilde{c}}$ a student'. In all of these derived words, the same consonants *d*-*r*-*s* remain constant with the same orthographic order.

The templatic tier represents "a sequence of empty timing tier entities [typically] notated as Cs and Vs" (Watson 2002: 126). In Arabic, the grammatical category of a word (such as active participle, passive participle, and instrumental noun) has its particular templatic patterns. For example, the active participle of Form I verbs has the templatic pattern $C_1\bar{a}C_2iC_3$, as in $k\bar{a}tib$ 'writing; writer m.'. A template may involve a particular affix, as in ma- of the Form I passive participle template: $ma-C_1C_2\bar{u}C_3$: $maft\bar{u}h$ · \dot{a} · \dot{c} · $\dot{c$

The vocalic melody is the third morpheme in the templatic representation. It consists of vowels which are inserted between the root consonants. It plays a major role in making variations in "the voice (active or passive) in verbs, agentive relations in nouns derived from verbs, and singular–plural relations in nouns" (Watson 2002: 126). The vocalic melody plays an important role in pronouncing Arabic words and indicating particular grammatical properties of a word. Table 4 exemplifies how applying various vocalic melodies to a particular root leads to changing the word class.

Root	Vocalic melody	Example	Word class
كەتب k-t-b	-a-a-	كَتَبَ katab	Active voice of the Form I verb
ك ت ب k-t-b	-u-i-	کُتِبَ kutib	Passive voice of the Form I verb
ك ت ب k-t-b	-aa-	كَتْبَ kattab	Active voice of the Form II verb

Table 4: Examples of various vocalic melodies applied to the root k-t-b

ك ت ب k-t-b	-ui-	كُتُبَ kuttib	Passive voice of the Form II verb
ك ت ب k-t-b	-u-u-	كْتُبُ kutub	Plural noun of the word <i>kitāb</i> 'book'

Besides the three autosegmenal tiers (consonantal, template and vocalic), Arabic words may have affixes that are not part of the word stem, each of which is represented by a fourth separate tier (an affixal tier) that functions as an independent morpheme. The term 'melodic elements' is used by McCarthy to indicate the components of these multiple autosegmental tiers (root, vocalic melody, *CV* template, affixal tier). These autosegmental tiers are linked by a process of association (mapping lines) based on universal association conventions (McCarthy 1981).

Association takes place as follows: melodic elements (vowels and consonants) are linked with melody-bearing elements (V and C slots) one-to-one from left to right, i.e. vowels are linked to V slots, and consonants are linked to C slots. This convention is illustrated in the following chart (McCarthy 1981: 382); capital letters represent melody-bearing elements and small letters represent melodic elements:

The second convention comes after applying the first convention where there is "one unassociated melodic element and one or more unassociated melody-bearing elements." In this case, the unassociated melodic element is associated with all of the unassociated melody-bearing elements (McCarthy 1981: 382). The following diagram illustrates the second convention:

$$\begin{array}{ccc} A B C D \\ | & | \\ x & y & z \end{array} \qquad \begin{array}{ccc} A B C D \\ | & | \\ x & y & z \end{array} \qquad \begin{array}{ccc} A B C D \\ | & \bigvee \\ x & y & z \end{array}$$

In the third convention, McCarthy states that "If all melodic elements are associated and if there are one or more unassociated melody-bearing elements, all of the latter are assigned the melody associated with the melody-bearing element on their immediate left if possible"
(McCarthy 1981: 382). This process causes what is described as automatic spreading as shown in the following diagram:



Furthermore, there is another convention that must be taken into consideration: the Well-Formedness Condition, which operates according to two rules. First, each C slot or V slot in the templatic tier should be linked with at least one melody element. Second, the association lines (mapping) should not cross each other (McCarthy 1981).

Following McCarthy's (1979, 1981) work, other linguists have presented various evidence in favour of the root-based approach. These works provide experimental evidence based on how native speakers create words. The root-based approach has also been adopted to analyze other (non-Semitic) languages, such as Spanish (Piñeros 1998). The proponents of this approach claim that the consonantal root serves as an independent morpheme that has psychological reality. Following McCarthy (1979, 1981), Prunet, Béland et al. (2000) found the consonantal root to be an independent morpheme. To support this view, they provided external evidence from a psycholinguistic perspective, which is based on the status of the lexical units embedded in the native speakers' mental lexicons. They also provide other external evidence in favour of the templates' morphemic status based on language games, slips of the tongue, and aphasic errors. Their analysis of metathesis errors made by ZT⁵ supports the argument for the root as a lexical unit, thus supporting the morpheme-based analysis of morphology. Nevertheless, this argument does not deny output-to-output word-formation processes.

In an article dealing with the formation of hypocoristics in colloquial Arabic, Davis and Zawaydeh (2001) present evidence in favour of considering the consonantal root as an independent morpheme. They argue that word formation of hypocoristics is based on output-to-output processes, which nonetheless need to make reference to the consonantal root. Furthermore, they claim that the argument of the formation of a word from another independent word does not counter the morphemic status of the consonantal root, as it is

^{5.} ZT is 'a bilingual Arabic-French aphasic patient' (Prunet, Béland et al. 2000: 642).

an "output-to-output word formation process". Thus, it is still possible to make reference to the consonantal root (Davis and Zawaydeh 2001: 514).

2.6.2 The stem-based approach

By contrast, a word- or stem-based approach to Semitic morphology has been proposed by a number of researchers. In this approach, a word is derived from an independent word, and the template act as a constraint to filter the compotation of vowels and consonants. The proponents of this approach claim that word formation is based on a whole word or stem input rather than the consonantal root. Heath (1987) adopts the stem-based approach in analyzing Moroccan Arabic, and Darden (1992) adopts it in analyzing Cairene Arabic.

In addition, McOmber (1995) argues that full words in Arabic can be analyzed and represented through *CV* templatic representation without reference to the consonantal roots. Both Bat-El (1994) and Ussishkin (1999) analyzed the formation of denominal verbs in Hebrew, arguing that a stem-based analysis can only account for the morphophonological regularities attested. Likewise, Ratcliffe (1998: 50) argues in favour of the morphological status of consonantal roots, claiming that "(phonologically possible) words rather than three-consonant roots are the primitive lexical entries of the Arabic lexicon".

Their arguments emphasize that there is no need to make reference to the consonantal root in word-formation processes. This is because they consider that the consonantal root does not act as an independent morpheme. According to Davis and Zawaydeh (2001: 518), the deniers of the morphemic status of the consonantal root claim "that units smaller than the word cannot be the object of a morphological strategy. Clearly, the consonantal root is the object of a morphological strategy in Arabic hypocoristic formation".

McCarthy's (1992) affixational theory is considered a distinct change from his earlier theory embodying the root-based approach (templatic analysis). Here, McCarthy downplays the notion of combination between templates and roots to form new words. He argues that "the core of the Arabic nominal system is templatic in character, with templates that conform to prosodic Morphology Hypothesis, i.e. templates are defined in terms of the authentic units of prosody" (McCarthy 1992: 2). However, he also argues that "the Arabic and Akkadian verb system is even more radically non-templatic; just a single template underlies all verb forms, and other morphological regularities are derived by rules of affixation, sometimes via prosodic circumscription" (McCarthy 1992: 2). In other words, there is only one template (Form I verb *C1VC2VC3*, like *katab* \simeq 'he wrote') from which the other verb forms are derived by affixational processes and prosodic circumscription; thus, a derived form here does not have its own template.

As a third approach, in a paper that considers the formation of verbal and nominal diminutives in San \circ āni Arabic, Watson (2006a) argues that both types of word formation (root-based and stem-based) can be found in Arabic, where native-speaker judgements confirm the two approaches depending on the semantics of the derivative. Watson (2006a: 189) finds that "semantic similarities between certain stems can only be accounted for by derivation from a fully vocalised stem". For example, diminutive verb stems, such as *twayzar* 'to pretend to be a minister' and *tmaydar* 'to pretend to be a manager', are derived from the fully vocalized stems (base noun) *wazīr* 'minister' and *mudīr* 'manager', respectively (Watson 2006a: 195).

Watson (2006a) also claims that some verbal diminutives must be formed from the root as the basic morphological unit. Evidence in favour of the root as the input to some diminutive formation includes:

(i) The existence of several base forms that share the same consonantal root with the diminutive verb, to explain particular formations, suggesting both that the tCayCaC form, from which the basic consonants are extractable, and that the consonantal roots (triliteral) are identified by native speakers as independent morphological units. To support this evidence, Watson (2006a: 193) presents the following examples:

txaybal 'to act as if stupid' was explained as *xabal* 'stupidity', *axbal* 'stupid', *xabaalih* or *mixbaalih* 'acting stupidly' (root /x-b-l/); *tlaygan* 'to bicker' as *lagaanih* and *layganih* 'bickering' (root /l-g-n/); *tbaylah* 'to act stupid' as *balah* 'stupidity' (root /b-l-h/); and *txayda9* 'to act to deceive' as *xada9* and *xadaa9ah* 'deceit; deception' (root /x-d-9/).

(ii) There are synonyms and near-synonyms which fall into specific abstract semantic fields. Among these, there are shared consonants (two or three) or there are consonantal roots of synonyms and semi-synonyms that share place of articulation features. Watson (2006a: 193) states that: Most verbs describing 'to be stupid/daft/naive' have root consonants with the features [guttural]/[labial]/([lateral]) in various orders, as in: /h-b-l/, /b-l-h/, /x-b-l/, /x-w-sh/, /x-j-f/, /x-b-S/. Verbs describing 'to cling/stick' have the initial root consonants /l- [sibilant]/, as in: /l-z-g/, /l-z-T/, /l-z-m/, /l-s-y/.

In contrast to previous literature, Watson's (2006a) work is open to consideration of both approaches (root-based and stem-based), and not restricted to adopting only one of them. Therefore, the input (root or stem) may differ depending on the linguistic data, the output, and how native speakers describe the data.

In this respect, it should be emphasized that each of the ten verbal forms in Arabic has different nominal derivatives. For example, *kātib* كاتب, *mukattib* مُكاتِب, and *mukātib*, and *mukātib* مُكاتِب are the active participles for the Form I *katab* كَتَّب, Form II *kattab* مُكاتَب, and Form II *kātab* كتَّب, respectively. In Arabic, Form I is the base form from which the other nine forms are derived.

Within the word- or stem-based approach the input may differ. The question therefore arises of what the input in a word- or stem-based approach is. McCarthy (1993) considers the perfect verb form to be the input to verb formation, whereas Benmamoun (1999) claims that the imperfective form is the basic stem in both inflectional and derivational morphology. He bases his argument on the fact that the vocalization of the perfective in Form I verbs can be predicted from the vocalic melody of the imperfective, but that the vocalic melody of the imperfective cannot be predicted from the vocalic melody of the perfective.

Benmamoun (1999) also claims that the imperfective form is not specified for tense where this is the only basic form which does not convey temporal properties. At the morphological level, he considers that there is a harmony between the unmarked case of the imperfective and its essential role in word formation. According to Benmamoun (1999: 175):

This role will be shown to be more pervasive than previously thought. This, in turn, allows for a unified analysis of nominal and verbal morphology. The implication then is that important parts of Arabic word formation are word-based rather than root-based. The special syntactic status of imperfectives therefore makes them better candidates as input forms.

Ussishkin (1999) regards the base nominal form as the input to denominal verb formation in Modern Hebrew. For example, the base nominal forms *dam* 'blood', *xam* 'hot', and *xad* 'sharp' are the inputs of *dimem* 'to bleed', *ximem* 'to heat', and *xided* 'to sharpen', respectively. He also claims in a recent work (2005) that *pa fal* forms in Hebrew (equivalent to Form I in Arabic), themselves fully vocalized words, serve as the base of affixation in forming other binyanim.

2.7 Verb classifications

Verb classification can be viewed from many perspectives. In the literature on English, there are philosophical, syntactic and semantic classifications. The interest in verb classifications serves various research fields in theoretical and computational linguistics. In recent years within the field of computational linguistics, verb classifications have been utilized in many applications such as word sense disambiguation, information access tasks (query generalization), question answering, machine translation, psycholinguistic modelling, and statistical lexical acquisition (Čulo, Erk et al. 2008).

2.7.1 The philosophical classification

Historically, the philosopher Aristotle was interested in categorizing words into their parts of speech in general. His basic notions inspired modern scholars in classifying verbs, and corresponded with what they presented. In modern literature, Ryle (1949), who was inspired by Aristotle, argues that verbs are classified into performance, achievement and activity. Similarly, Kenny (1963) classifies them into state, performance and activity verbs. Based on time, Kenny (1963: 171) distinguishes the three verb classes as "states may 'last for' a time, and activities 'go on for' a time; only performances 'take' time''. Furthermore, Kenny (1963) utilizes syntactic (grammatical) tests to characterize his three classes. For example, he argues that "only performance verbs have a true passive voice" (Kenny 1963: 178) and "all performance verbs have imperatives; no static verb has an imperative" (Kenny 1963: 188). Table 5 shows some examples of the three classifications of verbs presented by Kenny (1963: 175):

Verb class	Examples
State verbs	to be able, to be blue, to fear, to know how, to love, to mean, to perceive, to understand
Performance verbs	to convince, to cut, to discover, to find, to kill, to lift, to learn, to wash
Activity verbs	to enjoy, to keep a secret, to laugh, to listen, to talk

Table 5:	English	examples	of Kenny	's verb	classification
----------	---------	----------	----------	---------	----------------

Since then, Vendler's classification has become the most widely recognized classification of verbs in terms of lexical aspect (aspectual classification of verbs). Vendler (1968) established four basic classes: states, activities, accomplishments and achievements. Within this classification, state contrasts with the three process predicates: accomplishment, achievement and activity (Dowty 1979). These classes have been described by Crystal (2008) as follows:

- The *state verbs* "represent events which last for a period of time without evident change", such as believe, belong, need and know (Crystal 2008: 451).

- The *activity verbs* "represent a type of process event which need not reach a culmination point: walk, for example, is of this type, being dynamic and atelic in character" (Crystal 2008: 9).

- The *accomplishment verbs* "represent a type of process event which extends in time but reaches a culmination point: build, for example, is of this type, being both durative and telic" (Crystal 2008: 6).

- The *achievement verbs* "represent a type of process event which takes place instantaneously: arrive, for example, is of this type, being punctual in character" (Crystal 2008: 7).

In terms of staticity, durativity and telicity, the aspectual features of Vendler's classification have been characterized and schematized by Smith (1991). In his argument, the basic oppositions are punctual vs. durative events, telic vs. atelic (or bounded vs. unbounded), and static vs. dynamic, as illustrated in Table 6.

	Static	Durative	Telic
States	+	+	
Activities	_	+	
Accomplishments		+	+
Achievements			

Table 6: Aspectual features of Vendler's classes

2.7.2 The semantic classification

Levin (1993), largely considered a pioneering work in the literature, provides a semantic classification of English verbs based on the correspondence between their semantics of the verbs and their syntactic behaviour. Levin (1993: 1) assumes that "the behaviour of a verb, particularly with respect to the expression and interpretation of its arguments, is to a large extent determined by its meaning. Thus verb behaviour can be used effectively to probe for linguistically relevant pertinent aspects of verb meaning". Levin's classification relies on using syntactic frames (diathesis alternation) to identify the ability of group verbs to occur in certain syntactic contexts. In other words, a class of verbs is assumed to have common semantic features and certain syntactic alternations.

Levin (1993) identifies 78 diathesis alternations and classifies the 3,104 most common English verbs into 49 semantic classes, sub-divided into 192 sub-classes based on their syntactic alternations. She focuses on the relationship between the verb meaning and its syntactic behaviour. Thus, the verbs that share the same behaviour can be classified within a certain semantic class. Semantic classes are created from the verbs that undergo a certain number of alternations. Levin's classes can be classified within Vendler's; for example, Levin's perception and psychological state classes can be viewed as sub-classes of Vendler's state class, as below:

- Verbs of perception (Levin 1993):

(*detect, discern, hear, notice, see*) can be classified within Vendler's state verb class. - Verbs of psychological state (Levin 1993):

(*fear*, *enjoy*, *scare*, *shame*, *shock*) can also be classified within Vendler's state verb class.

Furthermore, the semantic classification of verbs can utilize the framework of the ontology of semantic entities. Within this framework, nouns and verbs can be classified semantically. Helbig (2006) provides a classification of semantic entities. The notion of entities, according to Helbig (2006: 409), "comprises all things about which something can be stated". The entities in this classification refer mainly to two categories of entity: concrete objects and abstract objects. In addition, he (2006: 45) argues that "entities, which can be thought of as objects, are called conceptual objects (or "objects" for short). This category comprises concrete objects as well as abstract objects (e.g. house, leg, or theory, law, respectively)". In the same way, any verb and its arguments can be classified whether it involves concrete objects or abstract objects. For example, the verb *darab* 'to hit'

involves a concrete entity, while the verb $\geq karih$ 'to hate' involves an abstract entity. The following figure illustrates the ontology of entities as presented by Helbig (2006: 410):



Within this framework of entities, a given semantic relation can be defined based on a number of categories, including: objects, situations, descriptors, qualities, graduators, quantities, and formal entities. These entities are based on dualism, within which the semantics of a language can be represented.

2.7.3 The syntactic classification

As universal classifications, verbs can be syntactically classified according to their tense (past, present and future), voice (passive and active), regularity (regular and irregular), and transitivity (transitive and intransitive). In this respect, it is worth noting that the interest of the present study is mainly confined to the classification of verbs in terms of transitivity, as well as the syntactic frames that can be used as a linguistic test.

The syntactic classification of verbs is not a target in itself; however, the semantic classification of verbs can benefit from syntactic tests to include or exclude verbs within a certain verb class as well as to determine the valid nominal derivatives of each class of verbs. An investigation of the link between the syntactic behaviour of verbs and their semantics can help in establishing semantic classes of verbs.

The semantic classification of verbs can also be informed by the notion of thematic roles. Initially, the notion of thematic role was introduced by Fillmore (1968), in which the relationship between the verbs and their arguments is labelled semantically. Fillmore (1971: 42) suggests nine types of thematic roles: agent, experiencer, instrument, object,

source, goal, location, time and path. These thematic roles can be utilized to describe the syntactic frame components of a class of verbs, and to test the semantic similarities between the arguments of a class of verbs to examine if the verbs of a semantic class can be alternatively used in specific syntactic frames.

2.8 The literature of computational linguistics on Arabic morphology

The field of computational linguistics is an outcome of the combination of language science and computer science. Morphology is fundamental in working on Arabic computational applications in view of its essential interactions with both orthography and syntax. The computational morphology techniques are not an end in themselves. They are the basis of other natural language processing (NLP) systems as they are very useful for many NLP applications, such as part-of-speech (POS) tagging, information retrieval (IR), dictionary automation, root extraction, text compression, data encryption, automatic diacritization, machine translation (MT), and automatic speech recognition (ASR) (Habash, 2010).

Most computational treatments of Arabic morphology have focused on morphological analysis rather than morphological generation. The most referenced works were done, in the form of morphological analyzers, by Beesley (1990, 1996, 2001), Buckwalter (2004), and Smrž (2007). In addition, there are more recent works such as Altantawy et al.'s (2010) and Gridach and Chenfour's (2011). In an early attempt, Beesley (1990) provides a description of a system that analyzes Arabic words based on Koskenniemi's (1983) two-level morphology. For example, the root k-t-b ---- 'to write' can be presented as ?*k?*t?*b?* which can be represented with the pattern *CaCaC* as *katab* 'he wrote'.

In addition, Beesley provides an important system for Arabic morphology, based on finitestate technology, called the Xerox Arabic Morphological Analyzer (XAMA) (Beesley 1996, 2001). This system contains 5,000 consonantal roots and 400 morphological patterns, with a capacity to generate up to 90,000 stems. This system offers a number of advantages: firstly, a capacity of huge coverage; secondly, it is based on rules and also provides an English glossary for each word. However, this system suffers from problems such as the overgeneration of word derivation, i.e. the derivation of non-existant words of the classical Arabic dictionary (Darwish 2002). One of the most referenced analyzers in the literature is Buckwalter's Arabic Morphological Analyzer (BAMA) (Buckwalter 2004). It consists of three parts (i) a dictionary, (ii) compatibility tables, and (iii) an analysis engine. This analyzer employs an interesting approach where the orthographic and morphotactics rules are designed into the lexicon rather than in general rules which interact to identify the output. This approach is described as concatenative lexicon-driven. The system comprises 38,600 lemmas which are used in different systems such as the Penn Arabic Treebank, the LDC Arabic POS-tagger, and the Prague Arabic Dependency Treebank. It was designed as a word-form database that interacts with concatenative databases in which words are processed separately, where the stem of each word is taken as the base form, and presents other information about the root (Attia, 2008).

Smrž (2007) provided a computational model of the Arabic inflectional morphological system seeking the working interface between morphology and syntax. He presented a morphological Arabic functional theory. He also developed the ElixirFM system that acts at a highly functional level as well as being an interactive implementation. The system is written in the Haskell programming language. According to Smrž, this NLP system is able to generate and inflect full words; it analyzes the word-form structure which identifies its grammatical functions. Furthermore, Smrž's work is concerned with the Arabic inflectional aspect of Arabic morphology.

Recently, Altantawy et al. (2010) presented a Morphological Analyzer and Generator for Arabic Dialects (MAGEAD). This system identifies general rules of the orthographic and morphophonemic scheme in order to derive allomorphs. The lexicon of this system was developed using the Elixir-FM's lexicon (Smrž 2007). This analyzer was helpful in processing word-forms depending on dialects' morphology. However, it still needs a comprehensive lexicon to deal with Arabic dialects.

More recently, Gridach and Chenfour (2011) presented a new approach to Arabic morphological analysis. It relies on the Arabic Morphological Automaton (AMAUT), which makes it faster and more efficient. Gridach and Chenfour propose a technique which uses the XMODEL language depending on the realization of morphological databases. Since Arabic morphology represents a special type of morphological system, it is based on a schema that represents Arabic words, in an attempt to improve Arabic morphological

automata. This approach has a development standardization feature, exploiting NLP systems such as semantic and syntactic analysis, orthographical correction, information retrieval and machine translation.

In the computational linguistics literature, many researchers believe that the overgeneration problem still requires linguistic solutions to be solved computationally. Darwish (2002) argues that one of the problems in the Xerox Arabic Morphological Analyzer (Beesley 1996, 2001) is overgeneration. He defines overgeneration in this case as the production of words that do not exist, or could not exist in traditional Arabic dictionaries.

Furthermore, Dichy and Farghaly (2007) point out that there is a problem resulting from considering morphological patterns (including the nominal derivative patterns) applicable to any verb or root to generate word forms. According to Dichy and Farghaly (2007) and the author's knowledge, no attempt has hitherto been made to provide linguistic solutions to determine which patterns can be applied to particular roots or verbs.

With respect to morphological generation, El-Affendi (1999) argues that there are many types of generation algorithms for Arabic. Some are classified as sliding window algorithms that use a matching approach to input Arabic words against roots' lists, prefixes, suffixes and morphological patterns. Others are algebraic algorithms (El-Affendi 1991) which use binary values in considering the morphological patterns of given words and then decompose these words into stem, prefix and suffix. Al-Shalabi and Evens (1998) developed permutation algorithms that operate on the word's letters in order to produce all possible sequences of triliteral or quadriliteral features without changing the original order of the given letters, and then compare them against the roots' lists. Furthermore, linguistic algorithms extract letters from a given word that is related to a list of prefixes and suffixes, then place the rest of the word into a list and test these components in order to find a match with a dictionary of morphological patterns that determine the stem form of a given word (Thalouth and Al-Dannan, 1990; Yagi and Harous, 2003).

Chapter Three

Research Methodology

3.1 Introduction

This chapter discusses the research methodology employed in the study. It highlights the research topic and research questions, and then details the data collection, data analysis framework, and research methods used. It describes our Arabic verb classification, involving semantic arguments and syntactic tests. The multi-level analysis of the nominal derivatives, involving semantics, morphology, and prosodic phonology is presented. Finally, the computational application for the nominal derivative system and the overgeneration problem are described.

3.2 Background to the study

The research relies largely on analytical descriptive methods throughout. The linguistic data consists of two parts, Form I verbs and their nominal derivatives, which are treated as inputs and outputs respectively. The nominal derivatives are divided into six types as follows:

- 1) The active participle (agent noun) اسم الفاعل ism al-fā °il,
- 2) The passive participle اسم المفعول ism al-maf^cūl,
- 3) The form of exaggeration حديغة المبالغة sīgat al-mubālagah,
- 4) The instrumental noun السم الألة ism al-³ālah,
- 5) The qualificative adjective الصفة المشبهة al-șifat al-mušabbahah,
- 6) The locative noun اسم المكان ism al-makān.

There are three stages of analysis. The first is the collection and classification of verbs; the second is the multi-level analysis which involves morphology, prosodic phonology, and semantics; and the third stage involves a computational application for the nominal derivatives. This computational application prevents the overgeneration problem, by which valid nominal derivatives (output) are generated from their Form I verbs (input).

It is worth recalling here the aim of this study, to address the following research question and sub-questions:

- What are the characteristics of the formation of nominal derivatives in the Arabic language semantically, morphologically and phonologically (prosodically)?
- What are the semantic restrictions of the formation of valid nominal derivatives? Why can nominal derivatives be formed from some verbs but not from others? From a computational linguistic perspective, the question is, how to prevent overgeneration in deriving nominal derivatives?
- What are the morphological characteristics of nominal derivatives in terms of morphological processes, and the input of their formation (is it the root, the stem, or both?)?
- What is the prosodic structure of the nominal derivatives?

3.3 Data collection

The first stage in the research involved collecting the data, consisting of Form I verbs, regarded as the basis of the whole verbal derivation system in Arabic. For this stage of research, I took data from contemporary Modern Standard Arabic: 980 Form I verbs from *Al-Mawrid* (Al-Ba^calbaki 1995). These represent one fifth of the 5,000 Form I verbs actively used in Modern Standard Arabic. *Al-Mawrid* is a bilingual Arabic-English dictionary, the most commonly used dictionary for English language learners, now in its 22nd edition.

The criteria used to select the verbs are as follows. First, they are all attested in Modern Standard Arabic today and listed in the current edition of *al-Mawrid*. Second, they represent identified events, activities, circumstances or states. Third, verbs within a specific semantic class correspond in terms of the nature of the subject, the nature of the object, and the nature of any complementary prepositional phrase. Fourth, verbs within a specific semantic class share the same syntactic behaviour in terms of their transitivity and syntactic frames.

3.4 Our classification of Arabic verbs

Each list of verbs selected from *Al-Mawrid* shares related semantic features, and they occur in a certain syntactic frame. The lists are used to establish 44 semantic verb classes. Each class was described in accordance with the semantic features of the collected verbs. The description of each class led to linking the 44 established classes with Levin's (1993) and Vendler's (1969) verb classes. The 44 class titles are mostly adapted from Levin's (1993) classes for English. However, the classes have their own descriptors, members, and syntactic criteria in a way that suits the peculiarities of Arabic.

The members of each class share a certain semantic function which expresses a certain meaning, such as motions, emotions, colours, and bodily qualities. In addition, the semantic relations play an important role in classifying and choosing the collected verbs. Synonyms, antonyms, polysemy, and hyponym are utilised in establishing the verb classes. For example, the verb 'to love' and the verb $e^{\frac{1}{2}}$ wadd 'to like' are synonyms, so both verbs fall within the same class (verbs of emotions). As an example of the antonymic relation, the verb karih $\approx \frac{1}{2}$ 'to hate' and the verb 'to love' fall into the same class.

Syntactically, each class of verb was examined according to two criteria. The first criterion examined whether the verb is transitive or intransitive. After that, a syntactic frame was tested for each class of verbs, within which the verbs of a given class can be used alternately. These syntactic frames are governed by three major components:

- The nature of the subject
- The nature of the object (if available) 6
- The nature of the complementary prepositional phrase (if available).

In describing and determining the nature of each component to determine whether to include or exclude verbs within a certain class, the current study benefits from the ontology of entities presented by Helbig (2006: 410). Specifically, there are two types of entity:

• Concrete object entity, expressing touchable and tangible objects that occupy physical space, and can be animate or inanimate. An animate object has life, spirit and agency, while an inanimate objects does not.

^{6.} The object here stands for the direct object

• Abstract object entity, expressing non-physical, untouchable and intangible concepts.

3.5 Multi-level analysis of the nominal derivatives

The second stage of this research started with manually deriving the valid nominal derivatives (six types) from their Form I verb inputs. Excel databases were created for the verb classes and their valid nominal derivatives. At this stage of the research, it is clear that some verb classes do not allow certain nominal derivatives (see section 5.3 and 5.4). Table 7 includes brief definitions of the nominal derivatives accompanied by examples.

The nominal derivatives	Definitions	Examples
	In Arabic ism al-fā cil اسم الفاعل (agent noun), a	
The active participle	derived verbal noun from a verb indicates the	reader' قارئ 'reader'
	doer of the action or the state of a verb	1
The passive participle	In Arabic ism al-maf ^c ūl المفعول اسم, a derived	maksiin
	verbal noun from a verb indicates the object	'hroken'
	of the action	broken
The form of	In Arabic <i>sīġat al-mubālaġah</i> صيغة المبالغة, a	
exaggeration	derived verbal noun from a verb referring to	<i>xabbāz</i> خبّاز 'baker'
	intensity or regularity of action	
The instrumental noun	In Arabic ism al-'ālah, a derived verbal noun	
	referring to a tool, machine, or device by	'a key' مفتاح <i>miftāḥ</i>
	which the action of the verb occurs	
The qualificative	الصفة In Arabic al-șifah al-mušabbahah	
adjective	المشبهة, an adjective, which functions as a	'alīm alc'knowing'
	noun, describes an agent who does the action	Kilowing
	inherently	

Table 7: Definitions of nominal derivatives

The locative noun	In Arabic <i>ism al-makān</i> , a derived verbal noun referring to a place in which the activity assigned by the verb happens	<i>maṣna^c</i> مصنع 'factory'

At the morphological level, this study analyzes and describes the formation of nominal derivatives, starting with defining and describing each of the six selected types of nominal derivative. At this level, it follows the stem-based approach adopted by Heath (1987); Darden (1992); Bat-El (1994, 2001); McOmber (1995); Ratcliffe (1998); Gafos (1999); Ussishkin (1999, 2000, 2003, 2005). By adopting this approach, the stem of a nominal derivative (the output) is analyzed as derived from the stem of Form I verbs (the input).

The rationale for adopting the stem-based approach is that formation of the nominal derivatives relies entirely on the verbal stem rather than the root. To derive some nominal derivatives (such as the instrumental noun, the locative noun, and the qualificative adjective), details are required that can only be taken from the stem, not from the root. Here are three pieces of evidence in favour of the stem-based approach:

- (i) Some nominal derivatives can be or cannot be derived based on grammatical features of the stem, such as syntactic features transitive vs. intransitive. There are two cases here:
 - The instrumental noun can only be derived from a transitive verb (Form I stem), such as the instrumental noun *miftāh* مفتاح 'key' derived from the stem (Form I verb) *fatah* فتتح 'to open'. Reference to the root is insufficient.
 - The qualificative adjective can only be derived from an intransitive verb, such as the qualificative adjective *jamīl* جميل 'beautiful' derived from the stem (Form I verb) *jamula* 'to be beautiful'. Reference to the root is insufficient.
- (ii) The formation of some nominal derivatives depends on the vocalic melody of the stem (Form I verb). The locative noun has two templates $maC_1C_2aC_3$ and $maC_1C_2iC_3$. The default template is $maC_1C_2aC_3$; where the Form I verb takes the stem vowel -i- in the imperfective, however, $(yaC_1C_2iC_3)$, such as *jalas yajlis yajlis* 'to sit', the locative noun takes the form maCCiC, carrying over /i/ from the imperfect stem, as in:

majlis مجلس 'council, seat'. The root essentially lacks a vocalic melody, therefore, reference to the root would be insufficient in the derivation of locative nouns.

Furthermore, the transitivity of a Form I verb may be reflected in the vocalic melody. Where the vocalic melody of the Form I verb is /u/ in the perfective and imperfective, then the verb is predicted to be intransitive. This is useful in the formation of qualificative adjectives which are only derived from intransitive verbs.

(iii) The nominal derivatives have different templatic patterns based on the verb stem from which they are derived. In Arabic, there are fifteen verb forms, of which ten are common in Modern Standard Arabic. Consider Table 8 where the root *k-t-b* is associated with eight verbal stems, and since no single root associates with all verbal forms, the root *d-k-r* for Form V and *s-w-d* for Form IX. Here we see that each verbal form has its unique nominal derivative templates for the active and passive participles (see table 8) based on the verbal stem, and for all the derived verb forms (II - X), the imperfective verbal stem is the base for the active participle, the perfective verbal stem the base for the passive participle. Derivation of the active participle according to the root-based approach, by contrast, would involve a combination of a root with any of the ten templatic patterns of the active participle; thus, considering the stem reduces the number of nominal derivatives that should be taken into account.

Root	Ve	erbal stem	Active	Template	Passive	Template
			Participle		Participle	
k-t-b	Ι	katab	kātib	$C_1 \bar{a} C_2 i C_3$	maktūb	$maC_1C_2\bar{u}C_3$
k-t-b	II	kattab	mukattib	$muC_1aC_2C_2iC_3$	mukattab	$muC_1aC_2C_2aC_3$
k-t-b	III	kātab	mukātib	$muC_1\bar{a}C_2iC_3$	mukātab	$muC_1\bar{a}C_2aC_3$
k-t-b	IV	°aktab	mu²aktib	$mu^{\circ}aC_{1}C_{2}iC_{3}$	mu²aktab	$mu^{\circ}aC_{1}C_{2}aC_{3}$
₫-k-r	V	ta <u>d</u> akkar	muta <u>d</u> akkir	$mutaC_1aC_2C_2aC_3$	muta <u>d</u> akkar	$mutaC_1aC_2C_2aC_3$
k-t-b	VI	takātab	mutakātib	$mutaC_1\bar{a}C_2iC_3$	mutakātab	$mutaC_1\bar{a}C_2a C_3$
k-t-b	VII	°inkatab	munkatib	mun $C_1 a C_2 i C_3$	munkatab	$munC_1aC_2aC_3$
k-t-b	VIII	°iktatab	muktatib	$muC_1C_2aC_2iC_3$	muktatab	$muC_1C_2aC_2aC_3$
s-w-d	IX	°iswadd	muswadid	$muC_1C_2aC_3iC_3$	muswadad	$muC_1C_2aC_3aC_3$
k-t-b	Х	°istaktab	mustaktib	$mustaC_1C_2iC_3$	mustaktab	$mustaC_1C_2aC_3$

Table 8: The verb Forms and their active and passive participles

At the prosodic level, the current study presents a templatic representation of the nominal derivatives. Following Watson (2002) and McCarthy and Prince (1986, 1990a, 1990b, 1993a, 1995, 1999), prosodic analysis describes the word formation processes and

prosodic strictures of the nominal derivatives, using diagrams (prosodic trees) to illustrate the stages of derivation of the nominal derivatives from their Form I verbs. The prosodic representation of the input stem (Form I verb) is presented, followed by the illustration of the templatic changes that take place to form the output stem of a nominal derivative. The prosodic structure of the templatic patterns of the nominal derivatives is analyzed in terms of these levels: syllabic, moraic and foot levels.

At the semantic level, the current study provides a semantic description of the nominal derivatives in accordance with the thematic roles that were introduced by Fillmore (1968), in which the relationship between the verbs and their arguments is labelled semantically. Fillmore (1971: 42) suggests nine types of thematic role: agent, experiencer, instrument, object, source, goal, location, time and path.

To determine the restrictions/constraints of deriving valid nominal derivatives, this study adapts Helbig (2006) model of the ontology of entities. Two kinds of entities (semantic features) are taken from Helbig (2006) model, namely: the object entity and the situation entity. The object entity is classified into two categories: abstract entities and concrete entities. The situation entity is classified into two categories: state entities and event entities. Each type of nominal derivatives will be described in terms of the object entity whether it refers to concrete entities or abstract entities, and will be described in terms of the situation entity whether it refers to state entities or event entities. In the same way, each class of the 44 verb classes will be described in terms of the object entity whether they refer to state entities or event entities. Our criteria to detairmine either the verb classes and nominal derivatives refer to concrete or abstract entities as well as refer to state or event entities are detailed in Table 9 below:

Semantic features	Verb examples	Nominal derivative examples
Concrete entity refers to an	- to go	- key
action, event or state which		6 7
	- to sit	- office
is accessible to one or more		
	- to hit	- nlough
of the five senses, i.e. the	- to int	- plough
entity can be seen, heard,		

Table 9: The semantic features of the object entities and the situation entities

touched, smelt, or tasted. It		
expresses tangible meanings		
that occupy physical space		
Abstract entity refers to a	- to think	- happy
concept, attribute, quality or	- to love	- free (liberated)
to any of the five senses. It	- to dream	- generous
expresses non-physical,		
unseen and intangible		
meanings		
State entity refers to a state	- to know	- nice
which is either permanent or	- to hate	- green
length of time	- to like	- noble
Event entity refers to a	- to open	- playground
change from one state to	- to cut	- swiming pool
another	- to dig	- mixer

To determine the validity or invalidity to derive a nominal derivative from a certain verb, this study relies on identifing compatibility between:

- the object entity (concrete or abstract) and the situation entity (state or event) of the type of nominal derivatives. For example, the instrumental noun *miftāḥ* مفتاح 'key' refers to a concrete object entity and an event situation entity.
- the object entity (concrete or abstract) and the situation entity (state or event) of the class of verbs. For example, the class of emotion verbs refers to an abstract object entity and a state situation entity.

If both a class of verbs and a type of nominal derivative share the same entities, then the nominal derivative is valid to be derived from that class of verbs. For example, the

instrumental noun refers to a concrete object entity and an event situation entity; therefore, the verb from which we can derive the instrumental noun must also refer to concrete (object entity) and event (situation entity), such as the class of verbs of combining and constructing. By contrast, we cannot derive an instrumental noun from a verb involving abstract and state entities, such as *karih* $\geq \zeta$ 'to hate'. Another example is the qualificative adjective, that involves an abstract or a concrete (object entity) and a state (situation entity), so the verb from which it can be derived must involve an abstract or a concrete (object entity) and a state (situation entity), such as *karih* $\geq \zeta$ 'to hate'. whose qualificative adjective is *karīh* $\geq \zeta$ 'distasteful' (see sections 5.3 and 5.4).

3.6 Computational application for the nominal derivatives

The computational part of the current study is completely based on the outcome of the findings of the lingusitic analysis (see section 6.5). The aim of the presented application is to retrieve the generated nominal derivatives of Form I verbs. The application is designed using Microsoft Access 2007/2010. The input of the application is Form I verbs which amount to 980 verbs. The application consists of two parts, a retriever and an applicability checker. By inputting a triliteral verb (Form I), the first part executes the querying process to retrieve the required data from the database. The output data can be classified into four types:

- General information, including: the transliteration and the title of the class of verb.
- Syntactic information, including: the imperfective form and transitivity of the verb.
- Semantic information, including the features of the verb (concrete/abstract and state/event) and English meaning.
- Nominal derivatives:

- the active participle	e - the	passive	participle
-------------------------	---------	---------	------------

- the form of exaggeration
- the qualificative adjective
- the noun of instance
- the locative noun
- the noun of manner⁷
- the mīmī maṣdar

- the instrumental noun

^{7.} The noun of instance, the *mīmī maṣdar*, and the noun of manner are nominal derivatives that can be generated validly from any Form I verb. They are defined as follows:

⁻ The noun of instance is called in Arabic *ism al-marrah*, a derived verbal noun referring to a single occurrence of the action, such as *raqsa* رقصة 'a dance'.

⁻ The *mimi maşdar* is a derived verbal noun that starts with the initial consonant $m\bar{n}m$ as a prefix, such as maxad a defect'.

An evaluation will be conducted to measure quantitatively the accuracy of generating valid nominal derivatives in order to determine the extent to which the overgeneration can be avoided (see section 6.6).

⁻ The noun of manner is called in Arabic *ism al-hay'ah*, a derived noun describing the state, condition, or manner in which the action of a verb is accomplished or happens, such as *wiqfah* وقفة 'a stand-up position'.

Chapter Four

Arabic verb classification

4.1 Introduction

This chapter presents a classification of Arabic verbs. The selected 980 Form I verbs are classified into 44 classes (see sections 3.3 and 3.4). The verbs are grouped in each class according to common semantic features as well as common syntactic behaviour. Table 10 shows the titles of our classes and their equivalent classes according to Levin and Vendler:

#	Our Arabic verb classes	Levin's classes	Vendler's classes
01	Verbs of the state of the body	Verbs involving the body	State verbs
02	Verbs of body parts	Verbs involving the body	Accomplishment verbs
03	Verbs of diseases	Verbs involving the body	State verbs
04	Verbs of social interaction	Verbs of social interaction	State verbs
05	Verbs of emotions	Psych-verbs (verbs of psychological state)	State verbs
06	Verbs of colouring	Verbs of colouring	State verbs
07	Verbs of bodily qualities	Verbs of change of bodily state	State verbs
08	Verbs of motion	Verbs of motion	Activity verbs
09	Verbs of swimming	Verbs of motion	Activity verbs
10	Verbs of locations/places	Verbs of entity-specific modes of being / Measure verbs	State verbs
11	Verbs of violence and abuse	Hurt verbs / Verbs of killing	Accomplishment verbs
12	Verbs of mental process	Verbs of perception	State verbs
13	Verbs of financial transactions	Cost verbs / Price verbs / Bill verbs	Activity verbs
14	Verbs of agriculture	Grow verbs	Activity verbs
15	Verbs of desire and request	Verbs of desire	State verbs
16	Verbs of intention	Verbs of future having	State verbs
17	Verbs of combining and constructing	Verbs of combining and attaching / Build verbs	Achievement verbs
18	Verbs of sending and carrying	Verbs of sending and carrying	Achievement verbs
19	Verbs of separating and disassembling	Verbs of separating and disassembling	Accomplishment verbs
20	Verbs of removing	Verbs of removing	Accomplishment verbs
21	Verbs of bending	Bend verbs	Accomplishment verbs
22	Verbs of decorating and transcribing	Image creation verbs / Scribble verbs	Achievement verbs
23	Verbs of measurement	Measure verbs	Achievement verbs
24	Verbs of quantity and size	Measure verbs	State verbs
25	Verbs of stability	Lodge verbs	State verbs
26	Verbs of creation	Create verbs	Achievement verbs
27	Verbs of preparing	Verbs of preparing / Cooking verbs	Activity verbs
28	Verbs of ingesting	Verbs of ingesting	Achievement verbs
29	Verbs of the five senses	Verbs of perception	State verbs
30	Verbs of ruling and government	Judgment verbs	State verbs
31	Verbs of the development of life	Verbs of existence	Accomplishment verbs

Table 10: Arabic verb classes and their Levin and Vendler's equivalents

32	Verbs of uttering	Say verbs	Activity verbs
33	Verbs of accepting		State verbs
34	Verbs of refusing and disobedience		State verbs
35	Verbs of preventing and prohibition	Keep verbs	Achievement verbs
36	Verbs of occurrence and progressing	Verbs of occurrence	State verbs
37	Verb of appearance	Verbs of appearance	Accomplishment verbs
38	Verbs of disappearing and ending	Verbs of disappearance	Accomplishment verbs
39	Verbs of bodily movements	Curtsey verbs	Achievement verbs
40	Verbs of taking and giving	Bring and take verbs / give verbs	Accomplishment verbs
41	Verbs of human sounds	Verbs of manner of speaking	State verbs
42	Verbs of sounds made by animals	Verbs of sounds made by animals	State verbs
43	Verbs of bodily care	Verbs of grooming and bodily care / Verbs of dressing	Achievement verbs
44	Verbs of winning		State verbs

4.2 Arabic verb classes

Each of our 44 classes is presented in a table below, that includes: the title of the class, class description, correspondence with Levin's class, correspondence with Vendler's class, transitivity test, class members with their meaning in English, syntactic frame of transitives and intransitives, nature of the subject, nature of the object (if available), nature of the complementary prepositional phrase (if available), and a sentence example of each verb listed.

Class	These verbs describe a state that affects an animate being (organism)	
Description	psychologically or physically.	
Levin's class	Verbs involving the body	
	Verbs of body-internal states of existence	
Vendler's class	State verbs	
Transitive or	Intransitive	
intransitive		
Class members	1. <i>°ariq</i> أرق 'to be sleepless; suffer insomnia'	
	2. <i>°alim</i> ألم 'to feel pain'	
	3. <i>bari</i> ² برى 'to recover'	
	4. <i>ta ° ib</i> 'io be or become tired'	
	5. <u>tamil</u> ثمِل 'to be (become, get) drunk'	
	6. $j\bar{a}^{c} \neq 3$ 'to be or become hungry'	
	7. <i>habil(t)</i> خبلت 'to be or become pregnant'	
	8. xadir خدر 'to be or become numb, benumbed'	
	9. xawā خوى 'to be or become hungry or empty, feel hungry'	
	10. $d\bar{a}x$ ذاخ 'to be or become dizzy'	
	11. rawiy رَوِيَ 'to quench one's thrist'	
	12. sakur سکر 'to be or become drunk(en), intoxicated, inebriated'	
	13. sahid سهد 'to be sleepless, suffer insomnia'	
	14. $šax$ شاخ to age grow old'	
	15. $\check{s}abi^{\circ}$ \check{t} to eat one's fill satisfy one's appetite'	
	$16 \ damar$ $into be or become atrophic: to be or become thin'$	
	17 zami ² to be thirsty feel thirsty suffer thirst?	
	$18 \ caria$ (b) weat'	
	$10. c_{atis}$ where 'to be thirsty feel thirsty suffer thirst'	
	20 $\dot{\sigma}_{ativ}$ $\dot{\sigma}_{c}$ 'to nauseate feel nausea be nauseated feel sick'	
	20. guiry in industate, reel nausea, be nauseated, reel sick	
	$21. jujum \rightarrow 10 \text{ weall}$	
	$22. \text{ kabir } \neq 0$ grow old, get old(er)	
	to be or become tired, fatigued, exhausted	
	24. <i>na^cas</i> نعس 'to be (or feel) sleepy, drowsy'	

4.2.1 Verbs of the state of the body (class 01)

	to be in childbed, be confined' نفِسَرات المرْأة) (25. nafisa(t)		
	26. harim هرم 'to age, grow old; to become decrepit'.		
The syntactic	Verb + Subject (+ the particle of cause $(fa-)$ + subordinate clause		
intransitives	(lesultative))		
The nature of the subject	The subject is human or animal and involves a change in state to the whole body.		
The nature of the subordinate clause (resultative)	This syntactic frame involves cause a in the subordinate clause by <i>fa</i> - is ca main clause.	nd effect, where the effect introduced used by the action of the verb in the	
Examples	 أرق محمدٌ فذهب إلى النوم ألم محمدٌ فذهب إلى الطبيب ألم محمدٌ فذهب إلى الطبيب بَرئ محمدٌ فذهب إلى الطبيب تعب محمدٌ فذهب عقله تعب محمدٌ فذهب إلى الطبيب جاع محمدٌ فذهب إلى الطبيب جاع محمدٌ فذهب إلى الطبيب حلتُ سارة فذهب إلى الطبيب حوى محمدٌ فذهب إلى الطبيب داخ محمدٌ فذهب إلى الطبيب داخ محمدٌ فذهب عقله داخ محمدٌ فذهب إلى الطبيب داخ محمدٌ فذهب الى الطبيب 	14. شاخ محمدٌ فذهب شبابه 15. شبع محمدٌ فذهب الجوع 16. ضمرتْ يد محمد فذهب إلى الطبيب 17. ظمئ محمدٌ فشرب الماء 18. عرق محمدٌ فتعب جسمه 19. عطش محمدٌ فتعب جسمه 20. غثي محمدٌ فذهب إلى الطبيب 21. كبر محمدٌ فذهب إلى الطبيب 23. كلّ محمدٌ فذهب إلى الطبيب 24. نفِسَت سارة فذهب إلى الطبيب 26. هرم محمدٌ فذهب إلى الطبيب	

4.2.2 Verbs of body parts (Class 02)

Class	Verbs denote movement involving one or more parts of the body of an	
Description	animate being.	
Levin's class	Verbs involving the body	
	Verbs of body-internal states of existence	
Vendler's class	Accomplishment verbs	
Transitive or	Intransitive	
intransitive		
Class members	1. <i>bakā</i> بکی 'to cry, weep, shed tears'	
	2. xalaj(t) (خلج (ت العين 'to twitch'	
	3. dama ^c دمع 'to water, tear, fill with tears, shed tears'	
	4. <i>daraf</i> ذرف 'to shed tears, weep, cry'	
	5. ra ^c af رعف 'to have a nosebleed'	
	6. <i>ramaš</i> رمش 'to blink, wink, bat'	
	7. <i>zafar</i> زفر 'to exhale, breathe out'	
	8. sa ^c al سعل 'to cough'	
	9. <i>šaxar</i> شخر 'to snort'	
	10. <i>šariq</i> شرق 'to choke'	
	11. šahaq شبعقَ 'to inhale, breathe in'	
	12. <i>cabis</i> عبس 'to frown, knit or contract the brows'	

	13. <i>°arij</i> عرج 'to limp, hobble, walk lamely, be lame'		
	14. <i>°atas</i> عطَس 'to sneeze'		
	15. ġ <i>amaz</i> غمَز 'to wink (at)'		
	16. latig للنغ 'to lisp'		
	17. lahit لهت 'to pant'		
	18. nafat نفث 'to expectorate, cough out, spit (out)'.		
The syntactic	Verb + Subject		
frame of			
intransitives			
The nature of	The subject is a specific body part of an animate being, or an animate		
the subject	being where the part affected is a specific part of the body, such as		
	eye, nose, mouth, hand, lungs, leg, brain or heart.		
examples	 بكى محمد 	11. شەق محمد	
	 خلجت عین محمد 	12. عبس محمد	
	3. دمع محمد	13. عرج محمد	
	4. ذرف محمد	14. عطس محمد	
	5. رعف محمد	15. غمَز محمد	
	6. رمش محمد	16. لثغ محمد	
	7. زفر محمد	17. لهٹ محمد	
	8۔ سعل محمد	18. نفث محمد	
	9. شخر محمد		
	10. شرق محمد		

4.2.3 Verbs of diseases (Class 03)

Class	These verbs denote a disease or syndrome that affects the whole or a part of		
Description	the animate being.		
Levin's class	Verbs involving the body		
	Verbs of body-internal states of existence		
Vendler's class	State verbs		
Transitive or intransitive	Intransitive		
Class members	 bariş برص 'to be or become a leper; to be or become leprous' bakim بَجْر 'to be or become dumb' jadir بَجْر 'to have smallpox' jadir بجز 'to have leprosy' jarib بجزب 'to be or become mangy; to be or become scabby' xabil بحزب 'to be or become mentally deranged' xaris خرب 'to be or become mute, dumb, speechless' xarif نحزب 'to be in one's dotage, become a dotard, become senile' ramid بحز 'to be or become sick, ill, ailing' salum 'to be or become sick, ill, ailing' sull 'to be or become paralyzed, paralytic' full taris 'to be or become deaf' 'to be or become deaf' 		

	to be sterile, barren عقم 16. ° agum	,	
	17. <i>camiš</i> عمش 'to be or become blear(y)'		
	18. <i>camiv</i> عمر 'to be or become blind, lose one's sight'		
	19. <i>aarih z</i> 'to ulcerate ulcer fester: to canker'		
	20 <i>aamil</i> في to be lice-infested lousy'		
	21. kasih کست to be crippled beco	me a cripple. to be rickety'	
	21 marid (2) to be or become sick?		
	22: marine 22 : name 23 nazaf 23 is bleed hemorrhage lose (much) blood?		
		, 1000 (Inden) 0100d .	
The syntactic	Verb + Subject (+ the particle of cause (<i>fa</i> -) + subordinate clause		
frame of	(resultative))		
intransitives			
The nature of	The subject is an animate being.		
the subject			
The nature of	This syntactic frame involves a causative structure, where the complement is		
the	a result caused by the subject. It starts with the particle of cause (fa-)		
subordinate	followed by perfect verb that involves a medical treatment.		
clause			
(resultative)			
Examples	 برص محمد فذهب إلى الطبيب 	13. شلَّ محمد فذهب إلى الطبيب	
	 بكم محمد فذهب إلى الطبيب 	14. طرِش محمد فذهب إلى الطبيب	
	 جدر محمد فذهب إلى الطبيب 	15. عقّر محمد فذهب إلى الطبيب	
	4. جذم محمد فذهب إلى الطبيب	16. عقم محمد فذهب إلى الطبيب	
	 جرب محمد فذهب إلى الطبيب 	17. عمِش محمد فذهب إلى الطبيب	
	 خبل محمد فذهب إلى الطبيب 	18. عمِيَ محمد فذهب إلى الطبيب	
	 خرس محمد فذهب إلى الطبيب 	19. قرح محمد فذهب إلى الطبيب	
	 ٤. خرف محمد فذهب إلى الطبيب 	20. قمل محمد فذهب إلى الطبيب	
	 رمد محمد فذهب إلى الطبيب 	21. كسِح محمد فذهب إلى الطبيب	
	10. زكم محمد فذهب إلى الطبيب	22. مرِضَ محمد فذهب إلى الطبيب	
	11. سقم محمد فذهب إلى الطبيب	23. نزَف محمد فذهب إلى الطبيب	
	12. سُلَّ محمد فذهب إلى الطبيب		

4.2.4 Verbs of social and personal behaviour (Class 04)

Class	These verbs refer to social interaction between people. They describe the		
Description	personal behaviour of the person in dealing with others.		
Levin's class	Verbs of social interaction		
Vendler's class	State verbs		
Transitive or	Intransitive		
intransitive			
Class members	1. <i>°atir الْزَ</i> ر 'to be or become egocentric, egoist'		
	2. <i>°atim</i> أثر 'to be sinner'		
	3. <i>°adub</i> أَدُب 'to be well-mannered, well-bred, polite'		
	4. baxul بخل 'to be or become stingy, niggardly, miserly parsimonious'		
	5. barr بر 'to be pious, dutiful, devoted, reverent, faithful, kind'		
	6. basul بسل 'to be brave, bold, courageous, fearless, intrepid'		
	7. bašš بش 'to have a cheerful face; to smile, wear a smile, look happy'		

8. <i>bațir</i> بطر 'to be ungrateful'
9. <i>bațal</i> بَطُل 'to be or become unemployed, jobless; to be or become idle'
to be or become luxurious, opulent, affluent' نرف to be or become luxurious'
to wrong, oppress, tyrannise, persecute, aggrieve' جار (على) 'to wrong
12. <i>jabun</i> جبن 'to cower, cringe, quail, shrink in fear, recoil in dread'
13. <i>jalud</i> + 'to be or become patient, tolerant, forbearing'
14. <i>hazum</i> حزم 'to be or become resolute, firm, unwavering, gritty'
to be highborn, wellborn, noble, of noble birth or origin'
16. <i>hašim</i> - شم 'to shame'
to be or become low, base, mean, lowly, vile, despicable'
18. hakum حکّ 'to be or become wise, judicious'
to be or become patient, forbearing, long-suffering, meek'
20. hamid حمد 'to praise commend laud extol eulogize'
21. hamia حمق 'to be or become stupid foolish'
22. $x\bar{a}b$ خاب 'to fail, be unsuccessful, fizzle out, flop, miss the mark'
23. $x\bar{a}r$ \dot{z} to weaken fail languish droop flag slacken'
24. $x\bar{a}n$ خان 'to betray sell out be disloval or faithless to'
25. <i>xabut</i> خنث 'to be or become malicious malevolent vicious wicked'
26. <i>xatal</i> ختل to deceive double-cross cheat defraud dupe trick'
27, xajil 4 27 , is be ashamed of be ashamed to face someone'
28. xada $c \neq i$ 'to deceive fool delude bluff beguile mislead to cheat'
29. <i>xadal</i> خذل let down, fail someone when most needed. disappoint'
30. xaziv $\dot{\epsilon}$ 'to be or become low, lowly, despicable, contemptible'
$31. xasir \leftrightarrow 31$ to lose, forfeit: to suffer or incur a loss'
32. xašav خشي 'to fear, apprehend, dread, be afraid (of)'
33. <i>xatur</i> خطر 'to be or become grave, serious, weighty, momentous'
34. <i>xafir خفر</i> 'to be or become shy, bashful, diffident'
35. xanā خنا 'to use obscene language; to be obscene, ribald, vulgar'
36. xana ° خنع 'to cringe before, truckle before'
37. dajal دجل 'to lie; to quack, play the quack; to be a quack, a charlatan'
38. dall ذلّ 'to be or become low, lowly, humble'
39. <i>damm</i> نزم 'to dispraise, disparage, censure; to vituperate, vilify, slander'
40. dahil ذهل 'to be or become distracted, absentminded'
41. ra ³ af دأف 'to have mercy upon, be merciful toward'
42. rahim رحم 'to have mercy upon, be merciful toward, pity'
43. radul دِنْل 'to be or become low, base, vile, despicable'
44. ra cun رعن 'to be or become lightheaded, frivolous, thoughtless'
45. raġud رغد 'to be or become pleasant, comfortable, easy'
46. rafiq دفق 'to treat with kindness, treat gently'
47. rafuh دفه 'to be or become luxurious, comfortable'
48. zahid زهد 'to abstain (from), to be abstemious'
49. <i>saxir س</i> خِر 'to mock (at)'
to be or become discontented, dissatisfied, indignant' سخط 50. saxat
to be or become silly, absurd, ridiculous' سخف to be or become silly, absurd, ridiculous'
52. saxā سخا 'to be or become generous, liberal, openhanded, freehanded'
53. sarif سرف 'to waste, squander, dissipate, lavish, spend lavishly'
to be or become impudent, insolent, pert, saucy' سلط 'to be or become impudent, insolent, pert, saucy'
to be or become magnanimous, forgiving, tolerant' سمح 55. samuh سمح
56. <i>šaruf</i> شرف 'to be honorable, noble'
57. <i>šarih</i> شره 'to be greedy'

	to complain, lodge (raise, make) a complaint; to complain, nag' شکا 58. šakā			
	59. şabar صبر 'to be patient, forbearing'			
	60. <i>sadaq</i> صدق 'to prove to be true or correct, turn out to be true, come true'			
	to be or become severe '' صرَّ fo be or become severe	e, strict, stern, rigorous, hard, harsh'		
	62. <i>sa^cir</i> صعر 'to be awry, wry (face)'	,,,,,		
	63. safah صفح 'to forgive, pardon, excu	ise, condone, overlook, remit'		
	64. <i>damin</i> ضمن 'to guarantee warrant	ensure secure sponsor youch for'		
	65 <i>tami</i> ^c طمع 'to be or become greedy	covetous avid avaricious'		
	to be or become greed	humorous'		
	$67 c\bar{a}h$ (see to mar disfigure spoil m	ake defective or faulty: to vitiate'		
	68 cadal lie 'to act justly establish in	stice be just be fair be equitable'		
	69 cadal i to hlame reproach twit admonish?			
	70° afglic 'to forgive pardon evoluse condone'			
	70. $a_{ja} = 0$ longive, particular, excuse, conducte 71. $\dot{a}_{a} dir = \dot{b}_{a}^{a}$ (to be trave soll out be disloyed or faithlass to?			
	ر الد الد الد الد المعنون المعنون المعنون المعنون الم			
	72. guss in cheat, swindle	a aandana avarlaak ramit'		
	75. gujur and to longive, pardoli, excus	/3. gafar عور to torgive, pardon, excuse, condone, overlook, remit		
	75 fatan isi 'to fascinate charm ench	ant cantivate infatuate enthrall'		
	75. julun = 10 laselliate, chann, chen	horrid ugly terrible abominable?		
	70. južu 200 to be of become normal	a ribald dirty foul filthy?		
	78 farir is 'to be or become proud h	e, molaid, difty, ioui, muiy		
	$78. jaxii \rightarrow 0$ to be of become proud, if	augily ad) deprayed pervert(ad) immoral'		
	79. jasaa = 10 be of become ugly up	sightly repulsive repugnant'		
	$80. qubun \rightarrow 10$ be of become ugry, un	anonhandad frachandad'		
	$61. karum z_{c} to be generous, noeral,$	alothful?		
	ا معلم المعلم	Sloullul		
	85. uaum = 10 be mean, ignoble, base	henter feel kid (around)?		
	to joke, jest, make run, مرح to joke, jest, make run,	d high minded meananimous'		
	85. <i>Nabul</i> 40 to be noble, noble-innide	a, mgn-mindea, magnanmous		
	to trust, nave confidence وقل 80. wajuq وقل المعالية المعالية المعالية الم	e in, confide in		
The syntestic	87. waau ^o E ⁻¹ to be meek, mild, genu	, peaceable		
The syntactic	verb + Subject (+ Prepositional phrase			
frame of				
intransitives		• • • • • • • • • • • • • • • • • • • •		
The nature of	The subject of these verbs is a human l	being who is characterized by social		
the subject	moral qualities.			
The nature of	The prepositional phrase is headed l	by the preposition fii e 'in'. In this		
the	context, the prepositional phrase expre	sses dealing with people.		
prepositional				
phrase	م و م و م			
Examples	 آير محمد في معاملة الناس 	45. رغد محمد في معاملة الناس		
	2. أتم محمد في معاملة الناس	46. رفق محمد في معاملة الناس		
	3. ادب محمد في معامله الناس	47. رفه محمد في معامله الناس		
	 بخل محمد في معاملة الناس 	48. ز هد محمد في معامله الناس		
	 بر محمد في معاملة الناس 	49. سخر محمد في معاملة الناس		
	6. بسل محمد في معامله الناس	50. سخط محمد في معاملة الناس مع		
	 بش محمد في معاملة الناس 	51. سخف محمد في معامله الناس		
	8. بطر محمد في معامله الناس م أبر محمد في معامله الناس	52. سخي محمد في معامله الناس		
	9. بطل محمد في معامله الناس محمد من معامله الناس	53. سرف محمد في معامله الناس		
	10. ترف محمد في معامله الناس	54. سلط محمد في معامله الناس		
	11 حار محمد في معاملة الناس	55 اسمُح محمد في معامله الناس		

12. جبن محمد في معاملة الناس	56. شرف محمد في معاملة الناس
13. جلد محمد في معاملة الناس	57. شره محمد في معاملة الناس
14. حزم محمد في معاملة الناس	58. شكا محمد في معاملة الناس
15. حسب محمد في معاملة الناس	59. صبر محمد في معاملة الناس
16. حشِم محمد في معاملة الناس	60. صدق محمد في معاملة الناس
17. حقر محمد في معاملة الناس	61. صرُم محمد في معاملة الناس
18. حكم محمد في معاملة الناس	62. صعِر محمد في معاملة الناس
19. حلُّم محمد في معاملة الناس	63. صفح محمد في معاملة الناس
20. حمِد محمد في معاملة الناس	64. ضمن محمد في معاملة الناس
21. حمق محمد في معاملة الناس	65. طمع محمد في معاملة الناس
22. خاب محمد في معاملة الناس	66. ظرُف محمد في معاملة الناس
23. خار محمد في معاملة الناس	67. عاب محمد في معاملة الناس
24. خان محمد في معاملة الناس	68. عدل محمد في معاملة الناس
25. خبث محمد في معاملة الناس	69. عذل محمد في معاملة الناس
26. ختل محمد في معاملة الناس	70. عفا محمد في معاملة الناس
27. خِجْل محمد في معاملة الناس	71. غدر محمد في معاملة الناس
28. خُدع محمد في معاملة الناس	72. غشّ محمد في معاملة الناس
29. خذل محمد في معاملة الناس	73. غفر محمد في معاملة الناس
30. خزي محمد في معاملة الناس	74. غنج محمد في معاملة الناس
31. خسر محمد في معاملة الناس	75. فتن محمد في معاملة الناس
32. خشبي محمد في معاملة الناس	76. فظع محمد في معاملة الناس
33. خطر محمد في معاملة الناس	77. فحش محمد في معاملة الناس
34. خفر محمد في معاملة الناس	78. فخر محمد في معاملة الناس
35. خنا محمد في معاملة الناس	79. فسد محمد في معاملة الناس
36. خنع محمد في معاملة الناس	80. قبح محمد في معاملة الناس
37. دجل محمد في معاملة الناس	81. كرم محمد في معاملة الناس
38. ذلّ محمد في معاملة الناس	82. كسل محمد في معاملة الناس
39. ذمّ محمد في معاملة الناس	83. لؤم محمد في معاملة الناس
40. ذهل محمد في معاملة الناس	84. مزح محمد في معاملة الناس
41. رأف محمد في معاملة الناس	85. نبل محمد في معاملة الناس
42. رحم محمد في معاملة الناس	86. وثق محمد في معاملة الناس
43. رذل محمد في معاملة الناس	87. ودع محمد في معاملة الناس
44. رعن محمد في معاملة الناس	

4.2.5: Verbs of emotions (class 05)

Class Description	These verbs denote that the subject is characterized by subliminal conscious, passionate qualities and feelings, whether they are absolute or restricted in a certain matter.	
Levin's class	Psych-verbs (verbs of psychological state)	
Vendler's class	State verbs	
Transitive or intransitive	Transitive Intransitive	
Class members	 habb حبّ 'to love, be (or fall) in love with, to adore, to like' harij حرج 'to be or become embarrassed' hasad حسد 'to envy, feel envy toward, regard with envy' dahiš دهش 'to be or become astonished' 	

	5. <i>°ašiq</i> عشِق 'to love passionately, adore'
	6. karih کره 'to hate, detest, loathe, abhor, abominate'
	7. maqat مقت 'to detest, abhor, abominate, loathe, hate intensely'
	8. <i>wadd</i> وڏ' to love; to like'
	9. <i>°asif</i> أسف 'to be or become sorry, regretful, sad'
	10. <i>ba³is</i> بئس 'to be or become miserable, wretched, poor, distressed'
	11. barim برم 'to be or become weary (of), bored (with), fed up (with)'
	12. bagad بغض 'to hate, detest, loathe, abhor, abominate'
	13. <i>jazi^o</i> جزع 'to be or become anxious'
	14. <i>hazan</i> حزن 'to be or become sad'
	to bear a grudge (against)' حقد to lear a grudge (against)'
	to yearn for, hanker after, crave, miss' حنّ to yearn for, hanker after, crave, miss'
	to be or become enraged by, furious at, angry with' حنق 17. hanaq حنق
	18. $x\bar{a}f$ خاف' to fear, apprehend, dread; to be or become afraid (of),
	scared (of), alarmed (by), frightened (by), terrified (by)'
	to fear, apprehend, dread, be afraid (of)' خشي 19. <i>xašiy</i> خشي
	20. <i>za ° al</i> زعل 'to huff'
	21. <i>to be or become delighted' طرب</i> 'to be or become delighted'
	22. <i>catab</i> عتب 'to admonish, reprove mildly, reproach gently, blame'
	23. <i>ġaḍib</i> غضب 'to be or become angry'
	24. <i>fazi</i> ^c فزع 'to be or become scared (of), alarmed (by), frightened'
	to be or become worried' ^{فل} ِق to be or become worried
	26. <i>kašir ک</i> شر 'to grin; to grimace'
	to lose one's temper, flame up with, turn into a tiger' نمر 'to lose one's temper, flame up with, turn into a tiger'
	28. rahib رهب 'to fear, dread, apprehend; to be or become afraid'
	29. <i>saʰim</i> سئم 'to be or become bored (with), fed up (with), weary (of)'
	30. <i>dajir</i> ضجر 'to be or become bored (with)'
	31. qarif قرف 'to be disgusted (of), be nauseated (by), be sick (of)'
	32. <i>qazz</i> فز 'to feel disgust (at, for), revolt (at, against), loathe, detest'
	33. <i>ka^oib</i> كنب 'to be or become sad, grieved, depressed'
	to be or become weary (of)' مل 'to be or become weary (of)'
	'to be unhappy, miserable' نکد 'to be unhappy, miserable'
	36. hamm هم 'to worry, trouble, disquiet, upset, make uneasy'
	37. <i>hali ^c</i> to be or become impatient, restless, uneasy, anxious
	38. bahij بهج 'to rejoice (at), jubilate (at); to be happy (at), glad (at)'
	$39. harij \neq 10$ to be (become, get) embarrassed
	40. sarr سر 'to be pleased (with), delighted (at), happy (at), glad (at)
	41. sa $^{\circ}$ to be or become happy, lucky, fortunate
	42. saquy $ue ve ve$
	43. $sa^{\circ}ar$ to reel, sense; to perceive, notice, realize
	44. sugif use to love, adore, be madly in love with
	45. sadam ΔL to shock
	$40. jarin = \sqrt{5}$ to be of become grad at, happy at, pleased with $47. raghib (in x)$ to desire?
	$47.1 ug (110) \longrightarrow 10 uc silc$
The syntactic	+0. num ∞ 10 Ian in 10ve with, be passionately in 10ve with
frame of	$\gamma = 0 + S u 0 J = 0 + O 0 J = 0$
transitives	
The syntactic	Verb + Subject (+ Prepositional phrase)
frame of	

intransitives			
The nature of	The subject of both transitive and intransitive verbs is a human being who is		
the subject	characterized by emotions.		
The nature of	The object is also a human being who can exchange emotions.		
the object			
(transitive			
verbs)			
The nature of the Prepositional phrase (intransitive verbs)	 The subject feels emotions towards the object of the prepositional phrase. The prepositional phrase is headed by one of the following prepositions: <i>c</i> alā <i>c</i> alā <i>c</i> on': with this preposition, the subject is emotionally influenced by what happened to the person referred to within the prepositional phrase. <i>min</i> من 'from': with this preposition, the subject is emotionally influenced by what was done by the person referred to within the prepositional phrase. <i>min</i> من 'from': with this preposition, the subject is emotionally influenced by what was done by the person referred to within the prepositional phrase. <i>c</i> alā <i>c</i> alā <i>c</i> on' or <i>min</i> <i>c</i> from': both prepositions can be used with the class members; however the meaning is different, when with <i>c</i> alā <i>c</i> on' the subject is affected by what happens for the person mentioned in the prepositional phrase, whele with <i>min</i> <i>c</i> from', the person that is mentioned in the preposition affected emotionally on the subject. <i>bi</i> - 'by': with this preposition, the subject is emotionally influenced by the person referred to in the prepositional phrase. 		
	with the person referred to in the	prepositional phrase	
Examples	Examples of transitive verbs	 أ. حَبَّ محمدٌ صديقه 2. حرَج محمدٌ صديقه 3. حسَد محمدٌ صديقه 4. دهِش محمدٌ صديقه 5. عشيق محمدٌ صديقه 6. کره محمدٌ صديقه 7. مقَت محمدٌ صديقه 8. ودّ محمدٌ صديقه 	
	Examples of intransitive verbs with	 أسف محمدٌ على صديقه 	
	the preposition <i>calā على</i> 'on'.	 أ. بأس محمدً على صديقه 1. برم محمدً على صديقه 1. برم محمدً على صديقه 12. بغض محمدً على صديقه 13. جزع محمدً على صديقه 14. جزن محمدً على صديقه 15. حقدً محمدً على صديقه 16. حنَّ محمدً على صديقه 17. حنق محمدً على صديقه 18. طرب محمدً على صديقه 19. عتَب محمدً على صديقه 20. كشر محمدً على صديقه 	
	the preposition <i>min</i> من 'from'.	23. قرير محمدٌ من صديقة 24. ضجر محمدٌ من صديقة 25. قرف محمدٌ من صديقه 26. قرَّ محمدٌ من صديقه	

	27. كَئِب محمدٌ من صديقه 28. ملّ محمدٌ من صديقه 29. نكِد محمدٌ من صديقه
	30. همّ محمدٌ من صديقه
Examples of intransitive verbs with both prepositions <i>calā على</i> 'on' and <i>min</i> من 'from'.	31. خاف محمدٌ على/مِن صديقه 32. خشِيَ محمدٌ على/مِن صديقه 33. زعِل محمدٌ على/مِن صديقه 34. غضِب محمدٌ على/مِن صديقه 35. فزع محمدٌ على/مِن صديقه 36. هلِع محمدٌ على/مِن صديقه 37. هلِع محمدٌ على/مِن صديقه
Examples of intransitive verbs with the preposition <i>bi</i> 'by'.	38. بهج محمدٌ بصديقه 39. حرج محمدٌ بصديقه 40. سرَّ محمدٌ بصديقه 41. سعِدَ محمدٌ بصديقه 43. شعِرَ محمدٌ بصديقه 44. شغِف محمدٌ بصديقه 45. صدَم محمدٌ بصديقه 46. فرح محمدٌ بصديقه
Examples of intransitive verbs with the preposition <i>fī</i> في in'.	47. رغب محمدٌ في صديقه 48. هام محمدٌ في صديقه

4.2.6 Verbs of colouring (class 06)

Class	These verbs denote that the subject is characterized by darkness or lightness,	
Description	or chromatic qualities, whether the subject is an animate or inanimate object.	
Levin's class	Verbs of colouring	
Vendler's class	State verbs	
Transitive or	Intransitive	
intransitive		
Class members	1. baqi ^c بقع 'to be or become spotted, stained, blotted'	
	2. bahit بهت 'to be or become lost brightness'	
	3. <i>hāl</i> 'to be or become bleached, faded, tarnished'	
	4. <i>ḥall</i> خل 'to fade, dim, pale, wan, blanch, tarnish, faint'	
	5. kadib خضب 'to be or become green, verdant'	
	6. <i>xadir</i> خضر 'to be or become green, verdant'	
	7. <i>dakin دک</i> ِن 'to darken or become dark, blackish, dusky'	
	8. <i>dahib ذهب</i> 'to be or become golden'	
	9. <i>zariq</i> زرق 'to be or become blue'	
	10. <i>sahim</i> سجم 'to be or become black'	
	11. samir سمِر 'to be or become brown or tan'	
	to be or become black' سود to be or become black	
	ito be or become pale' شخب to be or become pale'	
	to be or become blond, fair' شقُر to be or become blond, fair'	
	to be or become dyed' صبغ 'to be or become dyed'	

	 16. şahib سهب 'to be or become reddish, russet; blond' عبر 'to be or become dust-coloured, dingy, roan' غبر 'to be or become dark' 18. ġabiš غبر 'to be or become dark' 19. ġamiq غبر 'to be or become bold or dark' 20. faḥum فحم 'to be or become blacken' 21. qatim فتم 'to be or become darken or darkle' 22. kaḥil كحَل 'to be or become blacken with kohl'. 	
The syntactic	Verb + Subject	
intransitives		
The nature of the subject	The subject is a concrete entity that can exhibit colour	
Examples	 1. بقع اللَّوْنُ 2. بهت اللَوْنُ 3. حالَ اللَّوْنُ 4. حلّ اللَّوْنُ 5. خضب اللَّوْنُ 6. خضر اللَّوْنُ 7. دكِن اللَّوْنُ 8. ذَهِبَ اللَّوْنُ 9. زرق اللَّوْنُ 10. سحِم اللَّوْنُ 	12. سود اللَّوْنُ 13. شحَّب اللَّوْنُ 14. شقر اللَّوْنُ 15. صبغ اللَّوْنُ 16. عبر اللَّوْنُ 18. غبِش اللَّوْنُ 20. فحم اللَّوْنُ 21. كتل اللَّوْنُ

4.2.7 Verbs of bodily qualities (class 07)

Class	These verbs refer to a change in the bodily state of an animate or inanimate	
Description	object, including their size, weight, appearance, touch and temperature.	
Levin's class	Verbs of change of bodily state	
Vendler's class	State verbs	
Transitive or intransitive	Intransitive	
Class members	 badun بدن 'to be or become fat, corpulent, obese, stout, plump' barad برد 'to be or become cold; to cool, cool off, cool down, chill' baliy بلي 'to be or become old, worn, shabby, ragged, tattered' <u>taqul</u> نقل 'to be or become heavy' jasum جسم 'to be or become big, large, great, sizable, bulky, gross' ja^c ud جس 'to be or become curled, curly, frizzed, frizzly' jalid بط 'to be or become formed into ice, turn into ice' jamud خف' to be or become forzen, frosted' xašun خف' to be or become coarse, rough, harsh, tough, hard' xaff نفف 'to be or become thin, fine, delicate, slender, tenuous' <u>d</u>ab (خلب (جسمه) 'to be or become melt (away), liquefy, deliquesce, thaw' <u>d</u>ab (خلب 'to be or become pine away, waste away' raqub 'to be or become humid, moist, damp, wet' raqub 'to be or become thin, delicate, fine, slender, tenuous' 	

	to be or become thick' سمُك 'to be or become thick'		
	to be or become fat' corpulent, stout, obese, plump' سمُن 17. samun		
	18. saxin سخن 'to be or become hot or warm'		
	to be or become small, little, tiny, minute; to decrease' صغر 'to be or become small, little, tiny, minute; to decrease'		
	to be or become ha صلب 20. salub	rd, solid, firm, stiff, rigid; to harden'	
	21. daxum ضخم 'to be or become huge hig large hulky sizable'		
	22 , $t\bar{a}l$ $d\bar{b}$ to be or become long to lengthen extend elongate tall'		
	23. tariy d_{1} to be or become sof	t tender fresh'	
	23. <i>and ي</i> في الله المعني 23. <i>and and a carud</i> عرض to broaden widen expand be or become broad?		
	25 $\dot{\alpha}_{azur}$ $\dot{\gamma}$ 'to be or become abundant conjous plentiful ample'		
	26. aasa ito be or become hard solid rigid firm stiff callous'		
	27. asur be or become short'		
	$28 \ kabur \leq 10$ to be or become great(er) hig(ger) large(r) to grow?		
	29 katuf \dot{s} to be or become thick(er) dense(r), (more) concentrated?		
	30 karis is 5 'to be or become wrinkled shriveled'		
	$31 \ lan \dot{i}\dot{\lambda}$ 'to be or become soft tender flexible nliable sumple'		
	32. <i>našif</i> نشف 'to be or become dry dehydrate desiccate exsiccate'		
	33. <i>nadiy ندى</i> to be or become wet dewy moist damp'		
	34 , $na^{c}um$ $is be or become soft smooth tender fine'$		
		.,,,,	
The syntactic	Verb + Subject		
frame of			
intransitives			
The nature of	The subject refers to the bodies and	l objects that are characterized by a	
the subject	change in their size, weight, appea	rance, touch, and temperature. The	
	subject may be an animate or inanimat	e object.	
Examples	 بدُن الجِسمُ 	18. سخُن الجِسمُ	
	2. برُد الجِسمُ	19. صبغُر الجِسمُ	
	3. بلِيَ الجِسمُ	20. صلب الجسمُ	
	4_ ثقّل الجِسمُ	21. ضخُم الجِسمُ	
	5_ جسُم الجِسمُ	22. طال الجِسمُ	
	6. جعد الجسم	23. طري الجِسَمُ	
	7. جلِد الجِسمُ	24. عرُّض الجِسمُ	
	8. جمُد الحِسمُ	25. غزُر الشعر	
	9. خشُن الجِسمُ	26. قسا الجِسمُ	
	10. خفَّ الجِسمُ	27. قصُر الجِسمُ	
	11. دقَّ الْجِسمُ	28. كَبُر الْجِسَمُ	
	12. ذاب الجِسمُ	29. كَتْف الشعر	
	13. ذابٍ الجِسمُ	30. كرِش الجِسمُ	
	14. رطب الجِسمُ	31. لان الجِسمُ	
	15. رقّ الجِسمُ	32. نشف الجِسمُ	
	16. سمك الجسم	33. ندِي الجِسمُ	
	17. سمُن الجِسمُ	34. نعُم الجِسمُ	

4.2.8 Verbs of motion (class 08)

Class Description	These verbs describe the direction and type of movement of the	
	subject, such as 'to come', 'to go', 'to leave' and 'to run'.	
Levin's class	Verbs of motion	
Vendler's class	Activity verbs	
-----------------	---	
Transitive or	Transitive	
intransitive	Intransitive	
Class members	1. <i>°atā أتى</i> 'to come, arrive'	
	2. <i>bariḥ</i> برح 'to leave, depart from, go away from'	
	3. $j\bar{a}^{\circ}$,	
	4. $j\bar{a}b$ + 'to travel (through, over), tour, cruise, patrol'	
	5. <i>jāl</i> جال 'to wander about, walk about or around'	
	6. <i>jarā</i> جرى 'to run, race, rush, course'	
	7. $hab\bar{a} \rightarrow$ 'to crawl, creep; to go on all fours'	
	8. <i>xaraj</i> خرج 'to go out, come out, emerge; to walk out'	
	9. xaṭar (في مشيته 'to strut, prance, swagger, mince'	
	10. <i>dār</i> دار 'to turn, revolve, rotate, twirl, spin, gyrate, roll'	
	11. daraj درج 'to toddle, walk with short steps'	
	12. <i>dahab ذهب</i> 'to go, go away, leave, depart'	
	13. daxal دخل 'to enter, come in(to), go in(to), get in(to)'	
	14. $r\bar{a}h$ راح 'to go, go away, leave, depart'	
	15. <i>raji ° ر</i> جع 'to return, come back, go back; to revert'	
	16. <i>raḥal</i> رحل 'to depart, decamp, leave, go away, part'	
	17. <i>zār</i> زار 'to visit, trip'	
	18. zaḥaf زحف 'to crawl, creep; to go on all fours; to march'	
	ito tour, travel, journey, voyage, cruise, rove' ساح 19. <i>sāḥ</i> ساح	
	20. <i>sār</i> سار 'to walk, tread, go on foot; to pace; to march'	
	'to drive; to pilot, steer' ساق 'to drive; to pilot	
	22. <i>sarā</i> سرى 'to travel by night'	
	23. <i>sarah</i> سرح 'to go out; to depart, go away'	
	24. sa $c\bar{a}$ (المي) 'to proceed to, head for, go to, take to seek'	
	to fall (down), drop, tumble; to sink (down)' سقط to fall (down)	
	26. <i>šarad</i> شرد 'to run away, flee, escape, break loose'	
	27. <i>sa ^cid</i> صعد 'to ascend, climb, mount (up), scale; to rise'	
	to fly; to fly away, take wing'	
	29. <i>tāf</i> die 'to circle, circuit, circumambulate'	
	30. $c\bar{a}d$ all 'to fly; to fly away, take wing'	
	31. <i>cabar</i> عبر 'to cross, traverse, go across, pass across'	
	32. $cad\bar{a} \simeq c$ to run, race, course; to jog	
	33. farr فر to escape, flee, run away	
	34. qād ^a to drive, steer, pilot	
	35. qadim فدم to come, arrive, show up; to reach, get to	
	to jump, leap, spring, bound, skip, hop	
	37. maša مسى to walk, tread, go on foot, move on foot	
	to go, go away, leave, depart مصنى 38. <i>mada</i>	
	39. nazah برح to emigrate, to immigrate (to); to migrate	
	40. <i>nazal</i> درك to descend, come down, go down, get down	
	41. $naii \rightarrow 10$ jump, leap	
	42. <i>nafar</i> 10 to flee, run, away, escape, break loose	
	45. narab هرب to nee, run away, take to flight, escape	
	44. walab et a sound, bound et a sound, bound et a sound to sound	
	$45. waşal e^{-2} to reach, arrive at, get to, come to$	
	40. wajaa $e^{-\omega}$ to come to, arrive at, get to, reach; to visit	
	47. waqa ° وقع to fall (down), drop, tumble, sink .	

The syntactic frame	Verb + Subject (+prepositional p	hrase)
of intransitives		
The nature of the	The subject is an animate being or	r an inanimate object that can be
subject	moved from one place to another.	
The nature of the	An optional prepositional phrase of	can follow the subject. It can be
Prepositional phrase	ilā followed إلى headed by min or	d by noun expressing place.
Examples	 أتئى محمد من السوق إلى المنزل 	25. سقّط محمد من المقعد إلى الأرض
1	 برح محمد من السوق إلى المنزل 	26. شرّد محمد من السوق إلى المنزل
	 جاء محمد من السوق إلى المنزل 	27. صعِد محمد من السّلم إلى الطائرة
	 جاب محمد من السوق إلى المنزل 	28. طار محمد من بريطانيا إلى فرنسا
	 جال محمد من السوق إلى المنزل 	29. طاف محمد من السوق إلى المنزل
	 جرى محمد من السوق إلى المنزل 	30. عاد محمد من السوق إلى المنزل
	 حبا محمد من السوق إلى المنزل 	31. عبّر محمد من السوق إلى المنزل
	 ٤. خرج محمد من السوق إلى المنزل 	32. عدا محمد من السوق إلى المنزل
	 .9 خطر محمد من السوق إلى المنزل 	33. فرّ محمد من السوق إلى المنزل
	10. دار محمد من السوق إلى المنزل	34. قاد محمد من السوق إلى المنزل
	11. درّج محمد من السوق إلى المنزل	35. قدّم محمد من السوق إلى المنزل
	12. ذهَب محمد من السوق إلى المنزل	36. قفَّز محمد من المقعد إلى الأرض
	13. دخَل محمد من السوق إلى المنزل	37. مشي محمد من السوق إلى المنزل
	14. راح محمد من السوق إلى المنزل	38. مضي محمد من السوق إلى
	15. رجَع محمد من السوق إلى المنزل	المنزل
	16. رحَل محمد من المدينة إلى القرية	39. نزَح محمد من سوريا إلى الأردن
	17. زار محمد من المدينة إلى القرية	40. نزَلٍ محمد من الطائرة إلى المطار
	18. زحَف محمد من السوق إلى	41. نطِّ محمد من المقعد إلى الأرض
	المنزل	42. نفر محمد من السوق إلى المنزل
	19. ساح محمد من السوق إلى المنزل	43. هرِّب محمد من السوق إلى المنزل
	20. سار محمد من السوق إلى المنزل	44. وثب محمد من المقعد إلى الارض
	21. ساق محمد من السوق إلى المنزل	45. وصَل محمد من السوق إلى
	22. سرَى محمد من السوق إلى المنزل	المنزل
	23. سرّح محمد من السوق إلى المنزل	46. وفد محمد من السوق إلى المنزل
	24. سعى محمد من السوق إلى المنزل	47. وقع محمد من المقعد إلى الارض

4.2.9 Verbs of swimming (class 09)

Class	These verbs refer to events in which the subject is surrounded by water. The	
Description	subject can be an animate or inanimate object.	
Levin's class	Verbs of motion	
Vendler's class	Activity verbs	
Transitive or intransitive	Intransitive	
Class members	 sabah سبح 'to swim, bathe' tafā لفا 'to float, buoy' <i>cām</i> عام 'to float, buoy; to swim' <i>gāş</i> ناص 'to dive into, plunge into, submerge in, sink into' <i>garaq</i> غرق 'to sink; to founder; to drown' <i>gatas</i> نطَس 'to dive, dip, plunge, sink, submerge'. 	
The syntactic	Verb + Subject (+Time Adverb)	

frame of	
intransitives	
The nature of	The subject is an animate or inanimate entity that can float, swim or dive.
the subject	
The nature of	Optionally, the time adverb indicates a certain period of time.
the Time	
Adverb	
	 سبَح محمد دقيقتين
	2. طفًا محمد دقيقتين
	3. عام محمد دقیقتین
	4. غاص محمد دقيقتين
	5. غرق محمد دقيقتين
	 غطس محمد دقیقتین

4.2.10 Verbs of location/place (class 10)

Class	These verbs denote that the locative subject is characterized by inherent	
Description	qualities that express distance, direction or topographic nature	
Levin's class	Verbs of entity-specific modes of being	
	Measure verbs	
Vendler's class	State verbs	
Transitive or	Intransitive	
intransitive		
Class members	 ba^cud بعُد 'to be or become far, faraway, far-off, distant, remote' janub جنب 'to be or become in the east' rahub ترغب 'to be or become wide, spacious, roomy' sahuq تعفی 'to be or become remote, distant, far, faraway' sama 'mad 'mad' 'to be or become remote, distant, far, faraway' sama 'mad' 'to be or become smooth, even, level, flat' šanud 'mad' 'to be or become in the east' šanud 'to be or become in the north' šahug 'to be or become high' dāq 'to be or become high, lofty, elevated, exalted, sub-lime' šahug 'to be or become high, lofty, elevated, exalted, sub-lime' calā 'a ab 'to be or become high, lofty, elevated, exalted, sub-lime' canuq ab 'to be or become high, lofty, elevated, exalted, sub-lime' garub 'to be or become high, lofty, elevated, exalted, sub-lime' garub 'to be or become in the west' qarub 'to be or become remote, distant, far, faraway, far-off' qasiy 'to be or become remote, distant, far, faraway, far-off' qac'ur 'to be or become remote, sloping, slanting, oblique' wa 'ur' 'to be or become inclined, sloping, slanting, oblique' wa 'ur' 'to be or become wide, roomy, spacious, vast, extensive'. 	
The syntactic frame of intransitives	Verb + Subject	
The nature of the subject	The subject refers to a place.	

Examples	 بعد المكانُ 	11. علا المكانُ
-	2. جنُّب المكانُ	12. عمُق المكانُ
	 رح رح بالمكان 	13. غرُب المكانُ
	4. سحُق المكانُ	14. قرُب المكانُ
	5. سما المكانُ	15. قصبي المكانُ
	6. سبعُل المكانُ	16. قُعُر ٱلْمَكَانُ
	7. شرُق المكانُ	17. مال المكانُ
	8. شمُل المكان	18. و عُر المكانُ
	9. شهق المكان	19. وسُع المكانُ
	10. ضباق المكانُ	-

4.2.11 Verbs of violence and abuse (class 11)

Class	These verbs denote physical harming that is done by a (usually) animate	
Description	subject. The object is usually an animate being.	
Levin's class	Hurt verbs	
	Verbs of killing	
Vendler's class	Accomplishment verbs	
Transitive or	Transitive	
intransitive		
Class members	 baṭaḥ خلخ 'to prostrate; to throw down, fell; to stretch (out), outstretch; to flatten, level (off)' baṭaš نطش 'to assault, attack with violence; to knock (out), strike sharply; to destroy, ruin, devastate, ravage, crush, stamp out' jalad خد' to whip, lash, flog, flagellate, slash, thrash, lace, scourge' jaraḥ حرح 'to wound, injure, hurt' haraŋ حرت 'to burn, incinerate, destroy by fire, consume with fire' xataŋ خذق 'to kidnap, abduct' xanaq خذق 'to strangle, strangulate, throttle, choke to death' dahas نفف 'to run over' rajam > 'to stone, pelt with stones, throw stones at' rafaŋ 'to kick' rakal 'to kick' ramaḥ 'to kick' ramaḥ 'to kick' ramaḥ 'to spear, lance' sanah 'to drag along, trail along on the ground' samm 'uto poison; to envenom, venom' 	
	 18. šanaq شنق 'to hang, halter, gibbet, execute by hanging' 19. sadam صدم 'to collide (with), clash (with), run (into), bump (against), bang (against), knock (against), hit (against)' 20. sara^c صرع 'to throw down, fell, knock down, strike down, bring to the ground; to knock out (in boxing)' 21. sa^caq صنف 'to strike, hit, shock' 22. safa^c o to slap, cuff, buffet' 23. darab · ضرب 'to beat, strike, hit; to knock, punch; to slap, flap' 24. tarah · طرح 'to stab, thrust, pierce, lunge, transfix, jab' 	

	'to bite' عض [°] 26. [°]	
	to kill, slay, murder, ; قتل 27. qatal	assassinate'
	to throw, cast, f فذف (ب) 28. <i>qadaf</i>	ling, hurl, toss, pitch; to pelt (with)'
	'to bite, sting' قرص 29. garas	
	to shell, bombar قصف 'to shell, bombar	d. bomb. batter. cannonade. cannon.
	attack with bombs or artillery f	ire fire guns at open artillery fire at'
	to commit or engage 'to commit or engage	e in highway robbery'
	$32 aama^{\circ}$ ito curb check rest	rain bridle repress suppress subdue:
	to crush quell quash'	ram, ondie, repress, suppress, subdue,
	33 gahar vš 'to subdue subjugate	conquer vanquish'
	34 kasar ws to brook froature al	batter smash erash erush'
	10 bleak, flacture, si 25. ladaććs) (to sting hito)	nauer, smash, crash, crush
	$35. \ ladag = 1$ to sting, bite	
	to sting, bite; to nurt	(with words), offend
	to slap, cuff, buffect لطم 3/. latam	; to strike, hit
	to punch, box, strike الكز 38. lakaz	e with the fist'
	to punch, box, strik لکم 39. <i>lakazm</i>	ke with the fist'
	to prick, sting; to pick نخز 40. <i>naxaz</i>	erce'
	to bite, snap (at)' نهش 41. <i>nahaš</i>	
	42. <i>waxaz وخز</i> to prick, sting, twir	nge; to jab, pierce'.
The syntactic	Verb + Subject + Object	
frame of		
transitives		
The nature of	The subject is usually an animate entit	y that commits a violent act. It may be
the subject	an inanimate entity that can cause viol	ence.
The nature of	The object is usually an animate entity	affected by violence or abuse.
the object		5
Examples	 1. بطح محمد الخصيم 	22. صفّع محمد الخصمَ
1	2. بطَش محمد الخصيم	23. ضرَّب محمد الخصْمَ
	3. جلد محمد الخصم	24. طرّح محمد الخصيم
	4. جرّح محمد الخصيمَ	25 طعن محمد الخصيم
	5 حرّق محمد الخصيم	26. عضّ محمد الخصيم
	6 خذق محمد الخصيم	22. قتّل محمد الخصيم
	7 خطَف محمد الخصبة	ع: 28 قَذَف محمد الخصيمَ
	، جنّق محمد الخصية	29 قرّص محمد الخصرة
		<u>22</u> قريف محمد الخصيم 30
	11. رسق محمد الخصم	
	14. رمح محمد الحصم	
	15. سحل محمد الخصيم	36. لسع الدحل الرجل
	16. سلب محمد الخصم	37 لطم محمد الخصيمَ
	17. سمِّ محمد الخصمَ	38. لكز محمد الخصم
	18. تنتق محمد الخصم	39. لكم محمد الخصيم
	19. صدّمت السيارة الرجلَ	40. نخز محمد الخصم
	20. صرّع محمد الخصمَ	41. نهَش محمد الخصمَ
	21. صعَق محمد الخصمَ	42. وخَز محمد الخصمَ

4.2.12 Verbs of mental process (class 12)

Class	These verbs refer to processes in the mind such as perception,	
Description	introspection, memory, creativity, imagination,	
_	conception, belief, reasoning, volition, calculation and emotion. The subject	
	is a human being.	
Levin's class	Verbs of perception	
Vendler's class	State verbs	
Transitive or	Transitive	
intransitive		
Class members		
	1. <i>bahat</i> بحث 'to study, explore, inquire into, examine, investigate,	
	inspect, check out, deal with, treat, research, do research'	
	2. <i>hazar</i> $\neq i$ o guess, conjecture, surmise, estimate, assess, appraise	
	3. <i>hasab</i> to calculate, compute, count, number, enumerate; to	
	consider, deem, regard as, look at as, judge, think	
	4. <i>nasto</i> to tillik, suppose, assume, take it, consider, deem, regard	
	5 hafiz bis 'to memorise learn by heart commit to memory'	
	5. $hall$ (b) to solve (a problem) resolve settle work (out) upravel?	
	7 $x\overline{al}$ (b) to think believe suppose assume imagine fancy take as	
	take it consider deem regard as'	
	8. <i>xabur</i> $\dot{\epsilon}$ 'to know, realise'	
	9. darā درى 'to know (of), have knowledge (of); to be or become	
	cognizant of, aware of, familiar with'	
	10. dall (على 'to prove, establish, verify, substantiate, demonstrate,	
	show, evidence; to attest to, testify, be evidence of, be proof of'	
	11. <i>dahab</i> (ذهب (في) 'to be of the opinion (that), hold the view (that), think	
	(that), believe (that)'	
	12. sabar سبَر 'understand or study thoroughly'	
	13. <i>zann</i> () ظن (to think, assume, guess, take for, take as, consider, deem,	
	regard as, hold; to suspect; to accuse (of), charge (with), indict (for)'	
	14. <i>c araf عرف</i> 'to know; to be or become cognizant of, familiar with; to	
	familiarize oneself with, acquaint oneself with; to learn, come to know,	
	Find out (about); to recognize, perceive $15 - 6 - 15 + 5 + 5 + 5 + 5 + 5 + 5 + 5 + 5 + 5 $	
	15. " $allm(-)$ = 10 know (01), nave knowledge (01); to be of become	
	16 farad in it suppose assume presume hypothesize'	
	17 fatan نفن 'to realize discern notice see understand grasp: to be or	
	become aware of to learn (about) come to know (about) find out	
	(about)'	
	18. <i>fahim فه</i> 'to understand, grasp, comprehend, apprehend, realize,	
	conceive, perceive; to know (of); to learn (about), hear (of)'	
	19. <i>nabih</i> (ل) نبه 'to notice, perceive, see, observe, take notice of, pay	
	attention to, heed, mind'	
	20. wa $c\bar{a}$ وعى 'to realise, understand, recognize'	
	21. wahim وهم 'to imagine, fancy, conceive, think, suppose, assume; to	
	take for, take as; to suspect; to misconceive, misunderstand,	
	misconstrue, misinterpret'	
	22. yaqin (بيقن (من 'to ascertain, find out or learn with certainty about, know	

	for certain; to be or become cert of'.	ain of, sure of, convinced of, confident
The syntactic frame of transitives	Verb + Subject + Object	
The syntactic frame of intransitives	Verb + Subject + Prepositional phrase	e + Complement
The nature of the subject	The subject is a human being who has and considering.	mental capabilities such as thinking
The nature of the object	The object may be an issue, question,	matter, problem or case.
The nature of the prepositional phrase	The prepositional phrase is headed by a noun that refers to an issue, question	the preposition <i>fī في</i> , and followed by n, matter, problem or case.
Examples	Transitives	 بحَث محمد المسألة محمد المسألة محمد المسألة حسّب محمد المسألة حفظ محمد المسألة حفظ محمد المسألة محمد المسألة محمد المسألة محمد المسألة درّى محمد المسألة
	Intransitives	15. خال محمد في المسألة 16. ذهّب محمد في المسألة 17. ظنّ محمد في المسألة 18. فرّض محمد في المسألة 19. فطِن محمد في المسألة 20. نبِه محمد في المسألة 21. وهِم محمد في المسألة 22. يقِن محمد في المسألة

4.2.13 Verbs of financial transaction (class 13)

Class	These verbs refer to financial transactions that take place between people or
Description	institutions in the stock exchange, trade, selling and buying. The subject can
	be the human agent of the transaction.
Levin's class	Cost verbs / Price verbs / Bill verbs
Vendler's class	Activity verbs
Transitive or	Transitive
intransitive	Intransitive

Class members	1. bār (بالعَمَلُ 'to be futile, unsuccessful, unprofitable, unfruitful,
	unavailing, useless; to fail'
	2. $b\bar{a}^{c}$ 'to sell; to retail; to wholesale'
	3. baxas بخس 'to decrease, diminish, reduce; to depreciate, lessen the value
	of, disparage, belittle; to undervalue, underestimate, underrate'
	4. basam بصبّر 'to (im)print, impress, stamp; to make or leave a fingerprint
	(imprint, impression) on'
	5. <i>balaغ</i> (المقدارُ أو المَجْمُوغ) to amount to, come to, make, reach, add up
	to, sum to, total, number'
	6. <i>jarad</i> (التَضَائعَ أو المَوْجُودات) خَرَدَ (التَضَائعَ أو المَوْجُودات) 6.
	inventory (of)'
	7. $ian\bar{a}$ $\dot{\epsilon}$ 'to earn, get, gain, win, profit, reap, harvest; to attain, achieve,
	accomplish: to incur, bring upon oneself, draw on oneself'
	$8 hasah(1)$ \dot{z} to credit with'
	9 $hasab(a)$ 'to debit with charge to someone's account'
	المعنان (الح) بالمعنان (الع) بالمعنان المعنان (الع) بالمعنان (الع) بالمعنان (الع) بالمعنان (الع) بالمعنان (الع
	to lose forfeit: to suffer or incur a loss'
	$12 xasam$ $\frac{1}{2}$ to discount (a bill note etc.) deduct rebate take off
	subtract'
	13. xafad خفض 'to reduce, lower, lessen, decrease, diminish, cut, cut back,
	cut down, scale down, minimise, abate, pull down, mark down; to slash'
	14. dān دان 'to borrow, contract (raise, take up) a loan'
	15. <i>dafa ^c</i> دفع 'to pay, settle, discharge, clear, liquidate, pay up; to disburse,
	spend, expend, lay out, outlay, pay out'
	16. rāj (راجَرت البِضَاعَة) 'to sell well, find a (good) market, be in (great)
	demand, sell like hot cakes, be sold rapidly and in large quantities'
	17. rabih ربح 'to gain, profit, win'
	to cheapen, be or become cheap or inexpensive' رخص 18. raxus
	19. <i>rašā</i> رشا 'to bribe, buy off, corrupt'
	20. rasad (مالأ) 'to appropriate, earmark, set apart, set aside, designate,
	dedicate, destine, allocate, allot, assign'
	21. rafa ^c رفع 'to raise, increase, step up, boost, hike, up, scale up, jack up,
	skyrocket; to heighten, intensify, enhance'
	22. rahan دهن 'to mortgage, pawn, pledge, hypothecate, give in pledge,
	deposit as security'
	23. zād زاد to increase, augment, step up, scale up, boost, hike up, jack up,
	up, skyrocket; to intensify, enhance, build up, add to, enlarge, magnify'
	'to offer for sale' سامَ 'to offer for sale'
	25. sahab سحب 'to withdraw, draw back, draw off, take back, recall, call
	back, revoke, repeal, cancel; to pull out; to take out'
	26. <i>šarā</i> شرى 'to buy, purchase'
	27. şaraf (مالأ) 'to spend, expend, outlay, lay out, pay out, disburse'
	28. <i>to</i> subtract, deduct, discount' طرح 'to subtract, deduct, discount'
	29. $^{c}al\bar{a}$ s 'to rise (high), tower up, go up, ascend, lift; to be or become
	high, lofty, elevated, exalted, sublime; to be or become loud'
	30. <i>ġašš</i> غشّ 'to cheat swindle'
	31. $gala$ 'to be or become expensive, high-priced'
	32. qabad فبض 'to receive, get, cash (money)'
	33. qasad نصد 'to economise, tighten one's belt; to be economical, thrifty,
	frugal provident: to husband manage economically: to save?

	34. garad قرَض 'to lend, loan, advance (money to)'		
	35. kasab کسب 'to gain, win, profit, earn, get, obtain, acquire, attain, reap,		
	harvest'		
	36. kasad (ت البضاعة 'to be unsalable, be unmarketable, be a dead		
	stock, find no market, remain unsold'.		
The syntactic	Verb + Subject + Object		
frame of			
transitives			
The syntactic	Verb + Subject		
frame of			
intransitives			
The nature of	The subject may be a human being inv	volved in financial transactions. The	
the subject	subject may also be a bank company	foundation or a form of commercial	
the subject	transaction	Toundation, of a form of commercial	
The nature of	The object is goods prices or comme	reial transactions	
the object	The object is goods, prices, or comme		
Examples	Transitiva	ا بابت التحادية	
Examples	Transitive	۲. بر <u>ب</u> (بیبرر) ۲. رادن الرضاعة	
		5	
	Intransitive	6 113 acas Ilicitas	
	Intransitive		
		0. <u>بع</u> ر السوم مئة ديزار	
		ري جي سر النظرية 10 جرز محمد النظرية	
		11 جنَّ محمد الأرياح	
		12 جسّب محمد الأرياح	
		13 حسّب محمد الأرياح	
		14 حفظ محمد البضاعة	
		۔ 15 خَسَر محمد الأرياح	
		16 خصَم محمد الأرياح	
		17 خفض محمد الأرياح	
		18. دان محمد البضاعة	
		19. دفَع محمد ثمن البضاعة	
		20 رشا محمد المستهلك	
		21. رصد محمد الأرباح	
		22. رفع محمد الأسعار	
		23. ربقن محمد البضاعة	
		24. زاد محمد الأسعار	
		25. سام محمد البضاعة	
		26. سحّب محمد الأسهم	
		27. شرّى محمد البضاعة	
		28. صرّف محمد الأرباح	
		29. طرّح محمد الأسهم	
		30. علا سعر السهم	
		31. غشَّ محمد البضاعة	
		32. غلا سعر البضاعة	
		33. قبَض محمد ثمن البضاعة	

	34. قصُد سعر البضاعة
	35. قرّض محمد المستهلك
	36. كسّب محمد التجارة

4.2.14 Verbs of agriculture (class 14)

Class	These verbs describe activities undertaken by a human subject or changes		
Description	that occur in plants and on the land.		
Levin's class	Grow verbs		
Vendler's class	Activity verbs		
Transitive or	Transitive		
intransitive	Intransitive		
Class members	1. <i>°abar</i> أبر 'to pollinate'		
	2. bār (بارَ (تَ الأرض) 'to lie fallow, remain uncultivated or unsown'		
	3. badr بذر 'to sow, seed, disseminate, spread, strew'		
	4. <i>bazr</i> بزر 'to sow, seed'		
	5. <i>ba ° ar</i> بغر 'to fertilize, manure, drop dung'		
	6. <i>baqal</i> بقل 'to grow; to sprout, germinate'		
	7. balaġ (الثمر) 'to ripen, mature, be or become ripe or mature'		
	8. <i>to bear fruit, fructify, fruit; to yield, produce, pay (off),</i>		
	return, bring forth, bring in; to succeed, work (out), turn out well'		
	9. <i>jadab</i> ب دب 'to be or become barren, sterile, infertile; to be or become		
	arid, dry'		
	to pick, gather, reap, harvest, pluck out, pull off ² جنى 10. <i>janā</i> جنى		
	to plow; to till, cultivate حرت 'to plow; to till, cultivate		
	12. <i>haşad</i> (الررع) to harvest, reap, gather, crop; to mow, cut down		
	13. <i>hafar</i> 2 to disinter, disentomb, exnume, unearth, excavate, dig up		
	14. <i>dabal</i> دبل to fertilize, manure, dung		
	15. aaras (الجنطة) to thresh, thrash (grain)		
	16. <u>aabat</u> to wither, will, wizen, shrivel, lade, dry up, liag		
	17. $\underline{aara} = 10$ winnow, ian (grain)		
	10 add is to spray sprinkle shower splash spatter (water)'		
	20 rače in isprinkle, to spray sparge shower: to (be)spatter splash		
	splatter shout squirt to water'		
	21 rawin (c α) 'to be irrigated watered supplied with water'		
	22 $zara^{\circ} \in \mathcal{V}$ 'to sow: to plant raise grow gron: to cultivate'		
	22. zara = 200 to sow, to plant, raise, grow, crop, to cultivate		
	drink quench someone's thirst sate'		
	24. <i>šatal</i> شتل 'to plant' to transplant'		
	25. cašub = 100 'to be or become grassy. grass-covered'		
	26. ġaras غرس 'to plant'		
	27. gahat قحط 'to be withheld (rain)'		
	28. <i>qaḥal</i> نحل 'to dry up; to be or become dry, arid, barren'		
	to trim, clip, pare, cut back, lop'		
	30. <i>qaşa °</i> قصع 'to grind, crush, bruise'		
	31. qataf نطف 'to pick, gather, reap, harvest, pluck out, pull off'		
	32. qalam قلم 'to clip, trim, cut, cut back, pare, prune, lop'		

	33. lagah نقح 'to pollinate; to fertilize, impregnate, fecundate'		
	34. mahal محل 'to be or become barren, sterile'		
	35. nabat نبت 'to grow; to sprout, germinate; to rise, spring'		
	36. natar نثر 'to scatter, disperse, disseminate, strew, bestrew, sprinkle'		
	37. namā نما to grow; to develop; to increase, augment, multiply, build up;		
	to thrive, prosper, flourish'		
	38. vabis the or become dry: to desiccate, exsiccate; to dry up, wither,		
	shrivel to stiffen harden solidify'		
The syntactic	Verb + Subject + Object		
frame of			
transitives			
The syntactic	Verb + Subject		
frame of			
intransitives			
The nature of	In terms of transitive verbs, the subject	t is a human being involved in	
the subject	agricultural activity while the subject	of intransitive verbs is a kind of plant	
the subject	or agricultural land	of initialisitive veros is a kind of plant	
The nature of	The object in the case of transitive ver	he is a kind of plant or agricultural	
the object	land	os is a kind of plant of agricultural	
Examples	Transitive	1 أبر محمد الزخل	
Examples		2 بذر محمد الأرض	
		۲. بعر مصب (2 رض 5. جزَّر محمد الثوار	
		م جرزت محمد الأرجن	
		0. عرب معمد (دریص 7. جویز محمد الثمار	
		جفَر محمد الأرض	
		9. <u>سور سال محمد الأرض</u>	
		ا درَس محمد المحصول	
		11 ذرّى محمد المحصبول	
		12 ردّ محمد الأرض	
		13. رُشّ محمد الثمار	
		14. روّى محمد الأشجار	
		15. زرع محمد الأشجار	
		16. سقّى محمد الأشجار	
		17. شتّل محمد الأشجار	
		18. غرّس محمد الأشجار	
		19. قصّ محمد الأشجار	
		20. قصّع محمد الأشجار	
		21. قطف محمد الثمار	
		22. قلّم محمد الأشجار	
		23. نثَّرُ محمد البذار	
	Intransitive	24. بارت الأرض	
		25. بقَل الشجر	
		26. بلُغ الشجر	
		27. ثمر الشجر	
		28. جدّبت الأرض	
		29. ذبَّلت الثمار	
		30. ذوَت الثمار	
		31. عشُبت الأرض	

	32. قحِطت الأرض
	33. قحَلت الأرض
	34. لقِحت الأشجار
	35. محَلت الأرض
	36. نبّتت الثمار
	37. نمَت الثمار
	38. يبِس الثمار

4.2.15 Verbs of desire and request (class 15)

Class	These verbs express that the subject has a desire or request to obtain		
Description	material or virtual things. The subject is usually a human being.		
Levin's class	Verbs of desire		
Vendler's class	State verbs		
Transitive or	Transitive		
intransitive	Intransitive		
Class members	 <i>amal</i> أما 'to hope; to hope for; to look forward to, expect, anticipate' tāq نائ 'to long for, yearn for, hanker after, pine for, hunger for, thirst for, die to; to crave, desire, desiderate; to miss; to aspire to' jaši e خان 'a' 'to be or become greedy, covetous, avid, avaricious, grasping' haris - 'to be greedy, covetous, avid, avaricious' haris - 'to long for, yearn for, hanker after, pine for, hunger for, thirst for, crave, desire, miss' da e a a a a a a a a a a a a a a a a a a		

	20. našad نشد 'to seek, look for, search for'		
	21. hadaf (هدف (إلى 'to aim at, drive at, purpose to, design to'.		
The syntactic frame of transitives	Verb + Subject + Object		
The syntactic frame of intransitives	Verb + Subject + Prepositional phrase		
The nature of the subject	The subject is a human being who has a desire or request to obtain.		
The nature of the object	The object is something that refers to desire and request.		
The nature of the prepositional phrase	The prepositional phrase starts with the preposition $f\bar{i}$ في 'in', <i>ilā</i> اللى 'to', and <i>alā</i> على 'on' followed by a noun that refers to a desire or request.		
Examples	Intransitives	 أمل محمد في النجاح جشع محمد في تجارته رغب محمد في النجاح رغب محمد في النجاح طمّع محمد في جارته طمّع محمد في تجارته طمّع محمد إلى النجاح حنّ محمد إلى النجاح دعا محمد إلى النجاح 	
	Transitives	13. راد محمد النجاح 14. رام محمد النجاح 15. رجا محمد النجاح 16. سأل محمد النجاح 17. شحّذ محمد المال 18. طلّب محمد المال 19. عاز محمد المال 20. قصّد محمد النجاح 21. نشّد محمد النجاح	

4.2.16 Verbs of intention (class 16)

Class	These verbs denote the intention or aim of the subject. The subject should be
Description	a human being.
Levin's class	Verbs of future having
Vendler's class	State verbs
Transitive or	Intransitive
intransitive	
Class members	1. <i>dahab</i> (ذهب (إلى 'to head for'

	2. $ram\bar{a}$ (رمى (إلى) 'to drive at, aim at, intend to, purpose to; to be aimed	
	at, be intended to'	
	3. $sa^{c}\bar{a}$ (إلى، ل، وراء) to seek (to), attempt (to), endeavor (to), try	
	hard (to), make every effort (to), strive (to); to pursue, strive for,	
	seek after ²	
	 <i>to aspire to or after, seek (to), aim to; to long for, yearn for</i> 	
	5. <i>camad</i> عمد 'to intend, purpose, design, mean, do on purpose, do	
	intentionally'	
	6. <i>canā</i> عنى to mean, intend, purpose, have in mind'	
	7. qaṣad قصد 'to intend, purpose, design, mean, have in mind, drive at,	
	aim at'	
	8. nawā نوى 'to intend, purpose, plan, design, mean, have in mind; to	
	resolve, determine, make up one's mind'	
	9. <i>hadaf</i> هدف 'to aim at, drive at, purpose to, design to'.	
The syntactic	Verb + Subject (+ Prepositional phrase)	
frame of		
intransitives		
The nature of	The subject is a human being who intends to do something.	
the subject		
The nature of	The prepositional phrase is headed by the preposition $i \bar{i} \bar{i}$ للى 'to', $f \bar{i}$ في in', or	
the	$bi \rightarrow to'$ tollowed by a noun that refers to the object of desire or request.	
prepositional		
phrase		
Examples	 دهب محمد إلى قول الصدق 	
	2. رمى محمد إلى النجاح	
	3. طمّح محمد في النجاح	
	4 سعى محمد إلى النجاح	
	 قصد محمد إلى النجاح اذم يجدد مل العردة. 	
	٥. لوى محمد على المصدى	
1		

4.2.17 Verbs of combining and constructing (class 17)

Class Description	These verbs denote works of combining, mixing, attaching and building that	
Description	being.	
Levin's class	Verbs of combining and attaching / Build verbs	
Vendler's class	Achievement verbs	
Transitive or intransitive	Transitive	
Class members	 banā بنى 'to build, construct, erect, set up, put up, raise, rear; to build up, develop, create, establish, make, form' jabā جبى 'to collect, levy, raise' jabal جبل 'to blunge; to knead' 	

	4. <i>jama</i> ^c جمع 'to gather, collect; to pick (up); to combine, group; to join,
	unite, connect, pair; to assemble, put together, fit together, joint'
	5. habak جبك 'to weave: to knit: to crochet'
	6. <i>hazam</i> جزم 'to pack, package, parcel, tie up, wrap (up), bundle, bale.
	stow'
	7. $haša$ 'to stuff fill fill up fill in to pad to wad to cram ram pack
	charge: to insert foist into: to interpolate'
	8 hašad عثيد to gather collect assemble congregate crowd rally round
	un to accumulate amass nile un hean un stack'
	9 $hašar$ \dot{w} 'to gather assemble crowd congregate overcrowd'
	$10 \ xat$ sew stitch to tailor?
	11 xafaa خفف 'to beat whin whisk'
	12 <i>valat</i> to mix mingle blend commingle admix combine'
	12. $\lambda_{amai} = 10$ mms, mingle, olend, commingle, damax, comone 13. $d_{amai} = 10^{-1}$ to merge amalgamate unite join fuse incorporate
	integrate affiliate combine blend'
	14 rabat by 'to bind tie (up) fasten make fast attach moor to connect
	link join units attach couple to tie (in): to ligate to correlate?
	15 ratagin 'to mond notch sow up dorn fin drow'
	15. radam () (to fill up (with earth))
	10. <i>radam</i> وتر الله الله (white earth)
	17. <i>Tuzum</i> (Can pack, package, parcel, wrap up, bundle, bale
	10. $radd $ to plug up step up; to stepper; to seel shut (off); to block (up
	19. saua io piug up, stop up, to stopper, to seal, shut (011), to block (up,
	20. sanai = w 'to broid plait'
	20. saraj السرج to blaid, plait
	21. sagaf (abc) to fool, cell
	<i>22. sada</i> to tighten, tauten, strain, draw tight, stretch tight, pull taut
	د to bundle, bale, wrap (up), pack, package, parcel; to bind, the
	(up)
	24. <i>aagai</i> to press, compress, squeeze
	to join, unite, bring together, conjoin, connect, bind; to
	combine, amalgamate; to group; to couple, pair; to gather, collect
	to bury, inter; to embed; to fill up (with earth)
	21. caqad see to knot, tie, knit; to fasten, lock; to join
	28. gazal عرب to spin
	to flood, overflow, inundate; to engult, gult, overwhelm; to
	cover, overspread, spread over, suffuse, fill, fill up, pervade
	30. <i>la³am</i> لأم to dress, bandage, bind up
	31. <i>laḥam</i> لحم 'to weld, solder; to fuse; to mend'
	32. <i>lasiq</i> 'to stick (to), adhere (to), cling (to); to agglutinate,
	conglutinate
	33. <i>laff</i> i 'to connect, join, attach'
	to collect, gather' لمّ to collect, gather'
	35. <i>mazaj</i> مزج 'to mix, mingle, blend, admix, commix, commingle,
	combine'
	36. <i>nasaj</i> نسج 'to weave; to knit'
	o connect, link, join, unite, joint, attach, وَصَلَ (الشَّيْءَ بِالشَيْء) 37. waṣal
	couple, pair'.
The syntactic	Verb + Subject + Object
frame of	
transitives	

The nature of	The subject is a human being wh	o can do work of combining and
the subject	constructing.	
The nature of	The object can be people (construction) materials, tools, or clothes.
the object		
Examples	 بنّى محمد الجدار 	20. سرَج محمد الثوب
_	 جبَى محمد الإير ادات 	21. سقَفٌ محمد البيت
	 جبّل محمد المواد 	22. شدّ محمد الحبال
	 جمَع محمد المواد 	23. صرّ محمد الجعبة
	 حبك محمد الثوب 	24. ضنغًط محمد المواد
	 حزم محمد الحطب 	25. ضبمّ محمد الحبال
	7 حشاً محمد العشب	26. طمَر محمد الحفرة
	8. حشد محمد الطلاب	27. عقّد محمد الحبال
	 - حشر محمد الأغنام 	28. غزَل محمد الثوب
	10. خاط محمد الثوب	29. غمَر محمد الحفرة
	11. خفَق محمد المواد	30. لأم محمد الحفرة
	12. خلّط محمد المواد	31. لحَم محمد الشق
	13. دمَج محمد المواد	32. لصبق محمد المواد
	14. ربّط محمد الحبال	33. لفّ محمد الحبال
	15. رتَق محمد الثوب	34. لمّ محمد المواد
	16. ردّم محمد الحفرة	35. مزَّج محمد المواد
	17 رزّم محمد الحطب	36. نسَج محمد الثوب
	18. رقَع محمد الثوب	37. وصَل محمد الحبال
	19. سدّ محمد الحفرة	

4.2.18 Verbs of sending and carrying (class 18)

Class	These verbs describe actions that involve carrying animate and inanimate		
Description	objects.		
Levin's class	Verbs of sending and carrying		
Vendler's class	Achievement verbs		
Transitive or	Transitive		
intransitive			
Class members	1. ba cat بعث 'to send, send out, dispatch, forward, expedite'		
	2. <i>jarr</i> جَّ 'to draw, pull, drag, tug, haul, tow; to trail along (on the		
	ground), drag along'		
	3. <i>jaraf</i> جرف 'to sweep (away); to drift; to wash away, wash out; to carry		
	off, tear away; to erode; to shovel (away), spade; to plow; to remove'		
	4. <i>hamal</i> حمل 'to carry; to load up, lift; to convey, transport, deliver; to		
	transmit (to), communicate (to), take (to); to carry with oneself, to carry		
away, take away'			
	5. $dafa^{c}$ Let to propel, move, impel, drive forward, shove, thrust, push		
ahead, push along, push forward, rush'			
	6. $rafa^{c}$ c		
	upheave, jack up, hike, up'		
	7. zaff (زف (العروس) to carry home the bride in procession; to give away (in		
	a marriage ceremony)'		
	8. $s\bar{a}q$ ساق 'to carry (along), transport, transfer, transmit, deliver; to		
	convey, communicate, impart, bring, transmit; to send, dispatch'		

	9. saḥab سحب 'to pull, draw (off), dr	aft, drag, haul, tug, tow; to trail along	
	(on the ground), drag along'		
	10. sakab سکب 'to pour (out), shed, spill, empty'		
	11. šāl شال 'to raise, lift (up), pick up; to carry, convey, transport'		
	12. šaḥan شحن 'to ship, freight, consign, transport, forward'		
	13. <i>°abar</i> (ب.) عبر (to carry (take, transport) across or through or over'		
	14. <i>catal</i> عتل 'to carry'		
	to transport, carry, haul ' نقل 15. <i>naqal</i>	; to take; to move; to transfer, shift; to	
	transmit, deliver; to convey, comr	nunicate, impart, bring'	
	16. wazar وزر to carry, bear (a burde	n)'.	
The syntactic	Verb + Subject + Object		
frame of			
transitives			
The nature of	The subject can be an animate being,	vehicle, or natural factor that can	
the subject	move things from one place to anothe	er.	
The nature of	The object is an entity that can be mo	ved from one place to another.	
the object			
Examples	 بعَث محمد الأشياء 	9 سحب محمد الأشياء	
	 جرَّ محمد الأشياء 	10. سكب محمد الأشياء	
	 جرف محمد الأشياء 	11. شال محمد الأشيباء	
	 4. حمل محمد الأشياء 	12. شحَن محمد الأشياء	
	 دفع محمد الأشياء 	13. عبّر محمد الأشباء	
	 رفع محمد الأشياء 	14. عتَّل محمد الأشياء	
	 زف محمد الأشياء 	15. نقل محمد الأشياء	
	8. ساق محمد الأشياء	16. وزَر محمد الأشياء	

4.2.19 Verbs of separating and disassembling (class 19)

Class	These verbs denote an action in which a usually animate subject separates or	
Description	breaks up animate and inanimate objects.	
Levin's class	Verbs of separating and disassembling	
Vendler's class	Accomplishment verbs	
Transitive or intransitive	Transitive	
Class members	 batar بنر 'to cut off, sever, lop off' jadam خَمَ 'to cut off, chop off, lop off; to amputate, remove; to mutilate, maim' hazz خَعَ 'to cut, cut off' hall خَتْ 'to untie, unfasten, unbind, undo, unravel, loosen, unloose, unfix, unwind, unscrew, untangle, disentangle, disengage, free' daqq نق' 'to pound, grind, crush, bruise, bray, powder, beat, stamp, pulverize, comminute, pestle, triturate' dakk غَنْ 'to tear down, pull down, raze, wreck, demolish, destroy, ruin, crush, smash, devastate; to undermine' sahaq سحق' 'to crush, pound, grind, beat, bruise, stamp, bray, powder, pulverize, comminute, pestle, triturate' sahaq نظر 'to halve, bisect, divide into two (usually equal) parts; to split, intersect, cut across; to sunder, sever, cut off' 	

	fo split, cleave, fissure, شقّ 9. šaqq	crack, rift, break; to tear, rend, rip		
	(apart), rive; to cut open, slash or	pen, slit open, cut, incise, make		
	incisions into; to carve up, dissect'			
	$10 \ sada^{c} \ except{salar}$ to split cleave crack	s break rift'		
	$11 c_{azal}$ is separate isolate se	clude segregate set aside or anart		
	remove'	ciude, segregate, set uside of upurt,		
	12 fataha فتح 'to open unlock uncle	ose unfasten unfold unwran'		
	12. faraz j i 'to separate set apart of	r aside isolate'		
	13. juruz = 10 separate, set apart of $14. fangm = i $ (to minor abon (un) h	ash'		
	to mince, chop (up), in 15. fragm i i i i to datach digioin diga	asii		
	15. Jusux ito detacti, disjoin, disc	onneci, separate		
	to separate, part, divid	letach break un receive?		
	17 fable 'to ungenery'	letach, break up, resolve		
	17. <i>fakk</i> i to unscrew	ift much sime to be not been been a		
	18. <i>falaq</i> فلق 'to split, cleave, fissure, rift, rend, rip; to burst, break open, break apart'			
	19. <i>qasam نقس</i> م 'to divide, part, split, s subdivide'	eparate, break up, partition, section,		
	to cut, cut off, clip, snij فصّ 20. <i>qaṣṣ</i> نصّ to cut, cut off, clip, snij (wool): to mow, cut down (grass.	p; to scissor; to shear, shear off, fleece		
	21. aata ° نظم 'to cut, cut off, sever:	to chop off: to cut down (a tree): to		
	amputate: to break: to divide	section: to segment: to separate.		
	disconnect, tear (apart)'	disconnect tear (apart)?		
	22. <i>kasar</i> کسر 'to break, fracture, shatter, smash, crash, crush' 23. <i>maza</i> ^o مزع 'to split, break open, break apart' 24. <i>natar</i> نثر 'to scatter, disperse, disseminate, strew, bestrew, sprinkle' 25. <i>našar</i> نشر 'to saw' 26. <i>najar</i> نجر 'to hew (out), carve, plane; to whittle, pare' 27. <i>nagar</i> نقر 'to dig; to excavate, hollow out; to bore, drill, pierce, hole,			
	nit'	, p,, p,		
	to tear down pull doy فده 18	wn raze wreck demolish destrov		
	devastate: to subvert'			
The syntactic	Verb + Subject + Object			
frame of	verb + Subject + Object			
transitives				
The nature of	The subject can be an animate being	vahicle or natural factor that can		
the subject	separate or break up animate and inan	imate objects		
The nature of	The object is an animate or inanimate	antity that is able to be separated or		
the object	disassembled	entity that is able to be separated of		
Examples		15 فينت بعد القبان		
Examples		15 فمركل محمد العماس		
	2. جدم محمد الحبل	10. فلق برجيد الدان		
		۲۱. ۲۵ محمد الباب ۱۹ خاند در ۱۱ ار ت		
	4 حل محمد الحبن حديثة مد الحب	18. قلق محمد الحجاره		
	ζ_{1} co aca hamal ζ_{2}			
		20. قص محمد القماس		
	10 11 11 11 11			
	12. فتح محمد الصندوق	20. نفر محمد انجدار		

13. فرَز محمد المواد	27. نشّر محمد الخشب
14. فرَم محمد اللحم	28. هدّم محمد الجدار

4.2.20 Verbs of removing (class 20)

Class	These verbs refer to a physical action by which a usually animate subject		
Description	removes a usually inanimate object from its location.		
Levin's class	Verbs of removing		
Vendler's class	Accomplishment verbs		
Transitive or	Transitive		
intransitive			
Class members	 jahaf حف 'to scrape off, scratch off, shave off; to peel off; to sweep away, carry away' jaraf خرف'' to sweep (away); to drift, carry away; to wash out; to carry off, tear away; to erode; to shovel (away), spade; to plow; to remove' jazz '> 'to cut, cut off, clip; to shear, shear off, fleece (wool); to mow, cut down (grass, etc.)' hadaf خف'' to delete, cancel, strike off, expunge, cross off; to eliminate, cut out, take out; leave out, drop, skip; to take away, clip off' halaq i خف' to shave, shave off; to have one's hair cut, have a haircut' xala ' خاف 'to extract, pull out, pluck out, tear out, uproot, root up, root out, deracinate' šatab '' to strike off, strike out; cross out, scratch out, cancel, delete; to remove, take off, take out; to erase, efface, expunge' gahaf i 'to sweep away, carry away; to scoop; to scrape (off), graze' qašat b'' to scrape off, scratch off, rub off, shave off, abrade, graze, excoriate, gall; to remove' mahā '' to erase, wipe off, rub out, sponge out; to efface, obliterate, blot out, expunge; to wipe out, erase; to blot out, efface, obliterate; to wipe out, eradicate, exterminate, extinguish; to eliminate, abolish, put an end to' marat '> 'to pluck out, pull out, tear out' masah cu' to wipe off, rub out, erase; to blot out, efface, obliterate; to wipe out, eradicate, exterminate, exter, blot out, eradicate, exterminate, extinguish; to eliminate, abolish, put an end to' marat '> 'to pluck out, pull out, tear out (hair, feathers, etc.); to deplume (feathers); to depliate, (hair)' nataf '' to pluck out, pull out, tear out (hair, feathers, etc.); to deplume (feathers); to depliate, (hair)' nataf '' to shake (off), dust off''. 		
The syntactic frame of transitives	Verb + Subject + Object		
The nature of	The subject can be an animate entity, vehicle, or natural factor.		
the subject			

The nature of	The object is an inanimate entity that	can be removed or snatched from
the object	their places.	
Examples	 جحَف محمد التراب 	10. قلّع محمد الشجرة
	2. جرَف محمد التراب	11. كشَّط محمد التراب
	 جزّ محمد العشب 	12. محًا محمد الرسم
	4. حذّف محمد الملفات	13. مرَّط محمد الحبلُ
	5. حلَق محمد شعر ہ	14. مسّح محمد الماء
	 6. خلَع محمد الثياب 	15. نتَف محمد الشعر
	7. شطَّب محمد الملفات	16. نزّع محمد الشعر
	 8. قحَف محمد التراب 	17. نفَض محمد التراب
	9. قشَط محمد الماء	

4.2.21 Verbs of bending (class 21)

Class	These verbs denote an action by which the subject bends a usually inanimate		
Description	object.		
Levin's class	Bend verbs		
Vendler's class	Accomplishment verbs		
Transitive or	Transitive		
intransitive	Intransitive		
Class members	 baram برم 'to twist, twine, entwine, curl, kink' <u>t</u>anā ننى 'to fold, double; to tuck, roll up; to bend, flex, turn, twist, curve, inflect' <u>haraf</u> حرف 'to slant, cant, incline, tip, tilt, swerve, deviate, deflect' <u>t</u>awā عرف 'to fold, double; to roll up, tuck, pleat; to bend, flex, turn' <u>fatal</u> to twist, twine, entwine, curl, kind' <u>qalab</u> 'to turn; to turn over, overturn, tip over, capsize; to turn up(ward), upturn; to turn upside down; to turn inside out or outside in' <u>laff</u> 'to wrap up, roll up, fold up, furl; to wind, coil, spool, reel, convolute, curl; to twist, twine' <u>lawā</u> لوى 'to twist, curl, kink, contort, wrench, wring, writhe, wry; to bend, incline, turn, curve, flex, inflect'. 		
The syntactic frame of transitives	Verb + Subject + Object		
The nature of	The subject is a human being who has the ability to bend things such as		
the subject	clothes and bonds.		
The nature of	The object is an inanimate entity that is able to be bent.		
the object			
Examples	 برأم محمد الحبل 		
	2 تئی محمد الحبل		
	4. طوی محمد اللباب ح بتا ماند ا		
	8. لوى محمد الفضيب		

4.2.22 Verbs of decorating and transcribing (class 22)

Class	These verbs denote an action which	h involves painting, decorating, and	
Description	transcribing works. The subject who does the work should be a human		
	being.		
Levin's class	Image creation verbs		
	Scribble verbs		
Vendler's class	Achievement verbs		
Transitive or	Transitive		
intransitive			
Class members	 jaşş جص 'plaster, plaster of paris; gypsum; parget, stucco' xatam خت 'to seal, signet, stamp, impress, imprint, rubber-stamp; to postmark' xati 'خَتْ 'to write, pen, inscribe; to draw, trace, line, sketch, design; to inscribe, carve, engrave' damaġ نَحْن 'to stamp, imprint, impress, print' dahaŋ 'co draw, trace, sketch, delineate, design; to describe (a circle, etc.); to paint' sadaġ ci 'to polish, burnish, scour, paint, tincture, imbue; to pigment' şaqal 'o 'to polish, burnish, scour, smooth, sleek, slick, brush up, glaze, gloss, shine, (re)furbish, luster, buff, lap, finish; to calender; to refine, cultivate' taba ' to paint, daub' katab 'ito paint, daub' katab 'ito paint, to type, typewrite; to stamp, impress, imprint' to paint, daub' katab 'ito paint, daub' katab 'ito paint, daub' katab 'ito paint, to compose, compile, draw up, draft, indite' kašat 'ito scrape off, scratch off, rub off, shave off, abrade, graze, excoriate, gall; to remove' naḥat' ico hew (out), cut (out), carve, sculpture, chisel, grave' naqaš 'ito engrave, incise, inscribe, chase; to carve (out), sculpture, chisel' 		
	16. wašam وشم 'to tattoo'.		
The syntactic	Verb + Subject + Object		
frame of			
transitives			
The nature of	The subject is a human being who has the ability to paint, decorate or		
the subject	transcribe.		
The nature of	The object is an inanimate entity that can be decorated or inscribed.		
Examples			
Examples		9. طبع محمد الورقة 10. مالاً معدد المحدل	
	ے. حتم محمد الور ک	10. طلی محمد الجدار 11 - کترب محمد الرسیالة	
	$\frac{1}{2}$ $\frac{1}{2}$	11. حتب محمد الرسان- 12. كشط محمد الحدار	
1	+ -ر -	J	

 دهَن محمد الجدار 	13. نحَت محمد التمثال
 رسم محمد اللوحة 	14. نقَش محمد التمثال
7. صَبَغ محمد التُوب	15. وسَمَّم محمد اللوحة
8. صقّل محمد الزّجاج	16. وَشَمَ محمد يده

4.2.23 Verbs of measurement (class 23)

Class	These verbs denote an action by which the subject measures the weight or		
Description	size of an object.		
Levin's class	Measure verbs		
Vendler's class	Achievement verbs		
Transitive or intransitive	Transitive		
Class members	 rajaḥ حج 'to weigh something in the hand' razan رزن 'to weigh in the hands' sā ° صاع 'to measure; to weight' qās صاع 'to measure, gauge, quantify' kāl لكل 'to measure; to gauge; to weigh' wazan وزن 'to weigh, determine the weight of'. 		
The syntactic frame of transitives	Verb + Subject + Object		
The nature of the subject	The subject is a human being who has the ability to weigh or measure things.		
The nature of the object	The object is an animate or inanimate entity that can be weighed or measured.		
Examples	رجَح محمد الوزن رزَن محمد الوزن صاع محمد الكمية قاس محمد المسافة كال محمد الكمية وزَن محمد الكمية		

4.2.24 Verbs of quantity and size (class 24)

Class	These verbs refer to the quantity or size of an animate or inanimate subject.	
Description		
Levin's class	Measure verbs	
Vendler's class	State verbs	
Transitive or intransitive	Intransitive	
Class members	 s 1. daqq ندق 'to be or become small, little, tiny, minute' 2. zād زاد 'to increase, grow, augment, multiply, swell, build up, intensify, heighten; to be or become more, great(er), large(r)' 3. šaḥḥ شخ 'to run short, fall short, run out, fail, be scarce or insufficient, decrease, diminish, decline, dwindle' 	

	4. <i>sagur</i> صغر to be or become si	mall, little, tiny, minute; to decrease,
	diminish, lessen, shrink, wane, dy	windle
	5. <i>gazur</i> غزر 'to abound, supera	bound; to be or become abundant,
	copious, plentiful, ample	
	6. <i>fāḍ</i> فاض to abound, superaboun	d; to be or become (super) abundant,
	pientitui, copious	
	to be or become little, sm في to be or become little	nall, few; to lessen, decrease, diminish,
	drop (off), grow less; to be or be (than)'	come less (than), smaller (than), fewer
	to shrink, dwindle, re فلص 8. galus	cede, diminish, decrease, decline'
	9. <i>kabur</i> کبر 'to be(come) or grow	great(er), big(ger), large(r); to grow,
	increase, enlarge, augment, swell	
	to increase, grow, m کثر 10. katur	ultiply, augment; to abound, exist in
	large numbers or amounts: to be or become much, many numerous	
	abundant plentiful'	
	to be rare infrequent نَدُر	uncommon scarce'
	to decrease diminish نقص 12 nagus	lessen become less grow less drop
	(off) decline fall abate to be re	duced decreased diminished'
The syntactic	Verb + Subject	
frame of		
intransitives		
The nature of	The subject can be an animate or ina	nimate entity that is characterized by
the subject	possible increase or decrease in its qua	ntity or size.
Examples	 دق الطريق 	7_ قلّ الماء
	2. زاد الماء	8. قلّص الماء
	3۔ شحّ الماء	9 كبُر الطريق
	4. صَغُر الطريق	10. كثُر الماء
	5. غزُر الماء	11. ندُر الماء
	6. فاض الماء	12. نقُص الماء

4.2.25 Verbs of stability (class 25)

Class	These verbs denote that the subject is stable and unchangeable in a certain
Description	place.
Levin's class	Lodge verbs
Vendler's class	State verbs
Transitive or	Intransitive
intransitive	
Class members	 ^awā (الحى) أوى (إلى) 'to lodge at, take up lodgings at, put up at, stay at, take up quarters at; to take refuge in, seek or take shelter in, harbor in' bāt باتَ 'to spend or pass the night; to stay overnight' baqay بقى 'to remain, stay; to be left over, be left behind; to last, continue, persist, subsist, go on; to keep on (doing), keep doing, persist in, persevere in, stick to' <u>tabat</u> 'to be or become firm, fixed, stable, steady, solid, strong, enduring, durable, lasting, permanent, unshakable, unchangeable, established; to stand firm; to set, become fast or hard; to be or become

	constant, invariable, unchangeable, steady'	
	5. to stay at, remain at, dwell in, reside in, live in; to settle	
	(down) at'	
	6. <i>jatam</i> جدم 'to crouch, cower; to lie face down, fall prone, lie prone,	
	prostrate oneself $7 h (t)$ is helt store make a store dimension to the balt store make a store dimension to the store store store dimension to the store sto	
	<i>i</i> , <i>hati</i> to halt, stop, make a stop, dismount, alight; to encamp; to	
	$\begin{array}{c} \text{camp} \\ \text{surf}_{n} d \left(x + y \right) \\ \text{surf}_{n} d$	
	8. xalaa (\downarrow, \downarrow) to ablde in, remain al, stay at	
	9. <i>adm</i> $\sum_{i=1}^{n}$ to last, continue, go on, persist, subsist, endure, remain	
	10. $rasa$ (10 be of become fifth, fixed, stable, steady)	
	seated, firmly established; to be or become firm, fixed, settled, stable,	
	steady, solid, strong; to set, become fast'	
	12. rakad رك 'to stagnate; to be or become stagnant, dull, slack'	
	13. <i>rakan ر</i> کن 'to be or become firm, steady, solid'	
	14. <i>sakan</i> سکن 'to live in, dwell in, reside in, lodge in, house in, domicile in;	
	to inhabit, populate; to settle (down) in, stay in, remain in	
	to withstand, resist, oppose; to hold (out), hold one's own,	
	endure, stand, last, remain firm, be steadfast	
	to remain, stay' طل 'to remain, stay'	
	to be or become well-established, firmly established, fixed,	
	steady, stable, solid, enduring, durable, lasting, unsnakable; to be or	
	become stationary, immobile 19. a_{atan} ($(,,,,,)$) \therefore is the live in dwell in reside in lodge in the inhebit	
	ito five in, dwell in, feside in, lodge in, to innabit, populate'	
	19. $laja^{\circ}$ 'to resort to, turn to(for security or help), take or seek refuge or	
	shelter in or with, seek protection in or with; to refer to'	
	20. <i>labi<u>t</u></i> (بــ) 'to stay in, remain in, abide in, keep to'	
	21. <i>maka<u>t</u></i> (-:) مکث (to stay in, remain in, reside in, dwell in live in'	
	22. <i>nazal</i> (نِزَلَ (بِه في، عِنْدَ، على) to stop at, make a stop at, halt at; to stay at,	
	lodge at, put up at, stop over at, take lodgings at; to settle down in, reside in, live in, inhabit'	
	23. <i>watad</i> وطد 'to establish, settle, stabilize, make firm, make stable: to	
	strengthen, consolidate, brace, cement, firm up, reinforce'	
	24. <i>watan</i> (بطن (ب) 'to settle down in, reside in, dwell in, live in, inhabit'.	
The syntactic	Verb + Subject (+ Prepositional phrase)	
frame of		
intransitives		
The nature of	The subject is an animate entity who chooses to stay in a certain place.	
the subject	5 5 1	
The nature of	The prepositional phrase is headed by the preposition fi in followed by a	
the	noun that refers to a place.	
prepositional	1	
phrase		
Examples	14. سكّن محمد في المنزل 1. أوّى محمد في المنزل	
1	15. صمّد محمد في المنزل	
	16. ظلّ محمد في المنزل	
	17. قرّ محمد في المنزل . 17. ثبّت محمد في المنزل	
	18. قطَن محمد في المنزل 5.	
	19. لجَأ محمد في المنز ل	

 حط محمد في المنزل 	20. لبِث محمد في المنزل
 ٤. خلد محمد في المنزل 	21. مُكَثْ محمد فَى المنزل
 دام محمد في المنزل 	22. نزَّل محمد في المنزل
10. رسًا محمد في المنزل	23. وطَد محمد في المنزل
11. رسّخ محمد في المنزل	24. وطَن محمد في المنزل
12. ركّد محمد في المنزل	
13. ركّن محمد في المنزل	

4.2.26 Verbs of creation (class 26)

Class	These verbs denote an action by which the subject creates or makes animate
Description	or inanimate objects.
Levin's class	Create verbs
Vendler's class	Achievement verbs
Transitive or	Transitive
intransitive	
Class members	 bara² برأ 'to create (said of God)' banā نبن 'to build, construct, erect, set up, put up, raise, rear; to build up, develop, create, establish, make, form' jabal جب' to mold, fashion, form, shape, frame, work, create, make' ja 'al 'eal خلق 'to create, make, form, fashion' xalaq خاخ 'to create, make, originate; to mold, fashion, shape, form, work; to produce, bring into being, engender, generate, bring about' dara² خل 'to create' şāg żu' 'to form, shape, fashion, mold, frame, forge, work; to create, originate, make' sana ' خل 'to make, do, perform; to manufacture, fabricate, produce; to fashion, form, create; to work out, bring about' camil عبل 'to do, make; to act; to perform, carry out, execute, fulfill, accomplish; to produce, manufacture, fabricate' fatar نظر 'to create, make, originate, bring into being' wada^a وضع 'to create, make, originate, bring into being' wada^a وضع 'to create, make, originate, bring into being'
The syntactic	Verb + Subject + Object
trame of transitives	
The nature of	The subject is a creator or a maker who has the ability to create or make
the subject	animate or inanimate objects.
The nature of	The object can be an animate or inanimate entity that is made or created by
the object	the subject.
Examples	 7. صاغ محمد الذهب 1. برأ الله المخلوقات
	 عمد اللعبة منع محمد اللعبة د. بنی محمد المنزل
	 عمل محمد الأدوات جبل الله الكون
	10. فطر الله الخلق
	11. وضَع محمد الأشياء 5. خلق الله الكون
	6. ذرأ الله الخَلق

4.2.27 Verbs of preparing (class 27)

Class	These verbs denote daily work that takes place in houses or restaurants, such	
Description	as cooking, washing, cleaning and tidying.	
Levin's class	Verbs of preparing	
	Cooking verbs	
Vendler's class	Activity verbs	
Transitive or	Transitive	
intransitive		
Class members	1. <i>hakk حاف</i> 'to rub, scrub; to scratch, scrape; to scrape off, rub off, scratch	
	off, abrade, chafe, fret'	
	2. <i>xabaz</i> خبز 'to bake; to make bread'	
	3. xadam خدم to serve; to attend, wait on; to work (for); to render a service	
	to, do someone a favour	
	4. xazan \prec_{i} to store, stock, warehouse, reposit; to store up, lay up, lay	
	by; to noard, amass, accumulate, stockpile	
	5. $xaaa = 10 \text{ mix}$, mingle, blend, commingle, admix, combine $6 = da^{2}ak^{2}c^{3}c^{3}$ to rub: to massage: to scrub scour?	
	7 sahaa i = 10 i = 10	
	nulverize comminute nestle triturate'	
	8. <i>salaa</i> سلق 'to boil cook in boiling water'	
	9. <i>šataf</i> شطف 'to rinse, wash'	
	10. šawā شوى 'to grill, broil, roast, barbecue'	
	to roast, broil, grill' صلَى to roast, broil, grill'	
	12. dagat ضغط 'to press, compress, squeeze; to press, compress, push, force	
	(together), serry, pack (tight), jam (together); to squeeze'	
	13. <i>tabax</i> طبخ 'to cook'	
	to grind, mill, pulverise, crush' طحن to grind, mill, pulverise, crush'	
	15. tahā طها 'to cook'	
	16. $taw\bar{a}$ طوى to fold, double; to roll up, tuck, pleat; to bend, flex, turn'	
	17. cajan 'as to knead'	
	18. <i>casar</i> عصر to press (out), squeeze (out), express, compress wring	
	19. garaf $\exists c$ to ladle, scoop (up), dip out	
	20. gasai to wash, fillse, lave, flush, clean(se) 21. $\dot{\sigma}al\bar{a}$ \dot{b} (to boil bubble (up); to simmer?	
	$21. gala = 10 0011, 000010 (up), to similar 22 farak \frac{3}{2} is 'to rub, scrub: to chafe'$	
	$22. jaran = 2^{\circ}$ to rub, setub, to enale	
	$24 aara^{\circ} \varepsilon$ ito knock ran bang beat (on or at a door): to ring sound	
	toll (a bell)'	
	25. <i>aašš</i> نُشْنُ 'to sweep, broom, brush away'	
	د. 26. <i>galā قلى</i> 'to fry'	
	27. kabas (كبس (على 'to press, compress, squeeze'	
	28. kanas کنس 'to sweep, broom, scavenge; to vacuum, vacuum-clean'	
	29. kawā کوی 'to iron, press; to launder'	
	30. masah مسح 'to wipe; to mop (up); to rub; to clean, polish; to sweep; to	
	wash'	
	31. <i>mala</i> [°] مد [⊂] to fill, fill up; to fill in [°]	
	32. nafad نفض 'to shake (off), dust off'	
	33. naqa ^c نقع 'to soak; to steep, infuse; to saturate, drench; to macerate'	
	34. haras هرس 'to mash, squash, crush, bruise, pound, pestle'.	

Verb + Subject + Object	
The subject is a human being who w kitchen.	vorks in the house, particularly in the
The object is an inanimate entity such	as clothes, food or kitchenware.
 حكّ محمد الوعاء خبَز محمد الخبز خدَم محمد الضيوف خذَم محمد القمح خزَن محمد اللقمح خطَط محمد الثياب دعَك محمد القمح دعَك محمد القمح سحَق محمد اللعمم سحَق محمد اللحم شعَط محمد اللحم ضعَط محمد اللحم طحَن محمد اللحم 	 18. عصر محمد الليمون 19. غرّف محمد الطعام 20. غسّل محمد الثياب 21. غلّى محمد الماء 22. فرّك محمد الوعاء 23. فرّع محمد جرس الطعام 24. قشَّ محمد المنزل 25. قشَّ محمد اللحم 26. قشَ محمد الليتون 27. كبّس محمد الزيتون 28. كنّس محمد الثياب 29. كوّى محمد المنزل 30. مسّح محمد الوعاء 31. ملأ محمد الثياب 32. نقص محمد الثياب 33. محمد اللحم 34. هرّ س محمد الثياب
	Verb + Subject + Object The subject is a human being who w kitchen. The object is an inanimate entity such 1. حكّ محمد الوعاء 2. خبّز محمد الخبز 3. خبّز محمد الخبوف 4. خزّن محمد الضيوف 5. خنّم محمد القمح 6. خبّز محمد القمح 7. محمد القمح 8. سنّق محمد القمح 9. شطَف محمد اللحام 11. صنّى محمد اللحم 12. خبخ محمد اللحم 13. حمد اللحم 14. طحّن محمد اللحم 14. طحّن محمد اللحم 15. طبّع محمد اللحم 16. طوّى محمد القمح 17. عجّن محمد القمح

4.2.28 Verbs of ingesting (class 28)

Class	These verbs refer to vital events of drinking and eating performed by	
Description	animate beings.	
Levin's class	Verbs of ingesting	
Vendler's class	Achievement verbs	
Transitive or intransitive	Transitive	
Class members	 ^akal الكل 'to eat; to take, take in, have (a meal, etc.)' bala^c بلغ 'to swallow, gulp (down), gobble, ingurgitate; to swallow up, engulf, gulf, absorb, take in, devour' jara^c جرع 'to swallow, gulp (down), drink, pour down, quaff' <u>hasā</u> 'حسا 'to drink, sip' <u>dāq</u> إذ 'to taste, sample' <i>rašaf</i> رشف 'to sip, sup, suck; to drink (up)' <i>rada</i>^c رشف 'to suck (at the breast), suckle' šarib 'to drink' <i>calak</i> 'to chew, masticate' <i>qarad</i> قرض 'to gnaw, nibble at, champ, bite, eat into, eat away, corrode' <i>qaram</i> 'š' to nibble at, gnaw' 	

	to gnaw, nibble at, b [•] قضم 12. <i>qaḍam</i>	ite, champ'
	to cut off, sever; to lo 'to cut off, sever; to lo	p off, chop off'
	14. <i>lāka لاك</i> 'to chew, masticate'	
	to lick; to lick up, lap نحس 15. <i>lahas</i>	(up)'
	to lick; to lick up, lap (¹⁶ . <i>la^cig</i> ¹	up)'
	to devour, gobble, swal 'to devour, gobble, swal	llow up'
	18. lahim is 'to devour, gobble, swa	llow up, gorge, ingurgitate.
	gormandize'	r, 5- 8-, 8- 8- 8- 8- 8- 8- 8- 8- 8- 8- 8- 8- 8-
	to suck, suck up, soak ، مصّ ito suck, suck up, soak	up, absorb; to sip'
	to chew, masticate' مضغ 20. madag مضغ	
	'to drink' نهل 21. <i>nahal</i>	
	22. <i>haḍam</i> هضم 'to digest (food)'.	
The syntactic	Verb + Subject + (Object)	
frame of		
transitives		
The nature of	The subject is an animate being who h	as the ability to eat and drink.
the subject		
The nature of	The object is food or drink.	
the object		
Examples	 أكمل محمد الطعام 	12. قضّم محمد الطعام
-	2 بلّع محمد الطعام	13 قطم محمد الطعام
	3 جرّع محمد الشراب	14. لاك محمد الطعام
	4. حسّا محمد الطعام	15. لحَس محمد الطعام
	 ذاق محمد الطعام 	16. لعق محمد الطعام
	6. رشّف محمد الشرّاب	17. لقَم محمد الطعام
	7. رضّع محمد الشراب	18. لهم محمد الطعام
	8. شرب محمد الشراب	19. مصّ محمد الشرّاب
	9 علَّك محمد الطعام	20 مضبع محمد الطعام
	10. قرّض محمد الطعام	21. نهل محمد الطعام
	11. قرّم محمد الطعام	22. هُضَم محمد الطعام

4.2.29 Verbs of the five senses (class 29)

Class	These verbs denote that the subject is characterized by attributes that belong	
Description	to the five senses, including sight, hearing, touch, smell, and taste.	
Levin's class	Verbs of perception	
Vendler's class	State verbs	
Transitive or	Transitive	
intransitive	Intransitive	
Class members	1. baṣar (ڊ) بصر 'to see; to look at, set eyes on, lay eyes on, catch sight of,	
	descry'	
	2. <i>jass</i> جسّ 'to touch, feel, handle'	
	3. <i>ḥass</i> حسّ 'to touch'	
	4. <i>dāq</i> ذاق 'to taste, sample'	
	5. $ra^{\circ}\bar{a}$ c^{\dagger} c^{\dagger	
	discern'	
	6. <i>sama</i> ^c سمع 'to hear'	

	7. <i>šamm</i> شم 'to smell, sniff, scent, nose, snuff'	
	8. lamas لمس 'to touch, feel, handle, finger'	
	9. mass مسنّ 'to touch, feel, handle, finger'.	
The syntactic	Verb + Subject + Object	
frame of		
transitives		
The nature of the subject	The subject is an animate being who has the ability to see, hear, feel, taste, and smell.	
The nature of	The object can be an animate being or an inanimate object that can be	
the object	recognized.	
Examples	 بصر محمد الأشياء 	
	2. جسّ محمد الأشياء	
	3. حسّ محمد الأشياء	
	4. ذاقِ محمد الطعام	
	 رأى محمد الأشياء 	
	6. سمِع محمد الأصوات	
	7 شمّ محمد الطعام	
	8. لمَس محمد الأشياء	
	9. مسّ محمد الأشياء	

4.2.30 Verbs of ruling and government (Class 30)

ClassThese verbs denote that the subject is characteriDescriptionwho has the authority to make decisions.Levin's classJudgement verbsVendler's classState verbsTransitive orTransitiveIntransitiveIntransitiveClass members1. $^{o}amar$ $\int_{a} (to order, enjoin, direct, instruct, go instructions (to); to charge (with), require; the struct of the s$	zed by attributes of a ruler	
Descriptionwho has the authority to make decisions.Levin's classJudgement verbsVendler's classState verbsTransitive or intransitiveTransitiveClass members1. $^{2}amar$ 1 to order, enjoin, direct, instruct, g instructions (to); to charge (with), require; t 2. $hakam$ 2 to rule, reign, dominate, have per manage, control, regulate; to order, direct, disband, break up (a pa company, etc.)'4. xala c displace; to depose, oust, remove (from an displace; to succeed, be the successor of; t		
Levin's classJudgement verbsVendler's classState verbsTransitive or intransitiveTransitiveIntransitiveIntransitiveClass members1. $^{2}amar$ 1 (to order, enjoin, direct, instruct, g instructions (to); to charge (with), require; t 2. $hakam$ $\sim ^{2}$ (to rule, reign, dominate, have pomanage, control, regulate; to order, direct, direct, dissolve, disband, break up (a pa company, etc.)'4. xala $^{\circ}$ (to depose, oust, remove (from an displace; to dethrone'5. xalaf (to succeed, be the successor of; to to succeed, be the successor of; to		
Vendler's class State verbs Transitive or intransitive Transitive Class members 1. °amar أمر 'to order, enjoin, direct, instruct, g instructions (to); to charge (with), require; t 2. hakam 'to rule, reign, dominate, have pomanage, control, regulate; to order, direct, direct, dissolve, disband, break up (a pactor company, etc.)' 4. xala ^c 'to depose, oust, remove (from an displace; to dethrone' 5. xalaf 'to succeed, be the successor of; to succeed, be the success		
Transitive or intransitiveTransitiveIntransitiveIntransitiveClass members1. $^{\circ}amar$ $^{\circ}$		
intransitive Class members 1. °amar أمر 'to order, enjoin, direct, instruct, g instructions (to); to charge (with), require; t 2. hakam حكم 'to rule, reign, dominate, have po- manage, control, regulate; to order, direct, d 3. hall حلّ 'to dissolve, disband, break up (a pa- company, etc.)' 4. xala ^c خلغ 'to depose, oust, remove (from an displace; to dethrone' 5. xalaf خلف 'to succeed, be the successor of; t		
Class members 1. ^a amar أمر 'to order, enjoin, direct, instruct, g instructions (to); to charge (with), require; t 2. <i>hakam</i> حكم 'to rule, reign, dominate, have po- manage, control, regulate; to order, direct, c 3. <i>hall</i> 'Lo dissolve, disband, break up (a pa company, etc.)' 4. <i>xala</i> ^c خلع 'to depose, oust, remove (from an displace; to dethrone' 5. <i>xalaf</i> خلف 'to succeed, be the successor of; t	Intransitive	
replace, take the place of, substitute for, dis 6. ra ² as رأس 'to head, lead, be in charge of; to chairman of; to be or become the president 7. ra ^c ā رعى 'to govern, rule, regulate, control, 8. sās ساس 'to govern, rule; to administer, mar handle, conduct' 9. sann (قانوناً) 'to enact or pass (a law); to l establish, introduce, prescribe' 10. dabat ضبط 'to control, check, curb, contain, (back), keep under control, keep down, keep 11. tarad طرد 'to expel, deport, banish, exile'	give orders or directions or o dictate; to prescribe' ower over; to govern, lictate; to enjoin, decree' rliament, an organization, a office), dismiss, discharge, o follow, come after; to place, supplant, supersede' preside over; to chair, be the (head, chief) of' apply to, be applicable to' lage, direct, run, lead, egislate, make laws; to restrain, subdue, hold p back'	

	12. <i>zalam</i> ظلم to wrong, oppress, tyrannize, aggrieve, maltreat, treat		
	unjustly (inequitably, tyrannically, badly)'		
	13. <i>cadal عدل</i> 'to act justly, establish justice, be just, be fair, be equitable,		
	give a just judgment'		
	14. farad (فرض (على 'to impose (upon), enjoin (on), make something		
	someone's duty; to ordain, decree, prescribe; to dictate; to order' 15. $q\bar{a}d$ 'to lead; to guide, conduct, direct; to drive, steer, pilot'		
	16. <i>gadā</i> فضى 'to impose, enjoin, make incumbent (upon): to ordain. decree.		
	prescribe: to order, require: to necessitate, make necessary or requisite'		
	17. nafa is 'to banish, exile, expatriate, expel, relegate, deport'.		
The syntactic	Verb + Subject + Object		
frame of			
transitives			
The syntactic	Verb + Subject + Prepositional phrase		
frame of			
intransitives			
The nature of	The subject is a human being who has authority of some kind.		
the subject			
The nature of	The object refers to followers, decisions, or citizens.		
the object			
	 أمر الملك الوزير 	10. ضبّط الملك التشريعات	
	2. حكم الملكُ الرعيةَ	11. طرّد الملكُ الوزير	
	 حلّ الملكُ مجلسَ النواب 	12. ظلَّم الملك الرعية	
	 4. خلَع الملكُ الوزيرَ 	13. عدّل الملك حُكمه	
	 خاف الأمير الملك 	14. فرَض الملك القانون	
	 رأس الوزير الاجتماع 	15. قاد الملك الدولةً	
	 رعَى الملك الحفل 	16. قضَى الملك الحُكم	
	8. ساس الملك الدولة	17. نفَى الملكُ الوزيرَ	
	9. سنّ الملك التشريعات		

4.2.30 Verbs of the development of life (class 31)

Class	These verbs describe the developments that occur in the life stages of		
Description	animate beings.		
Levin's class	Verbs of existence		
Vendler's class	Accomplishment verbs		
Transitive or	Intransitive		
intransitive			
Class members	 balaġ بلغ 'to attain puberty, reach sexual maturity, be or become sexually mature (marriageable, pubescent); to attain manhood, become an adult' hayiy جبي 'to live, be alive, exist, subsist; to lead a life' rabā ريا 'to grow up' rašad رشد' to be in legal age, adulthood' šabb 'شبّ 'to grow up, become a youth, become a young man' cāš نقي 'to live, be alive, exist, subsist; to lead a life' gratu 'to grow up, become a youth, become a young man' fatiy غاش 'to be youthful, young, adolescent' fatis iduu 'to die' 		

	 9. fanā فنى 'to perish, pass away, cease to exist, become extinct; to evanesce, vanish; to be consumed, exhausted, worn-out' 10. kabir كبر 'to grow old, get old(er), be advanced in years; to grow up' 11. māt مات 'to die down, subside, abate, let up' 12. naša^o نشأ 'to grow up; to grow, develop, evolve; to thrive, flourish' 13. harim هرم 'to age, grow old; to become decrepit' 14. walad (ألحامل) 'to give birth (to), bear'. 	
The syntactic frame of intransitives	Verb + Subject	
The nature of the subject	The subject is an animate entity who passes through stages of life.	
Examples	1. بلغ محمد 2. حيي محمد 3. ربا محمد 4. رشد محمد	8. فطّس محمد 9. فنِي محمد 10. كبِر محمد 11. مات محمد
	5. شَبَّ محمد 6. عاش محمد 7. فتِي محمد	12. نشَأ محمد 13. هرم محمد 14. ولَد محمد

4.2.32 Verbs of uttering (class 32)

Class	These verbs describe the style of saving that is said by the subject, who		
Description	should be a human being.		
Levin's class	Say verbs		
Vendler's class	Activity verbs		
Transitive or intransitive	Transitive		
Class members	 bāḥ (السَرَ) باح (السَرَ) (السَرَ) bāḥ (السَرَ) بن 'to be or become known, revealed, disclosed, divulged, uncovered; to reveal, disclose, divulge, uncover, unearth, make known' batt بن 'to broadcast, to radio-broadcast, transmit, air; to telecast, televise' basat بسط 'to set forth, present, lay open; to explain, expound, elucidate' talā 'to read, recite' ḥakā حک 'to tell, relate, narrate, recount, recite, report, give an account of' xatab خطب 'to make or deliver an address or speech; to speak; to address; to preach, deliver a sermon' <u>d</u>akar نكر 'to mention, make mention of, refer to, make reference to, name, specify, state, indicate, point out to, cite, report' rawā دكر 'to ransmit, pass on; to quote (from), cite (from)' saja ' w' to relate, narrate, recite, recount, report, give an account of, to transmit, pass on; to quote (from), cite (from)' saja ' w' to relate, narrate, recite, recount, report, give an account of, we have ' to relate, narrate, recite, recount, report, give an account of; to transmit, pass on; to quote (from), cite (from)' saja ' w' to relate, narrate, recite, recount, report, give an account of, tell' 		

	clear, describe, depict, discuss'		
	12. ša car شعر 'to poetize, versify, compose poetry'		
	13. carad عرض 'to offer, suggest, pr	opose'	
	14. qāl قال 'to say, tell; to speak; to u	atter; to state, express in words'	
	to praise, commend, l ، قرظ to praise, commend, l	aud, extol, eulogise'	
	 16. <i>qaṣṣ قص</i> 'to narrate, relate, recount, tell, recite, report, give an account of' 17. <i>lafaẓa</i> (لَعَظَ (كَلِمَةً) 'to pronounce, utter, enunciate, say; to phonate, vocalize; to speak, talk' 18. <i>madah مدح</i> 'to praise, commend, laud, extol, eulogize' 19. <i>nabas مدح</i> 'to utter, say, speak' 20. <i>našar نشر</i> 'to spread, propagate, disseminate; to circulate; to promulgate, publicize, circularize, publish, make public; to popularize' 		
	to pronounce, ut الطق phanata up and in a magle to li	ter, enunciate, say, to articulate, to	
	phonate, vocalize, to speak, tak	vacant proise land avagainaly, avtal	
	to snower with extrav highly: to rayo talk at random?	vagant praise, laud excessively, extor	
	highly; to rave, talk at random'		
The exptentio	Verb + Subject (+ Object)	Jie, mumur, mutter .	
fromo of	Verb + Subject (+ Object)		
transitivas			
The nature of	The subject is a human being who has	the ability to express himself	
the subject	The subject is a numan being who has	the ability to express minsen.	
The nature of	The object refers to what is said by the	subject	
the object	The object fefers to what is said by the		
Examples	1 باح محمد السرّ	13 عرّض محمد الخير	
Enumpies	2 بتُ محمد الخبر	14 قال محمد الخبر	
	2 3 يُسَط محمد الخير	15 قرَّظ محمد الشعر	
	4 تلا محمد الخبر	16. قصّ محمد الخبر	
	5 حكى محمد الخبر	17. لفظ محمد الكلمة	
	6. خطب محمد الخبر	18. مدّح محمد أخاه	
	7. ذكّر محمد الخبر	19. نبَس محمد الكلمة	
	 ٤. شعر محمد قصيدة 	20. نشّر محمد الخبر	
	9. روَى محمد الخبر	21. نطَق محمد الكلمة	
	10. سجّع محمد الخبر	22. هرَف محمد الخبر	
	11. سرَدَ محمد الخبر	23. همّس محمد الكلمة	
	12. شررَح محمد الخبر		

4.2.33: Verbs of accepting (class 33)

Class	These verbs denote that the subject accepts or allows another person to do	
Description	something, or accepts an order, decision or request.	
Levin's class		
Vendler's class	State verbs	
Transitive or	Transitive	
intransitive	Intransitive	
Class members	1. <i>°axad</i> (ب) أخذ (بـ) to accept'	
	2. <i>°adin أذن</i> to allow, permit, give permission to; to license, authorize'	

	$3 tabic \cdot ij 'to obey follow pursue$	adont'	
	5. $table \stackrel{\circ}{\leftarrow}$ to obey, follow, pursue, adopt		
	4. $tartak(-) = 0$ to allow, let, permit		
	5. radiy ($(-, -)$) to be or become satisfied (with); to be or		
	(to) accept (to), agree (to), consent (to), assent		
	(10), accede (10), approve (01), subscribe (10), sanction, OK		
	to allow, permit, let, give permitssion (10), to admit,		
	to authorize, sanction, warrant, le	egitimate, legitimize, legalize; to	
	$7 = aabil()$ $\sqrt{3}$ (to accord (to) across (to) consent (to) accord (to)		
	i. $qaou(-i) = to accept (to), agree (to), consent (to), assent (to$		
	approve (of), subscribe (to), OK,	sanction; to take; to settle for, content	
	oneself with; to admit		
	8. $qani^{c}$ (-,) $iii be or become content with, satisfied with; to content$		
	oneself with, settle for'.		
The syntactic	Verb + Subject + Object		
frame of			
transitives			
The syntactic	Verb + Subject (+ Prepositional phrase)		
frame of			
intransitives			
The nature of	The subject is a human being who has the ability to accept or allow for		
the subject	another person to do something.		
The nature of	The object can be a human being or an order, decision or request.		
the object			
The nature of	The prepositional phrase is headed by the preposition $bi \rightarrow$ 'with' followed		
the	by a noun that refers to an order or a matter.		
Prepositional			
phrase		,,	
Examples	Transitives	 تبع محمد الأمر 	
		 يترك محمد الأمر 	
	Intransitives	 أخذ محمد بالأمر 	
		4. اذِن محمد بالإمر	
		 قنع محمد بالأمر 	
		6. رضِي محمد بالامر	
		7. سمَح محمد بالأمر	
		8 قبل محمد بالأمر	

4.2.34 Verbs of refusing and disobedience (class 34)

Class	These verbs denote that the subject refuses to do something (does not accept		
Description	a thing, person, or matter). The subject should be an animate being.		
Levin's class			
Vendler's class	State verbs		
Transitive or	Transitive		
intransitive	Intransitive		
Class members	1. <i>°abā أ</i> بى 'to refuse, decline, reject, turn down'		
	2. <i>jaḥad</i> جحد 'to deny, disown, disavow, disclaim; to repudiate, refuse,		
	reject'		
	3. <i>ḥani<u>t</u> حنث 'to break (or violate) one's oath; to perjure'</i>		

	to turn down, reject, ref	use, decline'	
	5. <i>radal</i> (ن خُل to reject, discard cast off or away'		
	6. rafad فضن 'to refuse, reject, turn down, decline, repudiate, disapprove;		
	to denv: to veto'		
	7. sadd $\Delta \omega$ 'to repel reject refuse deny'		
	8 $cas\bar{a}$ as it o breach break commit a breach of		
	9. <i>nahad</i> ننذ to discard reject'		
	10 $nafar(ic)$ is to avoid shun est	chew turn away from keen away	
	from'	enew, turn away nom, keep away	
	to break violate infringe: to renege break (go back on fail)		
	to keep) one's promise or word or commitment: to break faith'		
	$12 \text{ nakil}(i) \approx i c$) (K) 'to abstain from refrain from forbear desist from		
	ו בבט (على: م) to abstall from, refrain from, fordear, desist from,		
The exertentia	Varb + Subject + Object		
frame of	Verb + Subject + Object		
transitives			
The survey of a			
The syntactic	verb + Subject (+ Prepositional phrase	e)	
Irame of			
intransitives			
The nature of	The subject is a numan being who has the ability to refuse to do something.		
the subject			
The nature of	The object can be work, issue, opinion, promise, or matter.		
the object			
The nature of	The prepositional phrase is headed	about' عن about' about'	
the	followed by a noun that refers to a wor	rk, issue, opinion, promise, or matter.	
Prepositional			
phrase		<u> </u>	
	Transitives	 جحد محمد الحق 	
		2. حنث محمد وعده	
		3. رد محمد الشكوى	
		4. رذِل محمد الأمر	
		 رفض محمد الحق 	
		6. عصبی محمد صدیقه	
		7. نبذ محمد صديقه	
		8. نکث محمد و عده	
	Intransitives	 أبَى محمد عن قول الحق 	
		10. صِدّ محمد عن الحق	
		11 نفر محمد عن الحق	
		12. نكِل محمد عن الحق	

4.2.35 Verbs of preventing and prohibition (class 35)

Class	These verbs denote that the subject prevents an animate object from doing
Description	something.
Levin's class	Keep verbs
Vendler's class	Achievement verbs
Transitive or	Transitive
intransitive	

Class members	1. <i>°asar</i> اسر 'to capture, take prisoner, ar	rest, apprehend, jail; to intern; to
	bind, fetter, shackle, chain'	
	2. <i>habas</i> جبس 'to withhold, hold (back), r	restrain, detain; to retain; to
	confine, constrain; to obstruct, block,	bar; to debar; to prevent, hinder'
	3. <i>ḥajab</i> حجب 'to veil, cover, curtain; to	hide, obscure, blind, wrap,
	(en)shroud; to cloak, mantle; to make	invisible; to obstruct'
	4. <i>hajar</i> حجر 'to limit someone's legal co	ompetence; to place under
	guardianship; to interdict, prohibit (fr	om); to prevent (from)'
	5. <i>haram</i> حرم 'to deprive of, dispossess of	of, strip of, take away from; to
	disentitle; to deny, refuse (to grant), w	vithhold from; to debar, exclude'
	6. <i>hadar</i> حضر 'to ban, prohibit, forbid, in	nterdict, proscribe, enjoin, bar,
	outlaw; to taboo; to embargo'	
	7. $dara^{\circ}$ درأ 'to ward off, parry, fend off,	stave off, keep off, avert, turn
	away, repel, repulse, drive back'	
	8. rabat ربط to bind, tie (up), fasten, ma	ke fast, attach, moor; to connect,
	link, join, unite, attach, couple; to tie	(in); to ligate; to correlate'
	9. rada ^c (عن) 'to deter (from), disco	urage (from), prevent (from),
	keep (from), restrain (from), hold bac	k (from), inhibit (from)'
	10. zajar زجر 'to restrain, check, hold bac	k, prevent, deter'
	11. sajan سجن 'to jail, imprison, lock up, i	incarcerate, intern, confine,
	detain, keep or hold in custody'	
	12. sadd صد 'to repel, repulse, keep off or	back, hold off, avert, turn away;
	to check, rebuff, stop; to force out, sh	ut out, foreclose; to hinder'
	13. tarad طرد to drive away or out, expel	, throw out, force out, kick out,
	put out, oust, evict, eject, dismiss; to o	drive back, repel, repulse'
	14. <i>cāa</i> عاق 'to hinder, obstruct, block, sta	and in the way of, make difficult:
	to delay, slow: to prevent, hold (back)). countercheck: to frustrate'
	15. gat ° قطع 'to stop, suspend, discontinu	e. cease, halt, interrupt, break.
	cut, end, put an end to, terminate'	
	16. <i>aama</i> ^c فمع 'to curb. restrain. repress. s	suppress, subdue: to crush, quash.
	quench, squelch, put down, stifle, squ	ash. stifle. extinguish'
	17. <i>kabat كيت</i> 'to suppress, repress, subdu	e. stifle, restrain, inhibit, keep
	(back), hold (back), withhold, check,	curb. bridle. contain'
	18. <i>kabah کت</i> 'to rein in bridle' to check	curb keep under control keep
	down, restrain, suppress, repress, subo	due prevent to brake'
	19 kaff کف 'to prevent from keep from r	restrain from hold back from
	inhibit from dissuade from'	
	20 laiam \downarrow 'to bridle rein in restrain c	urb check control hold (back)
	keen down keen back'	
	$21 mana^{\circ}$ ito prevent hinder stop to	deter (from): to bar block: to
	forbid: to deny withhold from: to den	rive of: to debar exclude'
	22 nabad in 'to discard cast away throw	y away fling away reject
	remove eliminate dismiss abandon	renounce forsake give un dron'
	23 nagad is to revoke repeal rescind	countermand annul nullify
	abrogate invalidate cancel to reverse	e overrule: to quash vacate?
	$24 \ naha$ i to forbid prohibit interdict	han proscribe to restrain
	nrevent'	, oan, prosentoe, to restrain,
The syntactic	prevent. Verh \pm Subject \pm Object	
frame of	$v \in U + Subject + Object$	
transitives		
uansiuves		

The nature of	The subject can be an animate or an inanimate entity that can prevent	
the subject	something from doing something.	
The nature of	The object is an animate or inanimate entity.	
the object		
Examples	 أسر محمدٌ الخصمَ 	13. عاق محمدٌ الخصمَ
	2. حبّس محمدٌ الخصمَ	14. قطّع محمدٌ الخصمَ
	3. حجَب محمدٌ الخصمَ	15. قمَع محمدٌ الخصمَ
	4. حجَر محمدٌ الخصمَ	16. كَبَتْ محمدٌ الخصيمَ
	 حرام محمدً الخصم 	17. كبَح محمدٌ الخصمَ
	6. حضّر محمدٌ الخصمَ	18. كفٌّ محمدٌ الخصمَ
	7. درَأ محمدٌ الخصمَ	19. لجَم محمدٌ الخصمَ
	 ٤. ردع محمد الخصم 	20. منّع محمدٌ الخصيمَ
	 زجر محمد الخصم 	21. نبَذ محمدٌ الخصمَ
	10. سجَن محمدٌ الخصمَ	22. نقَض محمدٌ الخصيمَ
	11. صدّ محمدٌ الخصمَ	23. نهَى محمدٌ الخصبَمَ
	12. طررد محمد الخصم	

4.2.36 Verbs of occurrence and progressing (class 36)

Class	These verbs denote that the subject is characterized by attributes that refer to	
Description	starting, progressing, or delaying an event.	
Levin's class	Verbs of occurrence	
Vendler's class	State verbs	
Transitive or	Transitive	
intransitive	Intransitive	
Class members	1. <i>°atal أنثل</i> to be deep-rooted, firmly established'	
	2. <i>°ajil</i> 'to tarry, linger; to be late'	
	3. $b\bar{a}t$ \downarrow 'to become, come to be, grow (into), turn (into); to be; to reach a stage where, get to a point where'	
	 4. bada² بن¹ 'to begin, start, set in, commence, originate, get under way, come into being, come into existence, arise, rise, spring up, crop up' 5. tamm نتم 'to be complete, full, whole, entire, total; to be completed, 	
	concluded, finalized, terminated; to end'	
	 hadat حدث 'to happen, take place, occur, come to pass, transpire, go on, pass' 	
	 7. sabaq سبق 'to precede, antecede, forego, be (go, come) before or ahead of; to antedate, predate, be earlier than, happen before' 	
	8. <i>salaf</i> سلف 'to pass, elapse, go by, be past, be over, be bygone; to precede, antecede'	
	 <i>šara</i>^c شرع 'to begin, start, commence, set about, set to; to go into, enter upon' 	
	10. <i>gabar</i> غبر 'to pass, elapse, go by; to be past, elapsed, bygone'	
	 12. kamal کمل 'to be or become complete, full, total, perfect, consummate; to be or become completed, finished, concluded, consummated' 13. waqa ° فغ 'to happen, take place, occur, come to pass, transpire'. 	
The syntactic	Verb + Subject	
frame of		
---------------	---	-----------------
intransitives		
The nature of	The subject refers to a matter or an ev	ent.
the subject		
Examples	1. أنَّل الأمر	8. سلَّف االأمر
_	 أجَل الأمر 	9. شرّع الحدث
	3. بات الأمر	10. غبَر الزمن
	4. بدَأ الحدث	11. قدُّم الأمر
	5. تمّ الحدث	12. كمَلُ الحدث
	6. حدّث الأمر	13. وقَع الحدث
	7. سبَق الحدث	-

4.2.37 Verbs of appearance (class 37)

· · · · · · · · · · · · · · · · · · ·	Г Г		
Class	These verbs denote a description in which the subject appears.		
Description			
Levin's class	Verbs of appearance		
Vendler's class	Accomplishment verbs		
Transitive or	Transitive		
intransitive	Intransitive		
Class members	 bān بان 'to appear, come out, come to light, show, emerge, surfacereveal itself; to be or become visible, manifest, distinct, clear' batag 'to spout spurt squirt jet shoot out cause to flow or gush 		
	out: to extrude force out push out expel discharge throw out?		
	 badā יבי 'to appear, come out, come into view, emerge, manifest itself; to be or become visible, apparent, clear; to seem' 		
	 baraz برز 'to appear, come out, come into view, show, arise, spring up, manifest itself; to stand out; to come to the fore' 		
	 bazaġ بزغ 'to rise (sun, etc.); to dawn (day, era, etc.); to break (forth); to appear, emerge, come out' 		
	6. <i>jalā</i> → 'to be or become clear, plain, distinct, manifest, patent, evident, obvious, explicit, unequivocal; to manifest itself, surface, come to light'		
	 <i>jahar</i> جهر 'to appear, come out, show, reveal itself; to be or become known, revealed, disclosed; to be made public; to spread, circulate' 		
	 <i>hadar</i> حضر 'to attend, be present, be there; to report (for duty, to a certain place), present oneself; to come, appear; to reach, get to; to visit' 		
	9. <i>sața ^c</i> سطع 'to spread, diffuse, emanate; to rise'		
	10. <i>šahid</i> شهد 'to witness, see (personally or for oneself); to attend, be present at'		
	11. $tafa$ لطف 'to emerge, surface, crop up, rise, spring up, appear'		
	to emerge, rise, show, appear, surface' طل to emerge, rise, show, appear, surface'		
	13. <i>to</i> rise; to appear, come out, come into view or sight, show, emerge, surface'		
	14. <i>zahar</i> ظهر 'to appear, come out, come to light, show, emerge, crop up, surface, rise, arise, develop, reveal itself; to be or become apparent'		
	to appear, come into view, come out, present itself? مثل to appear, come into view, come out, present itself?		
	16. <i>naba</i> ^c نبع 'to well, well up, well out, gush out, gush forth, pour out, pour forth, stream, flow, flow out'		

	17. nabag نبغ 'to arise, emerge, appear, show'		
	18. nataj نتج 'to result from, ensue from, proceed from, follow from, arise		
	from, derive from, originate from, issue from, grow out of, come out of'		
	19. wadah وضح 'to be or become clear, plain, distinct, manifest; to clarify,		
	clear up; to appear, show; to follow clearly (from); to be clarified'		
The syntactic	Verb + Subject		
frame of			
intransitives			
The nature of	The subject can be an animate or inanimate entity that can appear clearly.		
the subject			
Examples	1. بان القمر	11. طفا السمك	
	2. بثق الضوء	12. طُلَّ محمدٌ	
	3. بدا القمر	13. طلّع الفجر	
	4. برَز الضوء	14. ظهَر محمدٌ	
	5 بزَغ الفجر	15. مثَّل محمدٌ	
	6. جلا الضوء	16. نبّع الماء	
	7. جهَر الصوت	17. نبّغ الماء	
	8. حضّر محمدٌ	18. نتَّج الثمر	
	9 سطّع الضوء	19. وضبح الفجر	
	10. شهد محمدٌ		

4.2.38 Verbs of disappearing and ending (class 38)

Class	These verbs denote a description in which the subject disappears or ends	
Description	something or makes something disappear.	
Levin's class	Verbs of disappearance	
Vendler's class	Accomplishment verbs	
Transitive or	Transitive	
intransitive	Intransitive	
Class members	1. <i>°afal أفل</i> 'to set, go down, sink; to decline, fade, dim'	
	2. $b\bar{a}d$ $\pm to$ perish, pass away, become extinct, cease to exist'	
	3. $xab\bar{a} \leftrightarrow$ 'to go out, die, be extinguished; to abate, subside, cool off; to	
	fade, tarnish, grow dull'	
	4. xaraj خرج 'to go out, come out, emerge, issue, egress; to walk out,	
	drive out, ride out, move out, exit; to leave, depart, go away'	
	5. xasif خسف 'to sink down, fall down, collapse, cave in, fall in, give way;	
	to subside'	
	6. <i>xafiy</i> خفي 'to disappear, vanish; to hide (oneself), conceal oneself; to be	
	hidden, concealed, obscured; to be unknown; to remain (a) secret'	
	7. xamad خمد 'to go out, die, be extinguished; to abate, subside, let up, die	
	down, die away, fade away, cool off, calm down'	
	8. <i>zāl</i> زال to disappear, vanish, clear (away), go (away); to abate; to cease	
	to exist; to end, come to an end, terminate'	
	9. $\dot{g}\bar{a}b$ 'to disappear, vanish; to hide (oneself), keep oneself out of	
	sight; to be or become hidden, concealed, sheltered, mantled, covered'	
	10. <i>ġār</i> غار 'to sink, sink down, sink in, fall in, cave in, collapse, go down,	
	subside, give way'	

	to be or become empty, void, vacant, unoccupied' فرغ to be or become empty, void, vacant, unoccupied'		
	12. kaman کمن 'to hide; to be hidden, concealed, latent; to lie in, be in,		
	exist in, be found in'		
	13. māt مات 'to die, expire, pass away'		
	14. nadab نضب 'to run out; to be or become exhausted, depleted; to peter		
	out; to dwindle, drain away, fade; to diminish, decrease, lessen'		
	15. nafid نفد 'to run out; to be or become exhausted, depleted, used up; to		
	come to an end'		
	16. <i>nafaq نف</i> ق 'to run out; to be exhausted, used up'.		
The syntactic	Verb + Subject		
frame of			
intransitives			
The nature of	The subject can be any entity that can disappear.		
the subject			
Examples	1. أفَّل النجم	9. غاب محمد	
_	2. باد الغنم	10. غار الماء	
	3. خبا الماء	11. فرغ الماء	
	4. خرَج محمدٌ	12. كمَن الماء	
	5. حسِف الشيءُ	13. مات محمدٌ	
	6. خفِي الماء	14. نضّب الماء	
	7. خمَدْت النار	15. نفد الماء	
	8. زال الماء	16. نفَقت الماشية	

4.2.39 Verbs of bodily movements (class 39)

Class	These verbs describe the bodily movements of an animate being.	
Description		
Levin's class	Curtsey verbs	
Vendler's class	Achievement verbs	
Transitive or	Intransitive	
intransitive		
The syntactic	1. barak برك 'to kneel down; to lie down, rest'	
frame of	2. <i>jatā</i> ختا 'to kneel, rest or fall on the knees; to genuflect, bend the knee'	
intransitives	3. <i>jalas</i> جلس to sit down, sit, take a seat, seat oneself	
	4. <i>ḥanā</i> (رَأْسَه أَو ظَهْرُه) 'to bow, bend (the head or body), stoop; to	
	hunch'	
	5. raqad رقد 'to lie down, repose, rest'	
	6. raqaṣ رقص 'to dance'	
	7. $raka^{c}$ رکع 'to kneel (down); to bow (down)'	
	8. sajad سجد 'to prostrate oneself (in worship), genuflect, kneel (with the	
	forehead touching the ground), bow (down)'	
	9. <i>°aţir</i> عثر 'to tumble'	
	to sit down, sit, take a seat, seat oneself' فعد 'to sit down, sit, take a seat, seat oneself'	
	to incline, slope, slant, skew, tilt, tip, cant; to be or become مال 11. māl	
	inclined, sloping, slanting, oblique'	
	12. <i>nām</i> نام 'to sleep'	
	13. nahad نهد 'to rise, get up'	

	14. nahad نهض 'to rise, get up'		
	to stand up, rise, get up, stand erect' وقف to stand up, rise, get up, stand erect'		
	16. <i>wați</i> ² وطِئ 'to tread on, step on, walk on, foot on; to set foot on; to		
	trample down, trample underfoot'.		
The syntactic	Verb + Subject (+ Prepositional phrase)		
frame of			
intransitives			
The nature of	The subject is an animate entity who can move its body parts.		
the subject			
The nature of	The prepositional phrase is headed by the preposition ^c ala على 'on' followed		
the	by a noun that refers to a place.		
Prepositional			
phrase			
Examples	 برك الجَمَلُ على الأرض 	9. عثر محمد على الأرض	
	 جأًا الجَمَلُ على الأرض 	10. قعَد محمدٌ على الأرض	
	 جأس محمدٌ على الأرض 	11. مال محمدٌ على الأرض	
	 4. حنّى محمدٌ على الأرض 	12. نهد محمدٌ على الأرض	
	 رقد محمد على الأرض 	13. نهَض محمدٌ على الأرض	
	 رقص محمدٌ على المسرح 	14. وقَف محمدٌ الأرض	
	 ركم محمدٌ على الأرض 	15. وطِئ محمدٌ على الأرض	
	 8. سجد محمد على الأرض 		

4.2.40 Verbs of taking and giving (class 40)

Class	These verbs denote an action in which the subject takes, gives or brings		
Description	something or someone.		
Levin's class	Bring and take verbs		
	Give verbs		
Vendler's class	Accomplishment verbs		
Transitive or	Transitive		
intransitive			
Class members	1. ^a xad أخذ 'to take; to pick up; to take in; to receive, obtain'		
	2. $j\bar{a}^{\circ}$ (-) + 'to bring, bring forward, fetch, get, produce, advance,		
	present, introduce'		
	3. <i>jalab</i> جلب 'to bring, fetch, get, bring forward, introduce'		
	4. $h\bar{a}z$ 'to hold, possess, have, take or gain possession of, lay hands		
	on, occupy, seize; to acquire, win, gain; to achieve, attain'		
	5. <i>hadar حد</i> ر to bring down, lower, drop'		
	6. haşal (حصل (على 'to obtain, get, acquire, win, gain, earn, receive; to		
	attain, achieve; to possess, own, have'		
	7. <i>haziy</i> (ب) خطي (to acquire, get, obtain, win, gain; to attain, achieve'		
	8. dān دان 'to borrow, contract (raise, take up) a loan, make or incur		
	debts'		
	9. raji c (ب) رجع 'to entail; to bring about; to bring in, return, yield,		
	produce, pay'		
	10. radd رد 'to return, give back, restore, restitute; to send back; to bring		
	back, take back, turn back; to put back, lay back, replace, place back'		
	11. saraq سرق 'to steal, pilfer, pinch, filch, purloin, abstract, thieve; to		
	burglarize, housebreak; to rob, rip off; to hold up'		

	12. salab سلب 'to steal, rip off, rob, abstract, fleece, plunder, pillage, rifle,	
	loot, spoil, ransack, maraud; to strip of, dispossess of'	
	to win, gain, obtain, get, attain, achieve' ظفر to win, gain, obtain, get, attain, achieve'	
	14. <i>qarad</i> قرض to lend, loan, advance (money to)'	
	15. kasab کسب 'to gain, win, profit, earn, get, obtain, acquire, attain, reap,	
	harvest'	
	16. manah منح 'to grant (to), give (to), donate (to), award (to), confer upon, bestow upon; to endow with, gift with; to contribute (to)'	
	17. <i>nāl</i> نال 'to obtain, get, acquire, win, gain, earn, receive; to attain, achieve, accomplish, realize'	
	18. <i>nahal</i> نحل 'to present with, gift with, make or give a present (gift,	
	donation) to, donate (to), grant (to), give (to), bestow upon' 19. nafah نفت 'to give (to), grant (to), present (to or with), donate (to),	
	accord (to), award (to), bestow upon, confer upon'	
	20. nahab نهب 'to plunder, loot, pillage, rifle, spoil, despoil, spoliate,	
	maraud, rob'	
	21. wahab وهب to donate (to), grant (to), give (to), accord (to), award (to) hastow upon confer upon; to present (to or with)?	
The syntactic	(to), bestow upon, conter upon, to present (to or with).	
frame of	$\gamma c_{10} + Subject + Object$	
transitives		
The syntactic	Verb + Subject (+ Prepositional phrase)	
frame of		
intransitives		
The nature of	The subject is a human being that can take and hold something.	
the subject		
The nature of	The object can be an animate being or an inanimate object that can be taken	
the object	by the subject. The propositional physical is headed by the propositions hi , (with) or $c_{\alpha} d\bar{z}$.	
the	The prepositional phrase is headed by the prepositions $bl \rightarrow with of all a label{eq:branching}$ is followed by a noun that refers to something that can be taken	
prepositional		
phepositional		
Examples	Transitives	
I I	 أخذ محمد الشيء 	
	2. جلَب محمدٌ الشيءَ	
	3. حاز محمدٌ الشيءَ	
	4. حدّر محمدٌ الشيءَ	
	5. دان محمدٌ الشيءَ	
	6. ردّ محمدٌ الشيءَ	
	7. سرِق محمدٌ الشيءَ	
	8. سلب محمد الشيءَ	
	9_ فرَض محمد الشيءَ	
	10. كسّب محمد الشيءَ	
	11. منح محمد التسيءَ	
	12. نال محمد الشيء 12. ب 14 - 14 - 1	
	13. تحل محمد السبيء 14. نفَسم محمد الشبي مَ	
	14. لكن محمد الشيء 15. ذيب محمدً الشيءَ	
	11. بهب المسبرية 16. و هَب محمدٌ الشبريءَ	

Intransitives	17. جاء محمدٌ بالشيءِ
	18. حصّل محمدٌ علّى الشيء
	19. حظِي محمدٌ بالشيءَ
	20. رجَعَ محمدٌ بالشيءَ
	21. ظفَر محمدٌ بالشيءَ
	. -

4.2.41 Verbs of human sounds (class 41)

Class Description	These verbs describe the sounds that are produced by a human being.		
Levin's class	Verbs of manner of speaking		
Vendler's class	State verbs		
Transitive or	Intransitive		
intransitive			
Class members	 <i>ann نَنْ</i> 'to moan, groan, whimper, whine, wail' <i>bakā بكى</i> 'to cry, weep, shed tears, break or burst into tears' <i>xann نَخَ</i> 'to twang, snuffle, nasalize, speak through the nose, speak in a nasal tone or manner, speak nasally' <i>za ciq iq iq ij</i> 'to cry, yell, shout, scream, shriek, screech, squall' <i>šaxar نَخَن</i> 'to sonort' <i>šadā 'to sing</i>, chant, warble' <i>şāt i to sound</i>, ring, toll, knell, make a sound or noise; to phonate, produce speech sounds; to vocalize; to shout, cry, yell' <i>şāda ui to cry</i>, yell, shout, scream, screech, shriek, squall, whoop' <i>şadah ui to sing</i>, chant, warble' <i>saaap ui to cry</i>, yell, shout, scream, shriek, screech, squall' <i>safar ui to cry</i>, yell, shout, scream, shriek, screech, squall' <i>safar ui to cry</i>, yell, shout, scream, shriek, screech, squall' <i>safar ui to cry</i>, yell, shout, scream, shriek, screech, squall' <i>safar ui to cry</i>, yell, shout, scream, shriek, screech, squall' <i>safar ui to cry</i>, yell, shout, scream, shriek, screech, squall' <i>safar ui to cry</i>, yell, shout, scream, shriek, screech, squall' <i>safar ui to cry</i>, to laugh' <i>to buzz</i>, hum, drone; to whizz; to ring' <i>nadab ui to buzz</i>, hum, drone; to whizz; to ring' <i>nadab ui to mourn (for)</i>, lament, bewail, wail over, weep (for or over), moan' <i>nafax ui to blow</i>, puff' <i>hataf ui to blow</i>, puff' <i>hataf ui to shout, cry</i>, well; to exclaim, cry out' 		
	io. nuuur سر الاستار الاستار الاستار المستار المستار المستار المستار المستار المستار المستار المستار المستار ا		
frame of intransitives	vero + Subject (+ prepositional phrase)		
The nature of	The subject is a human being who can produce sounds.		
the subject			
The nature of the prepositional phrase	Optionally, the prepositional phrase is headed by the preposition $bi \rightarrow$ 'with' followed by a noun that refers to sound.		
Examples	10. صرّخ محمدٌ (بصوت مرتفع) 11. صفَر محمدٌ (بصوت مرتفع) 12. ضحِك محمدٌ (بصوت مرتفع) 12. ضحِك محمدٌ (بصوت مرتفع) 13. خنّ محمدٌ (بصوت مرتفع)		

4. زعِق محمدٌ (بصوتٍ مرتفع)	13. طنّ محمدٌ (بصوتٍ مرتفع)
5 _. شخَر محمدٌ (بصوتٍ مرتفع)	14. ناح محمدٌ (بصوتٍ مرتفع)
6. شدًا محمدٌ (بصوتٍ مرتفع)	15. ندُب محمدٌ (بصوتٍ مرتفع)
 صات محمدٌ (بصوتٍ مرتفع) 	16. نفَخ محمدٌ (بصوتٍ مرتفع)
 8. صاح محمد (بصوتٍ مرتفع) 	17. هتَّف محمدٌ (بصوتٍ مرتفع)
 9. صدّح محمدٌ (بصوتٍ مرتفع) 	18. هذر محمدٌ (بصوتٍ مرتفع)

4.2.42 Verbs of sounds made by animals (class 42)

Class	These verbs describe the sounds that are produced by a non-human animate		
Description	being.		
Levin's class	Verbs of sounds made by animals		
Vendler's class	State verbs		
Transitive or	Intransitive		
intransitive			
Class members	 <u>taġā</u> (الخروف) to bleat' <u>za</u>³ar (الأسد) to roar' <u>saja</u>^c (الأسد) to coo' <u>sahal</u> (الحمام) سجع (الحمام) <u>sahal</u> (محمد (to neigh, whinny' <u>c</u> awā (الكلب) to new, to new, niaow' <u>mā</u>² (الكلب) to mew, meow, miaow' <u>nabah</u> (الحلب 'to bark (at), bay (at)' <u>na</u> cab (الغراب) to caw, croak' <u>na</u> cab (الغراب) to hoot, whoop' <u>na</u> cap (الغراب) to bark, croak' <u>na</u> cap (الغراب) to caw, croak' <u>na</u> cap (الغراب) to bary, hee-haw' <u>nabal</u> (الحمام) to coo'. 		
The syntactic	Verb + Subject (+ Prepositional phrase)		
frame of			
intransitives			
The nature of the subject	The subject is an animal that can produce sounds.		
The nature of	Optionally, the prepositional phrase is headed by the preposition $bi \rightarrow$ 'with'		
the preposition	followed by a noun that refers to sound.		
Examples	 1. تُغَا الخروف (بصوت مرتفع) 2. زأر الأسد (بصوت مرتفع) 3. سجَع الحمام (بصوت مرتفع) 4. صهل الخيل (بصوت مرتفع) 5. عوَى الكلب (بصوت مرتفع) 6. ماء الهر (بصوت مرتفع) 7. نبَح الكلب (بصوت مرتفع) 8. نعَب الغراب (بصوت مرتفع) 9. نعَق الغراب (بصوت مرتفع) 10. نهق الحمار (صوت مرتفع) 		

Class	These verbs describe the appearance, grooming and dressing of a human		
Description	subject.		
Levin's class	Verbs of grooming bodily care		
	Verbs of dressing		
Vendler's class	Achievement verbs		
Transitive or	Transitive		
intransitive			
Class members	 halaq لفي 'to shave, shave off; to have one's hair cut, have a haircut' xala ^c خلخ 'to take off, doff, put off, slip off; to undress, take off one's clothes, disrobe, strip; to striptease' şabaġ (الشعر) 'to dye, tint, tinge, colour, paint, tincture, imbue; to pigment' qala ^c خلخ 'to take off one's clothes, undress, strip' kaḥal خط 'to take off one's clothes, undress, strip' kaḥal خط 'to darken the (edges of the) eyelids with kohl; to paint or smear with kohl to feast one's eyes on' kasā ند 'to clothe, dress, garb, attire, robe; to drape, face; to cover; to overlay, coat, plate' labis منط 'to comb (the hair), do or do up (the hair), dress or style (the hair), coif or coiffure (the hair)' naza^c 'take off, to pull out, extract, pluck out, tear out; to remove, take away' 		
	deplume (feathers); to depilate, (hair)' 11. nad (نخب (الثوب عنه) to take off one's clothes undress'		
	• < - /	<i>`</i>	
The syntactic frame of transitives	Verb + Subject + Object		
The nature of	The subject is a human being who is in	nterested in appearance and dressing.	
the subject			
The nature of	The object can be a body part or clothe	es.	
the object			
Examples	حلَقِ محمدٌ شعره 1.	لبَس محمدٌ ثوبه 7.	
	خلع محمد ثوبه 2.	مشط محمد شعره .8	
	صبغ محمد شعره 3.	نزع محمد شعر وجهه . 9	
	قلع محمد توبه .4	نتف محمد تسعره. 10	
	کخَل محمد عينه .5	نضًا محمد توبه 11.	
	کسا محمدَ جسده .6		

4.2.43 Verbs of bodily care (class 43)

4.2.44 Verbs of winning (class 44)

Class	These verbs describe events, occasions, or activities in which the subject		
Description	wins or gains by success in competition.		
1			
Levin's class			
Vendler's class	State verbs		
Transitive or	Transitive		
intransitive	Intransitive		
Class members	1. <i>bazz</i> بز 'to surpass, excel, overtop, defeat, overcome'		
	2. baluġ بلغ 'to culminate'		
	3. <i>janā</i> جنّى 'to earn, gain, win, profit'		
	4. $h\bar{a}z$ 'to hold, possess, own, win, gain, earn'		
	5. <i>hasal</i> حصل 'to obtain, get, acquire, win, gain, earn'		
	6. rabah ربح 'to win, gain, earn'		
	7. rajah رجى 'to outweigh, outbalance, overbalance, overweigh, exceed		
	in influence or power to incline'		
	8. <i>saḥaq</i> سحق 'overwhelm, smash'		
	9. <i>zafar</i> ظفر 'to win, gain'		
	10. <i>ġalab</i> غلب 'to defeat, overcome, prevail'		
	11. <i>ġanam</i> غنم 'to gain, earn'		
	12. <i>fāz فاز</i> 'to win, gain, achieve'		
	ito excel, overtop' فاق to excel, overtop'		
	to excel, surpass, outshine' فضل to excel, surpass		
	to overcome, defeat' ^{قهر} to overcome, defeat		
	16. <i>nāf</i> ناف 'to outmatch'		
	to win, gain, earn, acquire' نال to win, gain, earn, acquire'		
	18. <i>najaḥ</i> نجح 'to succeed, pass, achieve'		
	19. <i>najaz</i> نجز 'to achieve'		
	20. <i>hazam</i> هزم 'to defeat, overcome'.		
The exertentia	Varb + Subject + Object		
frame of	verb + Subject + Object		
transitivas			
The syntactic	Varb + Subject + Propositional phrase		
frame of	verb + Subject + I repositional pillase		
intransitives			
The nature of	The subject is an animate entity usually human, that has the ability to win		
the subject	gain earn or overcome in competition		
The nature of	The object is an animate entity that has the ability to lose		
the object	The object is an annuale entity that has the ability to lose.		
The nature of	The prepositional phrase is headed by the preposition c_{ala} is 'on' followed		
the	by a noun that refers to a losing opponent		
Prenositional	by a noun that refers to a fosting opponent.		
nhrasa			
pinase			

Examples	Transitives	 بز محمد الخصم
		2. سحق محمد الخصم
		3. غلب محمد الخصم
		4. قهر محمد الخصم
		5. هزم محمد الخصم
	Intransitives	 6. فاق محمد على الخصم
		7 فضل محمد على الخصم
		 8. فاز محمد على الخصم
		 و. حصل محمد على الجائزة
		10. ظفر محمد على المنافسة
		11. ناف محمد على الخصم

Chapter Five

Morphological, Prosodic, and Semantic Analysis of the Nominal Derivatives

5.1 Introduction

This chapter analyzes the morphological and prosodic structure of the nominal derivatives. The word formation of each type of the nominal derivatives is defined, described, and analyzed by the stem-based approach and prosodic analysis. Within the stem-based approach, a nominal derivative is the output stem that is derived from the input stem, a Form I verb (see section 2.6.2). The prosodic analysis describes the word formation processes and prosodic structures of the nominal derivatives. The chapter then moves on to describe the nominal derivatives semantically. The semantic features of each type of the nominal derivative in terms of its object and situation entities are discussed in order to determine the semantic restrictions of deriving valid nominal derivatives from a given class of verbs. Finally, the chapter discusses the semantic features, in terms of its object and situation entities, of the verb classes, accompanied by the compatibility of deriving the six types of nominal derivatives.

5.2 Morphological and prosodic analysis of the nominal derivatives

The following six sub-sections will respectively provide morphological and prosodic analysis of the active participle, the passive participle, the form of exaggeration, the instrumental noun, the qualificative adjective, and finally the locative noun. In each section, the templatic patterns of the nominal derivatives undergo stem-based (see section 2.6.2) and prosodic (see section 2.5) analysis, exemplified by diagrams (prosodic trees) showing the stages of derivation of the nominal derivatives from their Form I verbs.

5.2.1 Morphological and prosodic analysis of the active participle

The active participle is a noun derived from a verb, denoting the doer of an activity, a process, or a state, so it involves transitory, temporary, or incidental actions or states (status). It conveys the basic meaning of the verb from which it is derived as well as describing the entity that is involved in an action, process, or state. The active participle is considered as "a deverbal adjective that may function as a noun" (Ryding 2005: 689). It

can be derived from a triliteral verb (Form I) and other verb forms, both transitive and intransitive. The current study is dedicated to examining nominal derivatives from Form I (triliteral) verbs. The active participle of Form I verbs has one morphological pattern $C_1\bar{a}C_2iC_3$ (*CVVCVC*). Table 11 shows examples of Form I verbs and their active participles.

Verb	Active Participle
'to read' قرأ [°] qara	'reader' قارئ 'qāri
'to write' کتب katab	writer' کاتب kātib
haras حرس 'guard'	guard' حارس hāris
<i>zār</i> زار 'to visit'	visitor' زائر zā°ir
baḥa <u>t</u> 'to research'	'researcher' باحث <i>bāḥi<u>t</u></i>

Table 11: Form I verbs and their participles

"[T]he active participle can function syntactically as a noun, verb or attributive adjective" (Holes 2004: 149). As a noun, the English equivalent of an active participle usually ends in *-er* or *-or*, as in: $q\bar{a}^{2id}$ 'leader' and *mumtahin* ممتحن 'moderator'. As a verb, it corresponds in English to the present participle that ends in *-ing*. The active participle often functions as an adjective, as in *al-jaww bārid* 'the weather is cold'. Some templatic patterns of the active participle involve modifications when a verb is geminate, hollow⁸, or defective (Ryding 2005). Table 12 shows the different verb types with their active participles:

Table 12: The different verb types with their active participles

Verb type	Verb	Active Participle
Strong verb صحيح șaḥīḥ	'to kill' قتل qatal	'killer' قاتل <i>qātil</i>
Geminate verb مضعّف <i>muḍa ^{cc}af</i>	habb حبّ 'to love'	'lover' حابٌّ <i>ḥābb</i>
mahmūz مهموز Hamzated verb	'to read' قرأ <i>qara</i>	'reader' قارئ <i>qāri</i> '

^{8.} This verb has the semi-consonats $w\bar{a}w \, \mathfrak{s}$ or $y\bar{a}^{\circ} \mathfrak{s}$ as its medial radical. According to Ryding (2005: 461), "These two semi-consonants undergo various mutations, turning into °alif, a short vowel, or a long vowel depending on the word structure and derivation".

Assimilated verb مثال <i>mi<u>t</u>āl</i>	to promise وعد wa ^c ad وعد	'promiser' واعد wā ^c id
ajwaf ^{، أ} جوف Hollow verb	$b\bar{a}^{c}$ باع 'to sell'	$bar{a}^{\circ}i^{c}$ بائع 'seller'
nāqiş ناقص Defective verb	'to forget' نسى <i>nasā</i>	'forgetter' ناسِ <i>nāsin</i>

Within the stem-based approach (see section 2.6.2), the stem of the Form I verb is the input deriving the active participle. A Form I verb is a fully vocalized stem that refers to the active perfective verb stem (the third person masculine singular past tense). The templatic pattern of Form I verbs is $C_1VC_2VC_3$, while the active participle (the output of the derivation) has the templatic pattern $C_1\bar{a}C_2iC_3$ (CVVCVC). For example, $l\bar{a} \ ib \ verb$ 'player' can be analyzed as derived from the verb stem $la \ ib \ verb$ is a moraic trochee foot.

The following diagrams show the prosodic representation (see section 2.5) of the formation of the active participle bahat باحث 'researcher' that is derived from the Form I verb bahat 'research'. The prosodic representation of the verb stem bahat بحث 'to research'. The prosodic representation of the verb stem bahat verb bahat 'researcher' is derived in (1a). From this the active participle $b\bar{a}hit$ 'researcher' is derived through three stages. First, reduplication of the initial mora, as shown in (1b). Second, association of the vowel /a/ with the reduplicated mora, resulting in a long vowel, as shown in (1c). Finally, melodic overwriting takes place where /a/ is overwritten by /i/, as shown in (1c).



- Prosodic representation of Form I verb baḥaṯ بحث 'to research'



There is only one templatic pattern CVVCVC for the active participle of Form I verbs. Prosodically, this templatic pattern can be analyzed within these levels: At the syllabic level of analysis, it comprises two heavy syllables *CVV* and *CVC*. At the moraic level, the initial syllable is bimoraic and the final one is monomoraic; for instance, $x\bar{a}sir$ 'loser', comprises one bimoraic syllable and one monomoraic syllable *CV*<. The final consonant in the prosodic template of the active participle is extrametrical. At the foot level, this word has only one moraic trochaic foot followed by an unparsed syllable (see section 2.5).

5.2.2 Morphological and prosodic analysis of the passive participle

The passive participle is derived from a verb to denote the entity that represents the entity affected by the action of the verb. The passive participle can be formed from any transitive verb, and from an intransitive verb that takes a complementary prepositional phrase (or adverbial phrase), such as *mamrūr bih* مرور به it m. is passed by' and *madhūb ^oilayhā* هذه هذه آله ونا المدرسة مذهوبٌ إليها

single morphological pattern $maC_1C_2\bar{u}C_3$ مفعول (maCVVC). Table 13 shows passive participles derived from Form I verbs:

Form I Verb	Passive Participle
'to eat' أكل 'to eat'	ma²kūl مأكول 'eaten'
'to open' فتح fatah	opened' مفتوح <i>maftū</i> h
'to steal' سرق <i>saraq</i>	stolen' مسروق <i>masrūq</i>
'to break' کسر <i>kasar</i>	broken' مکسور broken'
'to reveal' کشف <i>kašaf</i>	'revealed' مکشوف <i>makšūf</i>

Table 13: Passive participles derived from Form I verbs

Some templatic patterns of the passive participle involve modifications when a verb type is hollow or defective (Ryding 2005). Table 14 shows passive participles derived from different verb types:

Verb type	Verb	Passive Participle
Strong verb صحيح șaḥīḥ	karih کر 'to hate'	<i>makrūh</i> مکروہ 'hateful'
Geminate verb مضعّف <i>muda ^{cc}af</i>	<i>habb حبّ</i> 'to love'	'lovable' محبوب maḥbūb'
mahmūz مهموز Hamzated verb	to read' قرأ [°] qara	'read' مقروء <i>"maqrū</i>
Assimilated verb مثال <i>mi<u>t</u>āl</i>	'to endue' و هب wahab	endowed' مو هوب <i>mawhūb</i>
ajwaf ^{° أ} جوف Hollow verb	bā ^c باع 'to sell'	mabı ^c مبيع 'sold'
<i>nāqiṣ</i> ناقص	'to forget' نسی <i>nasā</i>	'forgotten' منسي <i>mansiy</i>

Table 14: Passive participles derived from different verb types

Under a stem-based analysis (see section 2.6.2), the stem of the passive participle, with the templatic pattern $maC_1C_2\bar{u}C_3$, is derived from the stem of the Form I verb $C_1VC_2VC_3$ (third person masculine singular past tense). For example, the passive participle $majr\bar{u}h$ $\rightarrow \infty$ 'injured' is derived from the verb $jarah \neq \infty$ 'to injure'.

 is disassociated from the left-most mora, and the final-mora is reduplicated. Finally in (2c), mora association and melodic overwriting of /a/ by /u/ take place to form the passive participle (see section 2.5).



^{9.} From now on I assume, where the derived template takes the form $CV-C_1C_2V(V)C_3(-VC)$, association of the left-most stem consonant with the left-most mora and disassociation of the initial stem vowel from the left-most mora.

The passive participle of Form I verbs, such as matlub 'required', has only the template $maC_1C_2\overline{u}C_3$ (CVCCVVC). Prosodically, this templatic pattern can be analyzed within these levels: At the syllabic level of analysis, this template is bisyllabic, containing the heavy syllable CVC and the superheavy syllable CVVC. At the moraic level, both syllables are bimoraic. The final consonant in the syllable CVV

5.2.3 Morphological and prosodic analysis of the form of exaggeration

The form of exaggeration is related to the active participle and used to indicate an exaggeration and abundance in the meaning of an event, action or state. The form of the active participle involves potentially both scarcity and abundance in its meaning. If there is a need to indicate abundance in terms of quality and quantity, the form of exaggeration can be better used. The form of exaggeration has three standard templatic patterns: $C_{1a}C_{2}C_{2}\overline{a}C_{3}$ is (CVCCVVC), $C_{1a}C_{2}\overline{u}C_{3}$ is (CVCVVC), and $miC_{1}C_{2}\overline{a}C_{3}$ is (CVCCVVC). The three patterns can be theoretically applied to any triliteral verb. Table 15 exemplifies patterns of the form of exaggeration:

Pattern	Verb	Form of exaggeration
$C_1 a C_2 C_2 \bar{a} C_3$	'to steal' سرق <i>saraq</i>	'robber, burglar' سرّاق <i>sarrāq</i>
$C_1 a C_2 \bar{u} C_3$	'to eat' أكل 'to eat	[°] <i>akūl</i> أكُول 'gluttonous'
$miC_1C_2\bar{a}C_3$	'to come' فَنِّرِم to come'	'audacious' مِقْدام miqdām

Table 15: Patterns of the form of exaggeration

 $C_1 a C_2 C_2 \bar{a} C_3$ is the most common pattern and can be applied to any transitive and intransitive triliteral verb, while $C_1 a C_2 \bar{u} C_3$ and $mi C_1 C_2 \bar{a} C_3$ are derived from transitive verbs only (Hasan 1969: 3/260). The verb type affects the formation of the form of exaggeration. If the verb type is hollow or defective, templatic patterns of the form of exaggeration involve some modifications. Table 16 shows the different verb types with their forms of exaggeration:

Table 16: The different verb types with their forms of exaggeration

Verb typeVerbForm of exaggeration	
-----------------------------------	--

Strong root صحيح șaḥīḥ	'to lie' کذب <i>kadِab</i>	'mendacious' كذَّاب kadٍdāb
Geminate root مضعّف <i>muḍa ^{cc}af</i>	'to like' وڏّ wadd	wadūd ودود 'affable'
Hamzated root مهموز <i>mahmūz</i>	'to mercy' رأف <i>raºaf</i>	ra ² ūf رؤوف 'gracious'
Assimilated root مثال <i>mi<u>t</u>āl</i>	'to endue' و هب wahab	'donatory' وهّاب wahhāb
ajwaf° أجوف Hollow root	'to sleep' نام <i>nām</i>	<i>nawwām</i> نوّام 'sleeper; slugabed'
nāqiş ناقص Defective root	'to call' دعی <i>da °ā</i>	da ^{c c} ā ^o دعّاء 'propagandist'

According to the stem-based approach (see section 2.6.2), the Form I verb stem $C_1VC_2VC_3$ is the input for the form of exaggeration that has three templatic patterns $C_1aC_2C_2\bar{a}C_3$, $C_1aC_2\bar{u}C_3$ and $miC_1C_2\bar{a}C_3$. The stem of the form of exaggeration with the pattern $C_1aC_2C_2\bar{a}C_3$, such as $hall\bar{a}q$ خلق 'barber', is derived from the Form I verb stem halaq'to shave'. In (3a), the prosodic representation of (the input of the derivation process) the verb halaq 'to shave' is presented. The output stem is formed by initial-mora reduplication and final-mora reduplication as shown in (3b). Finally, in (3c) the vowels are associated with the reduplicated moras (see section 2.5).





 Association of medial consonant with initial reduplicated mora; association of right-most vowel with final reduplicated mora

Prosodically, this templatic pattern can be analyzed within these levels: at the syllabic level of analysis, the template $C_1 a C_2 C_2 \bar{a} C_3$ is bisyllabic, with the heavy syllable *CVC* and the superheavy syllable *CVV*<*C*>; the final consonant is extrasyllabic. At the moraic level, both syllables in the template $C_1 a C_2 C_2 \bar{a} C_3$ are bimoraic. At the foot level of analysis, this template has two feet (see section 2.5).

The form of exaggeration that has the template $C_{1a}C_{2}\bar{u}C_{3}$, such as $\check{s}a\dot{g}a\bar{u}f$ 'humane, sympathetic' is derived from the Form I verb $\check{s}a\dot{g}af$ 'to be very fond of, to adore'. In (4a), the prosodic representation of the verb $\check{s}a\dot{g}af$ is presented. Here, the form of exaggeration is formed by final-mora reduplication, as in (4b); and association of the rightmost vowel /a/ with the reduplicated mora and the melodic overwriting of the vowel /a/ by /u/ are shown in (4c).



- Prosodic representation of Form I verb *šaġaf* شغف 'to be very fond of, to adore'



- Final-mora reduplication

- Association of the right-most vowel /a/ with the reduplicated mora; melodic overwriting of the vowel /a/ by /u/

Prosodically, this templatic pattern can be analyzed within these levels: at the syllabic level of analysis, the template $C_1 a C_2 \bar{u} C_3$ is bisyllabic, containing the light syllable *CV* and the superheavy syllable *CVVC*; the final consonant in this template is extrasyllabic. At the moraic level, the first syllable in the template $C_1 a C_2 \bar{u} C_3$ is monomoraic, while the other is bimoraic. At the foot level, there is one iambic foot.

Finally is the form of exaggeration with the template $miC_1C_2\bar{a}C_3$, such as $mih\underline{d}\bar{a}r$ 'babbler; loquacious' is derived from the form I verb $ha\underline{d}ar$ 'to babble'. The prosodic representation of the verb stem $ha\underline{d}ar$ هذر is presented in (5a). The stem of the form of exaggeration ($miC_1C_2\bar{a}C_3$) is derived, as shown in (5b), by prefixing the syllable $m\mu$ - to the template, associating the consonant /h/ with the left-most mora, disassociating the vowel /a/ from the left-most mora, and reduplicating the final-mora. Finally, in (5c), association of the left-most vowel /i/ with the prefixed mora and the right-most vowel /a/ with the reduplicated mora take place (see section 2.5).



- The prosodic representation of Form I verb *hadar هذر* 'to babble'.

 Association of the consonant /h/ with the left-most mora; disassociation of the vowel /a/ from the left-most mora; mμprefixation; final-mora reduplication

- Association of the vowel /i/ with the prefixed mora; association of the right-most vowel /a/ with the reduplicated mora

Prosodically, this templatic pattern can be analyzed within these levels: at the syllabic level of analysis, the template $miC_1C_2\bar{a}C_3$ is also bisyllabic, containing the heavy syllable miC and the superheavy syllable CVVC. At the moraic level, the template $miC_1C_2\bar{a}C_3$ has two bimoraic syllables. The final consonant in the three templates in the syllable CVV < C > is considered extrasyllabic (see section 2.5). At the foot level, this template has two feet and exhibits iambic feet.

5.2.4 Morphological and prosodic analysis of the instrumental noun

The instrumental noun is a noun derived from Form I triliteral verbs. It denotes a tool, device, or machine by which the action of a verb is performed. For example, the instrumental noun *miftāh* device, is a tool by which something is opened. Therefore, the verb from which the instrumental noun is derived should express a kind of making, forming, working, producing, functioning, handling, or crafting. The instrumental noun in Arabic has four canonical templatic patterns. Table 17 exemplifies Form I verbs and their instrumental noun patterns:

Form I Verb	Pattern	Instrumental Noun
barad برد 'to file'	مِفْعَل miC ₁ C ₂ aC ₃	<i>mibrad</i> مبرد 'rasp'
fataḥ فتح 'to open'	$miC_1C_2ar{a}C_3$ مفعال	'key' مفتاح miftāh
'to sweep' کنس kanas	$miC_1C_2aC_3ah$ مفعلة	broom' مکنسة <i>miknasah</i>
<i>ġasal</i> غسل 'to wash'	فعّالة C ₁ aC ₂ C ₂ āC ₃ ah	'washing machine' غسالة 'washing machine'

Table 17: Form I verbs and their instrumental noun patterns

The patterns $miC_1C_2aC_3$ مِغْتَل (CVCCVC), $miC_1C_2\bar{a}C_3$ مفعال (CVCCVVC), and $miC_1C_2aC_3ah$ مفعال (CVCCVVC) are the most frequent in Arabic of all eras, whereas the $C_1aC_2C_2\bar{a}C_3ah$ فعَالة (CVCCVVCVC) pattern emerged in the modern era.¹⁰ In Arabic, it is acceptable grammatically to derive the instrumental noun from the four mentioned patterns; however, not all valid potential forms are in use. For example, the valid instrumental nouns of the verb našar منشار to saw wood' can be minšar منشار nisšarah منشار naššārah منشار 'saw'. The potential to derive new semantically and grammatically valid instrumental nouns has allowed the Arabic language to keep up with developments in human civilization through the coining of new terms; hence potential words can be established for future use in order to enhance the productivity of the language. Some templatic patterns of the instrumental noun involve modifications only when a verb type is hollow (Ryding 2005). Table 18 shows the different verb types provided with their instrumental nouns:

^{10.} This pattern was approved by Academy of the Arabic Language in Cairo to be used for new words in Arabic.

Verb type	verb	Instrumental Noun
Strong root صحيح șaḥīḥ	to break' کسر <i>kasar</i>	'crusher' کسارۃ <i>kassārah</i>
Geminate root مضعّف <i>muḍa ^{c c}af</i>	<i>jarr</i> 'جر 'to drag'	<i>mijrār</i> مجرار 'bulldozer; tractor'
Hamzated root مهموز <i>mahmūz</i>	'to read' قرأ ⁽ qara	reader' مقراء <i>maqrā</i>
Assimilated root مثال <i>mi<u>t</u>āl</i>	'to tattoo' وشم wašam	'tattoo tool' موشم <i>mawšim</i>
ajwaf ^{° أ} جوف Hollow root	$b\bar{a}^{c}$ باع 'to sell'	bayyā ^c ah بيّاعة 'selling machine'
nāqiş ناقص Defective root	to fry' قلی qalā	'frying pan' مقلی <i>miqlā</i>

Table 18: The different verb types with their instrumental nouns

As a condition, the verb from which the instrumental noun is derived is generally a transitive verb. As the instrument is a tool by which the subject handles the object to convey the effect of the verb's action to the object. However, the verb may become intransitive when it denotes handling something practical, such as misfah 'filter' that is derived from the verb safa and 'to filter'.

The instrumental noun within the stem-based analysis (see section 2.6.2) is an output derived from the input stem Form I verb $C_1VC_2VC_3$. The stem of the instrumental noun has specific templatic patterns: $maC_1C_2aC_3$ مِفْعَل (CVCCVC), $miC_1C_2\bar{a}C_3$ مِفْعَل (CVCCVVC), and $C_1aC_2C_2\bar{a}C_3ah$ مِفْعَلَة (CVCCVCVC).

The templatic pattern $maC_1C_2aC_3$ مِفْعَل, such as mit_qab مثقب 'hole punch', is derived from the verb stem $C_1VC_2VC_3$ taqab ثقب 'to make a hole'. The prosodic representation of the verb stem taqab ثقب is presented in (6a). The stem mit_qab مثقب 'hole punch' is formed by prefixing $m\mu$ - to the template, associating the consonant /t/ with the left-most mora, and disassociating the vowel /a/ from left-most mora, as shown in (6b); and association of the vowel /i/ with the prefixed mora, as shown in (6c).



Prosodically, this templatic pattern can be analyzed within these levels: at the syllabic level of analysis, the template $miC_1C_2aC_3$ comprises two heavy syllables CVC / CV < C>. At the moraic level, the initial syllable is bimoraic while the final syllable is monomoraic. The final consonant in this template is extrametrical. At the foot level, there is one foot and one monomoraic syllable (see section 2.5).

The stem of the instrumental noun that has the prosodic template $miC_1C_2\bar{a}C_3$, such as $mihr\bar{a}t$ محرث 'plough', is derived from the verb stem $C_1VC_2VC_3$ harata حدث 'to plough'.

The prosodic representation of the verb stem *harat* حرث is presented in (7a). The stem *mihrāt* مِحْرَات 'plough' is formed by prefixing *mµ*- to the template, associating the consonant /h/ with the left-most mora, disassociating the vowel /a/ from the left-most mora, and reduplicating the vowel /a/ within the final-mora, as shown in (7b). Finally, there is a process of associating the vowel /i/ with the prefixed mora and the vowel /a/ with the reduplicated mora, as in (7c).



- The prosodic representation of Form I verb <u>harat</u> حرث 'to plough'

 Association of the consonant /h/ with the left-most mora; disassociation of the vowel /a/ from the left-most mora; mμprefixation; final-mora reduplication

- Association of the vowel /i/ with the prefixed mora, and the vowel /a/ with the reduplicated mora

Prosodically, this templatic pattern can be analyzed within these levels: at the syllabic level of analysis, this prosodic template is bisyllabic, with a heavy syllable miC and a superheavy syllable CVVC. At the moraic level, both syllables are bimoraic. The final consonant in the template is considered extrasyllabic. At the foot level, this template has two feet and exhibits iambic feet.

The stem of the instrumental noun that has the prosodic template $miC_1C_2aC_3ah$ such as *mijrafah* and 'shovel' is derived from the verb stem $C_1aC_2aC_3$ jaraf is shown in (8a). sweep away'. The prosodic representation of the verb stem jaraf and is shown in (8a). The stem *mijrafah* and 'shovel' is formed by prefixing $m\mu$ - to the template, associating the consonant /j/ with the left-most mora; disassociating the vowel /a/ from the left-most mora; and $m\mu$ -prefixation and μ -suffixation of *-at* as shown in (8b). In (8c), prefixed and suffixed moras are associated.





 Association of the vowel /i/ with prefixed mora; association of right-most vowel /a/ with suffixed mora

Prosodically, this templatic pattern can be analyzed within these levels: at the syllabic level, the prosodic template $miC_1C_2aC_3ah$ (*CVCCVCVC*) consists of three syllables, the heavy syllable *CVC*, light syllable *CV*, and finally heavy syllable *CV*<*C*>. At the moraic level of analysis, the initial syllable is bimoraic while the two other syllables are monomoraic. The final consonant in the template is extrametrical. At the foot level, this template has two feet and exhibits a trochaic foot.

Finally, the instrumental noun has the templatic pattern $C_1 a C_2 C_2 \bar{a} C_3 a h$ such as *xallāṭah* 'mixer' derived from the verb stem $C_1 V C_2 V C_3$ *xalaṭ* 'to mix'. This stem is formed by reduplicating the initial-mora, reduplicating the final-mora, and μ -suffixing *ah* to the template, as shown in (9b). Finally (9c) shows associating the medial consonant /l/ with the left-most reduplicated mora, associating the right-most vowel /a/ with the rightmost reduplicated mora, and associating-*ah* with the suffixed mora.



- The prosodic representation of Form I verb *xalat* خلط 'to mix'



- Initial-mora reduplication; final-mora reduplication; μ -suffixation

- Association of medial consonant /l/ with left-most reduplicated mora; association of right-most vowel /a/ with right-most reduplicated mora; association of -*ah* with suffixed mora

Prosodically, this templatic pattern can be analyzed within these levels: at the syllabic level of analysis, the template $C_{1a}C_{2}C_{2}\bar{a}C_{3}ah$ comprises three heavy syllables CVC/CVV/CVC. At the moraic level, the first two syllables are bimoraic, while the final syllable is monomoraic. The final consonant in the syllable CV < C> is considered extrametrical. At the foot level of analysis, this template has two feet and one monomoraic syllable and exhibits iambic feet (see section 2.5).

5.2.5 Morphological and prosodic analysis of the qualificative adjective

The qualificative adjective is an adjective that functions as a noun. It can only be derived from an intransitive verb, otherwise it is not valid grammatically. Three aspects must exist in the qualificative adjective: a pointer to the description of a state or quality, such as colour; a pointer to an entity (person or thing) that is described or characterized by a state or quality, for example, *jamīl* جميل 'beautiful' which indicates an entity (person or thing) that is characterised by beauty; third, its description is inherent and permanent throughout all times (past, present, and future), such as °*abyad* 'white' (Hasan 1969: 3/283).

There are similarities and differences between the active participle and the qualificative adjective. The similarity is found where both refer to an entity and an action, process, or state. The difference between them is that the active participle refers to an accidental non-permanent action, process or state, while the qualificative adjective refers to a stable permanent state or quality. For example, from the same verb *karum* λc 'to be generous' we can derive the active participle *kārim* λc and the qualificative adjective *karīm* $\lambda c c$. The former refers to someone who is temporarily (occasionally) generous, whereas the latter refers to someone who is by nature generous in the past, present, and future.

The qualificative adjective has three main templatic patterns $C_1VC_2VC_3$ فعل (CVCVC), ${}^{a}C_1C_2VC_3$ فعيل (CVCCVC), and $C_1aC_2\overline{i}C_3$ فعيل (CVCVVC):

- *CVCVC* ($C_1aC_2iC_3$) governs the qualificative adjectives that are derived from Form I verbs with the templatic pattern *CVCVC* ($C_1aC_2iC_3$). Here, there is no templatic change as both of verb and its derived qualificative adjective have the same pattern; for example, the qualificative adjectives *fariḥ* فر 'happy', *tacib* 'itred', and *dajir* ضجر 'bored' are respectively derived from the verbs *fariḥ* 'to be happy' *tacib* 'ize 'to be tired', and *dajir* 'to be bored'.
- [°]*aCCVC* ([°]*aC*₁*C*₂*aC*₃) governs the qualificative adjectives that denote colours, flaws and bodily qualities, such as [°]*ahmar* أحمر 'red', [°]*a*^c*war* أعور 'one-eyed', and [°]*ahwar* أحور 'one who has beautiful eyes', derived respectively from the verbs *hamir* حور 'to be red', ^c*awir* عور 'one-eyed', and *hawir* خور 'having beautiful eyes'.
- *CVCVVC* ($C_1 a C_2 \overline{i} C_3$) governs the qualificative adjectives that describe a permanent quality relatively, such as the qualificative adjective *hazīn* حزين 'sad' that is derived from *hazin* حزن 'to be sad'.

The formation of the qualificative adjective involves modifications when a verb is hollow. Table 19 shows the different verb types presented with their qualificative adjectives:

Verb type	Verb	Qualificative Adjective
Strong verb صحيح șaḥīḥ	sa ^c ud سعيد 'to be happy'	sacīd سعيد 'happy'
Geminate verb مضعّف <i>muḍa ^{cc}af</i>	'to be bitter' مرَّ <i>marr</i>	bitter' مرير marīr

 Table 19: The different verb types with their qualificative adjectives

Hamzated verb مهموز <i>mahmūz</i>	$s\bar{a}^{\circ}$ ساء 'to offend'	bad' سيّء bad'
Assimilated verb مثال <i>mi<u>t</u>āl</i>	wahim و هم 'to be illusory'	'deceived' وهِمِّ wahimun
ajwaf ^{، أ} جوف Hollow verb	'to die' مات <i>sād</i>	'dead ميّت sayyid
Defective verb ناقص nāqiş	'to be blind' عمِيَ <i>jarā</i>	'blind' أعمى [°] a ^c mā

The stem of the qualificative adjective, within the stem-based approach (see section 2.6.2), is derived from the stem of Form I verbs. The stem of the qualificative adjective that has the templatic pattern $C_1VC_2VC_3$ does not involve any templatic change as both the verb and its qualificative adjective share the same templatic pattern, such as *CVCVC marih* α_{10} , 'cheerful' derived from the verb stem *CVCVC marih* α_{10} , 'to be cheerful', as shown in (10a, 10b, and 10c). Only the context determines whether it is a qualificative adjective or a Form I verb.





Prosodically, the template $C_1VC_2VC_3$ can be analyzed within these levels: at the syllabic level of analysis, it comprises two light syllables CV < C >. The final consonant in the word is extrametrical. At the moraic level, both syllables CV are monomoraic. At the foot level of analysis, there is one moraic trochaic foot in this template.

The stem of the qualificative adjective which has the templatic pattern ²*aCCVC* is derived from the Form I verb $C_1VC_2VC_3$. For example, the qualificative adjective ²*axras* أخرس 'dumb' is derived from the verb *xaris* نجرِس 'to be dumb'. The prosodic presentation of the verb *xaris* نجرِس is presented in (11a). The stem ²*axras* أخرِس is formed by prefixing ² μ - to the template, associating the consonant /x/ with the left-most mora, and disassociating the vowel /a/ from the left-most mora, as shown in (11b). Finally in (11c), there is association of the vowel /a/ with the prefixed mora and melodic overwriting of the vowel /i/ by /a/.



- Prosodic representation of Form I verb *xaris* خرس 'to be dumb'



 Association of the consonant /x/ with the left-most mora; disassociation of the vowel /a/ from the left-most mora; ^aμprefixation

 Association of the left-most vowel /a/ with the prefixed mora; melodic overwriting of the vowel /i/ by /a/

Prosodically, this templatic pattern can be analyzed within these levels: at the syllabic level, the template ${}^{2}aC_{1}C_{2}VC_{3}$ is bisyllabic, containing two heavy syllables *CVC*. At the moraic level, the first syllable is bimoraic, while the other is monomoraic. The final consonant in the syllable *CV*<*C*> is considered extrametrical. At the foot level of analysis, this template has one moraic trochaic foot and one monomoraic syllable (see section 2.5).

Finally, the qualificative adjective that has the templatic pattern $C_{1a}C_{2}\bar{u}C_{3}$ (CVCVVC), such as *jamīl* جميل 'beautiful', is derived from the verb stem *jamul* خميل 'to be beautiful'. The prosodic representation of the verb stem *jamul* جميل is presented in (12a). The stem *jamīl* جميل is formed by reduplicating the final-mora, as shown in (12b). Finally in (12c), association of the right-most vowel with the reduplicated mora, and melodic overwriting of /u/ by /i/ take place.



- Prosodic representation of Form I verb *jamul* جمُل 'to be beautiful'

- Final-mora reduplication

- Association of right-most vowel /u/ to reduplicated mora; melodic overwriting of the vowel /u/ by /i/

Prosodically, this templatic pattern can be analyzed within these levels: at the syllabic level, the template $C_{1a}C_{2}\bar{u}C_{3}$ (*CVCVVC*) is bisyllabic, containing the light syllable *CV* and the superheavy syllable *CVVC*. The final consonant in this template is extrasyllabic. At the moraic level of analysis, the first syllable in the template *CV/CVVC* is monomoraic, while the other is bimoraic. At the foot level, there is one iambic foot (see section 2.5).

5.2.6 Morphological and prosodic analysis of the locative noun

The locative noun is derived from a Form I verb, and refers to the location in which the action of the verb takes place. The locative noun combines the core meaning of the verb and the place in which the action of the verb occurs. The locative noun can be derived from both transitive and intransitive verbs, such as $la^{c}ib$ 'to play' and *jalas* 'to sit', for example the locative nouns *mal* ^c *ab* (Hasan 1969: 3/318).

The locative noun of Form I verbs $C_1VC_2VC_3$ has the single morphological pattern, $maC_1C_2aC_3$ (CVCCVC). Regularly, the vowel that follows the second consonant in the morphological pattern maCCVC is /a/ vowel. However, in two cases the morphological pattern of the locative noun is maCCVC ($maC_1C_2iC_3$ with /a/ vowel after the second consonant). In the first case, the Form I verb $C_1VC_2VC_3$ is an assimilated verb (starting with the consonant w_3), such as $wada^\circ e^\circ$ to put' for which the locative noun is $mawdi^\circ$ vowel after the second consonant in its imperfective form, such as $raji^\circ e_{-2} - yarji^\circ e_{-2}$ to get back', the locative noun is $mac_1C_2iC_3$, as in majlis vowel after the second consonant in the second consonant in the first case. The pattern of the second consonant is mawdi or e_{-2} ($e_{-2}iC_3$, including the /i/ vowel after the second consonant in its imperfective form, such as $raji^\circ e_{-2} - yarji^\circ e_{-2} + e_{-3}$ to get back', the locative noun is $mac_1C_2iC_3$, as in majlis vowel after input verbs:

Verb	Pattern	Locative Noun
'to exit' خرج xaraj	$maC_1C_2aC_3$	exit' مخرج <i>maxraj</i>
'to play' لعب la ^c ab' لعب	$maC_1C_2aC_3$	playfield' ملعب mal ^c ab' ملعب
'to sit' جلس jalas	$maC_1C_2iC_3$	'council, seat' مجلس <i>majlis</i>

Table 20: Examples of locative nouns and their input verbs

Unlike other types of nominal derivatives, the locative noun formation does not involve modifications when a verb is geminate, hollow, or defective. Table 21 shows the different verb types presented with their locative nouns:

Verb type	Verb	Locative Noun
saḥīḥ صحيح Strong verb	'to go out' خرج <i>xaraj</i>	'exit' مخرج <i>maxraj</i>
<i>muḍa ^{cc}af</i> مضعّف	'to pass' مرَّ <i>marr</i>	'path' ممرَّ path'
Hamzated verb مهموز <i>mahmūz</i>	'to capture' أسر [»] asar	'prison' مأسر ma ^o sar
Assimilated verb مثال <i>mi<u>t</u>āl</i>	'to stop' وقف waqif	'stop' موقف mawqif
ajwaf° أجوف Hollow verb	'to sell' سار sār	'track' مسار <i>masār</i>
nāqiş ناقص Defective verb	<i>jarā</i> جرى 'to flow'	'stream' مجری <i>majrā</i>

Table 21: The different verb types with their locative nouns

The stem of the locative noun has the single templatic pattern $maC_1C_2VC_3$, which can be analyzed with the stem-based approach (see section 2.6.2); for example, when derived from the stem of a Form I verb *CVCVC*, such as the locative noun *mahbat* have 'airstrip' derived from *habat* have 'to land'. The prosodic representation of the verb stem *habat* is presented in (13a). The stem *mahbat* have is formed by prefixing $m\mu$ - to the template, associating the consonant /h/ with the left-most mora, and disassociating the vowel /a/ from it, as shown in (13b). Finally in (13c), there is association of the left-most vowel /a/ to the prefixed mora, and melodic overwriting if the template is $maC_1C_2iC_3$ (with /i/ vowel after the second consonant), with /a/ overwritten by /i/.



- The prosodic representation of Form I habat هبط 'to land'



 Association of the consonant /h/ with leftmost mora; disassociation of the vowel /a/ from left-most mora; *mμ*-prefixation

Association of the vowel /a/ with the prefixed mora; association of right-most vowel /a/ with right-most reduplicated mora and association of with the vowel /a/

At the syllabic level of analysis, this templatic pattern comprises two heavy syllables maC and CVC. At the moraic level, the initial syllable CVC is a bimoraic syllable, and the final syllable CV < C > is a monomoraic syllable. The final consonant in the syllable CV < C > is extrametrical. At the foot level, there is one foot and one unparsed monomoraic syllable (see section 2.5).
5.3 Semantic analysis of the nominal derivatives

The semantic features of the nominal derivatives and the verb classes (see chapter 4) will be described here in order to determine the validity of deriving each type of nominal derivatives from each class of the 44 verb classes. Our model to determine the validity is based on identifying the compatibility between the nominal derivatives and the verb classes in terms of two types of entity (object entities and situation entities) inspired by Helbig (2006: 410) (see section 3.5). Our model describes and classifies both nominal derivatives and verb classes, in terms of the object entity,¹¹ wheather they refer to concrete or abstract entities, and in terms of the situation entity wheather they refer to state or event entities.

To determine the semantic restrictions to deriving valid nominal derivatives from a given class of verbs, compatibility between the following aspects is considered and determined:

- The semantic features (object entities and situation entities) of the nominal derivative.
- The semantic features (object entities and situation entities) of each verb class.

The following figure illustrates the semantic features that will be examined to measure the compatibility between the nominal derivative and the verb classes:



Figure 2: Semantic feature compatability between the nominal derivative and the verb classes

Our own criteria for determining the verb classes and nominal derivatives into concrete or abstract entities and into state or event entities, are detailed below:

^{11.} The word 'object' here does not refer to the grammatical sense that denotes 'noun governed by a transitive verb'.

- *Concrete* entity refers to an action, event or state which is accessible to one or more of the five senses, i.e. the entity can be seen, heard, touched, smelt, or tasted. It expresses tangible meanings that occupy physical space.
- *Abstract* entity refers to a concept, attribute, quality or state which is not accessible to any of the five senses. It expresses non-physical, unseen and intangible meanings.
- *State* entity refers to a state which is permanent or will last for a significant length of time.
- *Event* entity refers to a change from one state to another.

To derive a valid nominal derivative from a given verb, there must be compatibility between the semantic features of the nominal derivative and the semantic features of the verb. For example, a nominal derivative, such as an instrumental noun, has to have semantic features that involve the concrete and event entities; therefore, the verb from which we can derive the instrumental noun must also involve concrete (object) and event (situation) entities, such as the verb *xalat* to mix¹² whose instrumental noun is *xallātah* 'to mixer'. On the contrary, a verb involving abstract and state entities, such as *hazin* خزن nistract (object) and state (situation) entities, a qualificative adjective such as *hazīn* 'sad, doleful' that is derived from the verb *hazin* 'cio be sad' (see section 3.5).

5.3.1 Semantic features of the active participle

In the literature of semantics, the active participle as an agent noun has been discussed according to its thematic roles. Fillmore (1971: 42) states that the agent is a thematic role along with instrument, experiencer, object, source, goal, location, time and path. In this sense, the agent is the initiator of an activity, a process, or a state. Luján (2010) argues that, "Agents are prototypically animates, especially humans, and are characterized by control and intentionality over the action that they perform. Some inanimate entities can also have control over the action, but obviously they cannot have any intentionality".

^{12.} The verb *fataḥ* فتح 'to open' involves a concrete entity in terms of it requires a physical space (touchable), and involves an event entity in terms of it refers to an action and a change.

^{13.} The verb *karih* كره 'to hate' involves an abstract entity in terms of it does not require a physical space, and involves a state entity in terms of it does not refer to an action and a change.

Table 22 shows the semantic features of the active participle, including its object and situation entities:

Active Participle						
Object Entity	Situation Entity					
The semantic features of the active	The semantic features of the active					
participle refer to either concrete or abstract	participle refer to either event or state					
entities, where some active participles refer	situation entities, where some active					
to a concrete entity, and other active	participles refer to an event entity, and other					
participles refer to an abstract entity:	active participles refer to a state entity:					
- Concrete entities (animate or inanimate)	- Event entity involves a change from one					
express tangible meanings (of an action,	state to another.					
event, or state which is accessible to one or	- State entity describes a state which is					
more of the five senses) that occupy	permanent or will last for a significant time.					
physical space.						
- Abstract entities express non-physical,						
unseen and intangible meanings (which are						
not accessible to any of the five senses) that						
do not occupy physical space.						

Table 22: Semantic features of active participle

Due to the fact that semantic features of the active participle refer to either concrete or abstract (object) entities, as well as either event or state (situation) entities, valid active participles can be derived from all of the 44 verb classes. In all cases, the semantic features of any class of verbs (object and situation entities) are compatible with the semantic features of the active participle (object and situation entities). The following figure shows the semantic feature compatibility (see section 3.5) of deriving the active participle:



Figure 3: The semantic feature compatability of deriving the active participle

In the Arabic linguistic literature, there is an argument that involves constraints on deriving an active participle, based on its semantics. Hasan (1969: 3/240) argues that a verb from which the active participle can be derived must indicate an impermanent event. He claims that we can only derive a qualificative adjective¹⁴ from a verb that refers to a permanent state meaning, even if it has the same morphological pattern as the active participle. Thus, Hassan excludes these forms from being active participles, and claims they are qualificative adjectives. Several nominal derivatives of the pattern $C\bar{a}CiC$ that denote permanent state meanings are found and used in modern and classical Arabic, such as $k\bar{a}rih \sim \Sigma_{c}$ 'hater' and $n\bar{a}jih \simeq \omega$ 'successful'.

5.3.2 Semantic features of the passive participle

The semantics of the passive participle is, to some extent, relevant to a number of the semantic roles, including recipient, patient and experiencer. Finegan (2008: 204) define them as follows:

^{14.} The qualificative adjective was discussed in section 5.2.5.

- Patient semantic role: "the entity that undergoes a certain change of state".
- Recipient semantic role: "the entity that receives a physical object".
- Experiencer semantic role: "the entity that receives a sensory input".

Table 23 shows the semantic features of the passive participle, including its object and situation entities:

Passive Participle						
Object Entity	Situation Entity					
The semantic features of the passive participle	The semantic features of the passive					
refer to either concrete or abstract entities,	participle refer to either event or state					
where some passive participles refer to a	entities, where some passive participles					
concrete entity, and other passive participles	refer to an event entity, and other passive					
refer to an abstract entity:	participles refer to a state entity:					
- Concrete entities (animate or inanimate),	- Event entity involves a change from					
express tangible meanings (of an action, event	one state to another.					
or state which is accessible to one or more of	- State entity describes a state which is					
the five senses) that occupy physical space.	permanent or will last for a significant					
- Abstract entities express non-physical,	time.					
unseen and intangible meanings (which are not						
accessible to any of the five senses) that do not						
occupy physical space.						

Table 23: The semantic features of the passive participle

Due to the fact that semantic features of the passive participle refer to either concrete or abstract (object) entities, as well as either event or state (situation) entities, valid passive participles can be derived from all of the 44 verb classes. In all cases, the semantic features of any class of verbs (object and situation entities) are compatible with the semantic features of the passive participle (object and situation entities). The following figure shows the semantic feature compatibility (see section 3.5) of deriving the passive participle:



Figure 4: The semantic feature compatability of deriving the passive participle

5.3.3 Semantic features of the form of exaggeration

+

The form of exaggeration is semantically similar to the active participle. Both represent the entity that refers to the doer of an activity, process or state. However, the form of exaggeration specifically signifies intensivity in the meaning in terms of the quantity and frequency of an activity, process or state. As with the active participle, the semantics of the form of exaggeration are related to the agent thematic role (see section 5.4.2). In Arabic, the form of exaggeration is usually used to indicate professions, such as *hallāq* خلاق 'barber', *xabbāz*' خبّاز 'baker', *harrāt*' حرّات' ploughman', *sawwāq*' driver', and *tayyār* driver'.

Table 24 shows the semantic features of the form of exaggeration, including its object and situation entities:

Form of Exaggeration						
Object Entity	Situation Entity					
The semantic features of the form of	The semantic features of the form of					
exaggeration refer to either concrete or	exaggeration refer to either event or state					
abstract entities, where some forms of	entities, where some forms of exaggeration					
exaggeration refer to a concrete entity, and	refer to an event entity, and other forms of					
other forms of exaggeration refer to an	exaggeration refer to a state entity:					
abstract entity:	- Event entity involves a change from one					
- Concrete entities (animate or inanimate)	state to another.					
express tangible meanings (of an action,	- State entity describes a state which is					
event, or state which is accessible to one or	permanent or will last for a significant time.					
more of the five senses) that occupy						
physical space.						
- Abstract entities express non-physical,						
unseen and intangible meanings (which are						
not accessible to any of the five senses) that						
do not occupy physical space.						

Table 24: The semantic features of the verb of exaggeration

Due to the fact that the semantic features of the form of exaggeration refer to either concrete or abstract (object) entities, as well as either event or state (situation) entities, valid forms of exaggeration can be derived from all of the 44 verb classes. In all cases, the semantic features of any class of verbs (object and situation entities) are compatible with the semantic features of the form of exaggeration (object and situation entities). The following figure shows the semantic feature compatibility (see section 3.5) of deriving the form of exaggeration:



Figure 5: The semantic feature compatability of deriving the form of exaggeration

5.3.4 Semantic features of the Instrumental Noun

Semantically, the notion of the instrument has been discussed under thematic roles. According to Fillmore (1971), the instrument is a thematic role. Fromkin, Rodman et al. (2011: 165) define the instrument thematic role as "the means used to accomplish the action". Another semantic description of the instrument is provided by Pawlak and Bielak (2011: 54), as "a physical object which is manipulated by the agent and serves as the 'intermediary' in the transmission of energy".

Table 25 shows the semantic features of the instrumental noun, including its object and situation entities:

Instrumental Noun					
Object Entity	Situation Entity				
The instrumental noun refers to a concrete	The instrumental noun can be described in				
entity that occupies physical space.	terms of the situation entity as an event that				
Furthermore, it must be an inanimate object	induces change from one state to another. It				

Table 25: The semantic features of the instrumental noun

that does not have life, spirit or agency.	must particularly express a kind of making,
	forming, working, producing, functioning,
	handling or crafting something.

To derive valid instrumental nouns, the semantic features (object and situation entities) of the instrumental noun should be identical with those of the verb classes. Therefore, the object and situation entity of a verb class must be, respectively, concrete and event. Otherwise, a derived instrumental noun will be invalid semantically (see Figure 11). The following figure shows the semantic feature compatibility (see section 3.5) of deriving the instrumental noun:

Figure 6: The semantic feature compatability of the instrumental noun



5.3.5 Semantic features of the qualificative adjective

Similarly, the qualificative adjective functions semantically like the active participle, and can be described under the thematic role of the agent (see section 5.3.1) where the qualificative adjective refers to an entity that involves a state. For example, the qualificative adjective $kab\bar{i}r$ \geq 'big' refers to an entity (person or thing) and the state of being big. This state indicates stability and continuity in the meaning. Furthermore, the

qualificative adjective is derived from intransitive verbs that are apparently related to state, not event, entities. The qualificative adjective is not required to indicate event. Table 26 shows the semantic features of the qualificative adjective, including its object and situation entities:

Qualificative Adjective						
Object Entity	Situation Entity					
The qualificative adjective may refer either to a	The situation entity of the					
concrete entity or an abstract entity:	qualificative adjective refers only					
- Concrete entities (animate or inanimate) express	to the state entity that describes a					
tangible meanings (of an action, event, or state	state which is permanent or will					
which is accessible to one or more of the five senses)	last for a significant time.					
that occupy physical space.						
- Abstract entities express non-physical, unseen and						
intangible meanings (which are not accessible to any						
of the five senses) that do not occupy physical space.						

Valid qualificative adjectives can be derived from a class of verbs whose semantic features (object and situation entities) are compatible with the semantic features of the qualificative adjective features (object and situation entities) (see Figure 11). The following figure shows the semantic feature compatibility (see section 3.5) of deriving the qualificative adjective:



Figure 7: The semantic feature compatability of the qualificative adjective

5.3.6 Semantic features of the locative noun

In the literature of semantics, the locative is considered a type of semantic role (Fillmore 1971: 41). It is the entity that denotes the physical place or location in which the action of the verb occurs. Therefore, it should be a tangible entity that can mainly be measured by the sense of touch. Three of the semantic roles presented by Fillmore (1971) can fall under the locative role:

- Goal that represents the location to which the action is targeted.

- Source that represents the location from which the action initiates.

- Path that represents the location through which the action moves.

Table 27 shows the semantic features of the locative noun, including its object and situation entities:

Table 27:	The semantic	features of th	e locative noun
-----------	--------------	----------------	-----------------

Locative Noun					
Object Entity	Situation Entity				
The locative noun refers only to a concrete	The situation entity of the locative noun				
entity that occupies physical space	may refer either to an event entity or a state				
(accessible to the sense of touch).	entity:				
Moreover, it is entirely an inanimate object	- Event entity involves a change from one				
that does not have life, spirit or agency.	state to another.				
	- State entity describes a state which is				
	permanent or will last for a significant time.				

To derive valid locatives nouns, the semantic features (object and situation entities) of the locative noun should be identical with those of the verb classes. Therefore, the object and situation entity of a verb class must be respectively concrete and state or event; otherwise, a derived locative noun will be invalid semantically (see Figure 11). The following figure shows the semantic feature compatibility (see section 3.5) of deriving the locative noun:

Figure 8: The semantic feature compatability of deriving the locative noun



5.4 Semantic features of the verb classes and their compatibility with the nominal derivatives

The following 44 tables provide the semantic features (object and situation entities) of our Arabic verb classes. In terms of the object entity, each of the 44 verb classes is labelled as an abstract or concrete entitity, while in terms of the situation entity they are labelled as state or event entities. The compatibility of each class of verbs with the six types of nominal derivatives is determined, with the validity or invalidity of deriving a nominal derivative clarified with the justification and examples (see section 3.5).

Class 01: Verbs of the state of the body					
Object entity	Situation entity				
These verbs refer to abstract entities, expressing	These verbs refer to state entities. They				
non-physical, unseen and intangible meanings	describe a state which is permanent or				
(which are not accessible to any of the five	will last for a significant time.				
senses) that do not occupy physical space, such					
as 'to feel pain', 'to be hungry' and 'to be					
thirsty'.					

Comparishing with the active participle

There is compatibility between the semantic features of *Verbs of the state of the body* (that involve abstract and state entities) and those of *the active participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate an active participle from *Verbs of the state of the body*. For example, $j\bar{a}^{\circ}i^{\circ}$ جائع 'hungry', $n\bar{a}^{\circ}is$ ناعس 'sleepy', $d\bar{a}^{\circ}ix$ دائخ 'dozy'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of the state of the body* (that involve abstract and state entities) and those of *the passive participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of the state of the body*. For example, $ma^{o}r\bar{u}q$ مأروق 'wakeful', $mask\bar{u}r$ مسكور 'drunken', $mat^{c}\bar{u}b$ 'arage for the state'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of the state of the body* (that involve abstract and state entities) and those of *the form of exaggeration* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of the state of the body*. For example, $mi^{c}t\bar{a}s$ مِعْطَاش 'very thirsty', $ta^{c}\bar{u}b$ 'so tired', *'allām* ألأم 'feeling pain'.

Compatibility with the instrumental noun

There is no compatibility between the semantic features of *Verbs of the state of the body* and the semantic features of *the instrumental noun*, where the features of the verbs involve abstract and state entities, and those of the instrumental noun involve concrete and event entities. Therefore, it is not valid semantically to generate an instrumental noun from *Verbs of the state of the body*.

Compatibility with the qualificative adjective

There is compatibility between the semantic features of *Verbs of the state of the body* (that involve abstract and state entities) and those of *the qualificative adjective* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a qualificative adjective from *Verbs of the state of the body*. For example, *c ariq* أرق 'insomniac', *xadir خ*ر 'anesthetized, doped', *zami*' نظمئ 'thirsty'.

Compatibility with the locative noun

There is no compatibility between the semantic features of *Verbs of the state of the body* and the semantic features of *the locative noun*, where the features (object entity) of the verbs involve abstract entities, and those of the locative noun involve concrete entities. Therefore, it is not valid semantically to generate a locative noun from *Verbs of the state of the body*.

	Class 02: Verbs of body parts											
The object entity			The situation entity									
			,	5							5	
These	verbs	refer	to	concrete	entities.	These	verbs	refer	to	state	entities.	Thev
					,					~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~		

expressing tangible meanings (of an action,	describe a state which is permanent or will
event, or state which is accessible to one or	last for a significant time.
more of the five senses) that occupy	
physical space. They involve a physical	
action that can be seen, heard, or touched,	
such as 'to blink, wink, bat', 'to cough', and	
'to sneeze'.	

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of body parts* (that involve concrete and state entities) and those of *the active participle* (that may also involve concrete and state entities). Therefore, it is valid semantically to generate an active participle from *Verbs of body parts*. For example, $b\bar{a}kin$ بالكي 'one who cries', $s\bar{a}^cil$ ناطس 'one who coughs', $\bar{a}tis$ one who sneezes'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of body parts* (that involve concrete and state entities) and those of *the passive participle* (that may also involve concrete and state entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of body parts*. For example, *maxlūj* نخلوج 'twitched', *madrūf* مذروف 'shedded tears', *malhūt* 'panted'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of body parts* (that involve concrete and state entities) and those of *the form of exaggeration* (that may also involve concrete and state entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of body parts*. For example, $bakk\bar{a}^{\circ}$, often tearful', $sa^{\circ}\bar{a}l$ (coughing a lot', $cat\bar{a}s$ عطّاس 'sneezing a lot'.

Compatibility with the instrumental noun

There is no compatibility between the semantic features of *Verbs of body parts* and the semantic features of *the instrumental noun*, where the features of the verbs involve concrete and state entities, and those of the instrumental noun involve concrete and event

entities. Therefore, it is not valid semantically to generate an instrumental noun from *Verbs* of body parts.

Compatibility with the qualificative adjective

There is compatibility between the semantic features of *Verbs of body parts* (that involve concrete and state entities) and those of *the qualificative adjective* (that may also involve concrete and state entities). Therefore, it is valid semantically to generate a qualificative adjective from *Verbs of body parts*. For example, $dam\bar{i}^c$ دميع 'tearful', *bakiyy* 'weepy', and sa^cil سعل 'cougher'.

Compatibility with the locative noun

There is compatibility between the semantic features of *Verbs of body parts* (that involve concrete and state entities) and those of *the locative noun* (that may also involve concrete and state entities). Therefore, it is valid semantically to generate a locative noun from *Verbs of body parts*. For example, *madraf* مذرف 'a place from which tears shed', *mar c af* مرعف 'a place from which a nosebleed bleeds', *mašhaq* مشهق 'a place through which the air is breathed in'.

Class 03: Verbs	s of diseases			
The object entity	The situation entity			
These verbs refer to abstract entities,	These verbs refer to state entities. They			
expressing non-physical, unseen and	describe a state which is permanent or will			
intangible meanings (which are not accessible	last for a significant time.			
to any of the five senses) that do not occupy				
physical space, such as 'to be deaf or dumb'.				

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of diseases* (that involve abstract and state entities) and those of *the active participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate an active participle from *Verbs of diseases*. For example, $b\bar{a}kim$ باکم 'one who suffers from dumbness', $c\bar{a}qim$ عاقم verbs of diseases.

'one who suffers from infertility', zākim زاكم 'one who suffers from cold'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of diseases* (that involve abstract and state entities) and those of *the passive participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of diseases*. For example, *mašlūl* مشلول 'sufferer from being paralyzed', *mazkūm* مزكوم 'sufferer from catching a cold', *mabrūs* of sufferer from being a leper'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of diseases* (that involve abstract and state entities) and those of *the form of exaggeration* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of diseases*. For example, *marrād*, مرّاض 'one who is very ill', *jarrāb* مرّاض 'one who has a lot of mange', *mizkām* مرَكَام 'one who suffers from a strong cold'.

Compatibility with the instrumental noun

There is no compatibility between the semantic features of *Verbs of diseases* and those of *the instrumental noun*, where the features of the verbs involve abstract and state entities, and those of the instrumental noun involves concrete and event entities. Therefore, it is not valid semantically to generate an instrumental noun from *Verbs of diseases*.

Compatibility with the qualificative adjective

There is compatibility between the semantic features of *Verbs of diseases* (that involve abstract and state entities) and those of *the qualificative adjective* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a qualificative adjective from *Verbs of diseases*. For example, *abras* ماير (leper; leprous', *acmā* أعمى 'blind', *kasīḥ* كسيح 'lame'.

Compatibility with the locative noun

There is no compatibility between the semantic features of Verbs of diseases and those of

the locative noun, where the features (object entity) of the verbs involve abstract entities, and those of the locative noun involves concrete entities. Therefore, it is not valid semantically to generate a locative noun from *Verbs of diseases*.

Class 04: Verbs of social and personal behaviour				
The object entity	The situation entity			
These verbs refer to abstract entities,	These verbs refer to state entities. They			
expressing non-physical, unseen and intangible	describe a state which is permanent or			
meanings (which are not accessible to any of	will last for a significant time.			
the five senses) that do not occupy physical				
space, such as 'to be wise, judicious', 'to have				
mercy upon' and 'to be generous, forgiving,				
tolerant'.				

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of social and personal behaviour* (that involve abstract and state entities) and those of *the active participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate an active participle from *Verbs of social and personal behaviour*. For example, *şābir* - one who is patient', *bāxil* + one who is stingy', *kārim* - one who is generous'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of social and personal behaviour* (that involve abstract and state entities) and those of *the passive participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of social and personal behaviour*. For example, $ma^{2}t\bar{u}r$ a passive participle from *Verbs of social and personal behaviour*. For example, $ma^{2}t\bar{u}r$ and 'generate 'generate' (behaviour)', $maxd\bar{u} \, c$ مخدوع 'cheated', $mabr\bar{u}s$ 'merciful (behaviour)'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of Verbs of social and personal

behaviour (that involve abstract and state entities) and those of the form of exaggeration (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a form of exaggeration from Verbs of social and personal behaviour. For example, $s\bar{a}bir$ صبور 'one who is so patient', $ra^{2}\bar{u}f$ ويؤف 'affectionate', mikrām مكرام 'munificent'.

Compatibility with the instrumental noun

There is no compatibility between the semantic features of *Verbs of social and personal behaviour* and those of *the instrumental noun*, where the features of the verbs involve abstract and state entities, and those of the instrumental noun involves concrete and event entities. Therefore, it is not valid semantically to generate an instrumental noun from *Verbs of social and personal behaviour*.

Compatibility with the qualificative adjective

There is compatibility between the semantic features of *Verbs of social and personal behaviour* (that involve abstract and state entities) and those of *the qualificative adjective* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a qualificative adjective from *Verbs of social and personal behaviour*. For example, *halīm حليم* 'clement; indulgent', *raḥīm رحيم* 'merciful', *xajil حليم* 'shy'.

Compatibility with the locative noun

There is no compatibility between the semantic features of *Verbs of social and personal behaviour* and those of *the locative noun*, where the features (object entity) of the verbs involve abstract entities, and those of the locative noun involves concrete entities. Therefore, it is not valid semantically to generate a locative noun from *Verbs of social and personal behaviour*.

Class 05: Verbs of emotions			
The object entity	The situation entity		
These verbs refer to abstract entities,	These verbs refer to state entities. They		
expressing non-physical, unseen and	describe a state which is permanent or will		

intangible meanings (which are not accessible	last for a significant time.
to any of the five senses) that do not occupy	
physical space, such as 'to love', 'to hate' and	
'to fear'.	

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of emotions* (that involve abstract and state entities) and those of *the active participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate an active participle from *Verbs of emotions*. For example, $c\bar{a}siq$ عاشق 'one who is lover', $k\bar{a}rih$ كار 'one who is hater', $q\bar{a}liq$ one who is anxious'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of emotions* (that involve abstract and state entities) and those of *the passive participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of emotions*. For example, *mabgūd* معفرض 'detested', *majzū*' معضروب 'anxious', *maġdūb* مغضروب 'angry'.

Compatibility with the form of exaggeration

Compatibility with the instrumental noun

There is no compatibility between the semantic features of *Verbs of emotions* and those of *the instrumental noun*, where the features of the verbs involve abstract and state entities, and those of the instrumental noun involves concrete and event entities. Therefore, it is not valid semantically to generate an instrumental noun from *Verbs of emotions*.

Compatibility with the qualificative adjective

There is compatibility between the semantic features of *Verbs of emotions* (that involve abstract and state entities) and those of *the qualificative adjective* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a qualificative adjective from *Verbs of emotions*. For example, *baġīd* بغيض 'hateful; repugnant', *hazīn* 'esad', *sacīd* سعيد 'happy'.

Compatibility with the locative noun

There is no compatibility between the semantic features of *Verbs of emotions* and those of *the locative noun*, where the features (object entity) of the verbs involve abstract entities, and those of the locative noun involves concrete entities. Therefore, it is not valid semantically to generate a locative noun from *Verbs of emotions*.

Class 06: Verbs of o	colouring
The object entity	The situation entity
These verbs refer to concrete entities, expressing	These verbs refer to state entities.
tangible meanings (of an action, event, or state	They describe a state which is
which is accessible to one or more of the five	permanent or will last for a significant
senses) that occupy physical space. They are	time.
accessible by one of the five senses i.e. they can	
be seen, such as 'to be or become blue', 'to be or	
become black', and 'to be or become pale'.	

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of colouring* (that involve concrete and state entities) and those of *the active participle* (that may also involve concrete and state entities). Therefore, it is valid semantically to generate an active participle from *Verbs of colouring*. For example, *bāhit* باهت 'pale', *qātim* 'jale', *qātim* 'dark', *zāriq* 'jolue'.

Compatibility with the passive participle

There is compatibility between the semantic features of Verbs of colouring (that involve

concrete and state entities) and those of *the passive participle* (that may also involve concrete and state entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of colouring*. For example, *maṣbūġ* مصبوغ 'dyed', *masūd*' مخضوب 'black', *maxdūb*', *maxdūb*', *maxdūb*'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of colouring* (that involve concrete and state entities) and those of *the active participle* (that may also involve concrete and state entities). Therefore, it is valid semantically to generate an active participle from *Verbs of colouring*. For example, *zarrāq* زرّاق 'one who makes something blue', *xaḍḍār* ضفّار 'one who makes something green', *şaffār* 'one who makes something yellow'.

Compatibility with the instrumental noun

There is no compatibility between the semantic features of *Verbs of colouring* and those of *the instrumental noun*, where the features of the verbs involve concrete and state entities, and those of the instrumental noun involves concrete and event entities. Therefore, it is not valid semantically to generate an instrumental noun from *Verbs of colouring*.

Compatibility with the qualificative adjective

There is compatibility between the semantic features of *Verbs of colouring* (that involve concrete and state entities) and those of *the qualificative adjective* (that may also involve concrete and state entities). Therefore, it is valid semantically to generate a qualificative adjective from *Verbs of colouring*. For example, *azraq* أزرق 'blue', *asham*' black, '*asham*' black, '*asfar*' أصفر 'yellow'.

Compatibility with the locative noun

There is compatibility between the semantic features of *Verbs of colouring* (that involve concrete and state entities) and those of *the locative noun* (that may also involve concrete and state entities). Therefore, it is valid semantically to generate a locative noun from *Verbs of colouring*. For example, *maxdarah* مخضرة 'a green place (meadow)', *masfar* 'a yellowish place', *mazraq* مزرق 'a bluish place'.

Class 07 Verbs of bodily qualities			
The object entity	The situation entity		
These verbs refer to abstract entities, expressing	These verbs refer to state entities. They		
non-physical, unseen and intangible meanings	describe a state which is permanent or		
(which are not accessible to any of the five	will last for a significant time.		
senses) that do not occupy physical space, such			
as such as 'to be or become hot or warm', 'to be			
or become soft, smooth' and 'to be or become			
big, large'.			
Compatibility with the a	active participle		
There is compatibility between the semantic fea	atures of Verbs of bodily qualities (that		
involve abstract and state entities) and those of <i>th</i>	<i>e active participle</i> (that may also involve		
abstract and state entities). Therefore, it is va	alid semantically to generate an active		
participle from Verbs of bodily qualities. For exa	mple, <i>nā ° im</i> ناعم 'soft', <i>kārih</i> 'بادن 'fat',		
<i>qāliq</i> ساخن 'hot'.			
Compatibility with the passive participle			
There is compatibility between the semantic features of Verbs of bodily qualities (that			
involve abstract and state entities) and those of the passive participle (that may also			
involve abstract and state entities). Therefore, it is valid semantically to generate a passive			
participle from Verbs of bodily qualities. For example, matqūl مرقون heavy', marqūq مرقوق heavy', marqūq			
'thin', <i>madub</i> مذوب 'melt'.			
Compatibility with the form of exaggeration			
There is compatibility between the semantic features of Verbs of bodily qualities (that			
involve abstract and state entities) and those of the form of exaggeration (that may also			
involve abstract and state entities). Therefore, it is valid semantically to generate a form of			
exaggeration from Verbs of bodily qualities. For example, <u>taqqāl</u> someone/something			
that makes something heavy', <i>barrād</i> برّاد 'someone/something that makes something			
cold', <i>saxxān</i> سخّان 'someone/something that makes something warm'.			

Compatibility with the instrumental noun

There is no compatibility between the semantic features of *Verbs of bodily qualities* and those of *the instrumental noun*, where the features of the verbs involve abstract and state entities, and those of the instrumental noun involves concrete and event entities. Therefore, it is not valid semantically to generate an instrumental noun from *Verbs of bodily qualities*.

Compatibility with the qualificative adjective

There is compatibility between the semantic features of *Verbs of bodily qualities* (that involve abstract and state entities) and those of *the qualificative adjective* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a qualificative adjective from *Verbs of bodily qualities*. For example, *badīn* بدین 'obese; fat', *samīk* صغیر 'thick', *şaġīr* صغیر 'small; tiny'.

Compatibility with the locative noun

There is no compatibility between the semantic features of *Verbs of bodily qualities* and those of *the locative noun*, where the features (object entity) of the verbs involve abstract entities, and those of the locative noun involves concrete entities. Therefore, it is not valid semantically to generate a locative noun from *Verbs of bodily qualities*.

Class 08: Verbs of motion					
The object entity	The situation entity				
These verbs refer to concrete entities,	These verbs refer to event entities. They				
expressing tangible meanings (of an action,	involve a change from one state to another.				
event or state which is accessible to one or	These verbs involve a movement from a				
event, of state which is accessible to one of	These veros involve a movement from a				
more of the five senses) that occupy	location to another through a path. They				
, IJ					
physical space, such as 'to come', 'to go',	involve a kind of functioning.				
'to leave' and 'to run'.					

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of motion* (that involve concrete and event entities) and those of *the active participle* (that may also involve

concrete and event entities). Therefore, it is valid semantically to generate an active participle from *Verbs of motion*. For example, $\bar{a}tin$ (in who is coming', $d\bar{a}hib$ (in who is going', $z\bar{a}hif$ (in who is crawling'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of motion* (that involve concrete and event entities) and those of *the passive participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of motion*. For example, $ma \, ^c b \bar{u} r$ معبور 'crossed', $marj \bar{u} \, ^c$ 'returned to', $mawf \bar{u} d$ موفود 'come to'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of motion* (that involve concrete and event entities) and those of *the form of exaggeration* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of motion*. For example, *jawwāb* جوّاب 'frequent traveller', *dahhāb* ذهّاب 'frequent goer', *tayyār* طَيّار 'pilot'.

Compatibility with the instrumental noun

There is compatibility between the semantic features of *Verbs of motion* and those of *the instrumental noun*, where both of them involves concrete and event entities. Therefore, it is valid semantically to generate an instrumental noun from *Verbs of motion*. For example, *sayyārah مِصْعَد 'car', mis cad مِصْعَد 'elevator'*, and *darrājah در 'bicycle'*.

Compatibility with the qualificative adjective

There is no compatibility between the semantic features of *Verbs of motion* and those of *the qualificative adjective*, where the features (situation entity) of the verbs involve event entities, and those of the qualificative adjective involves state entities. Therefore, it is not valid semantically to generate a qualificative adjective from *Verbs of motion*.

Compatibility with the locative noun

There is compatibility between the semantic features of *Verbs of motion* (that involve concrete and event entities) and those of *the locative noun* (that may also involve concrete

and event entities). Therefore, it is valid semantically to generate a locative noun from *Verbs of motion*. For example, *madxal* مدخل 'entry (way)', *maxraj* مخرج 'escape ; exit', and *ma cbar* معبر 'corridor ; path'.

Class 09: Verbs of swimming				
The object entity	The situation entity			
These verbs refer to concrete entities, expressing	These verbs refer to event entities.			
tangible meanings (of an action, event, or state	They involve a change from one state			
which is accessible to one or more of the five	to another. These verbs refer to events			
senses) that occupy physical space, such as 'to	that involve a kind of functioning.			
swim', 'to float' and 'to dive'.				
Compatibility with the a	etive perticiple			
Compationity with the a	ictive participie			
There is compatibility between the semantic feat	ures of Verbs of swimming (that involve			
concrete and event entities) and those of the active participle (that may also involve				
concrete and event entities). Therefore, it is valid semantically to generate an active				
participle from Verbs of swimming. For example, sābih سابح 'one who is swimming', gāriq				
one who is drowning', <i>ġāṭis</i> غاطس 'one who is diving'.				
Compatibility with the passive participle				
There is compatibility between the semantic features of Verbs of swimming (that involve				
concrete and event entities) and those of the passive participle (that may also involve				
concrete and event entities). Therefore, it is valid semantically to generate a passive				
participle from Verbs of swimming. For example, masbūh مسبوح 'swum in', magtjūs				
'drowned in', <i>maġrūq</i> مغروق drowned in'.				

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of swimming* (that involve concrete and event entities) and those of *the form of exaggeration* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of swimming*. For example, *sabbāḥ* (professional)

swimmer', *cawwām* عوّام 'professional floater', *gattās* غطّاس 'expert diver'.

Compatibility with the instrumental noun

There is compatibility between the semantic features of *Verbs of swimming* and those of *the instrumental noun*, where both of them involves concrete and event entities. Therefore, it is valid semantically to generate an instrumental noun from *Verbs of swimming*. For example, *ġawwāṣah* غطّاسة *'submarine'*, *cawwāmah* عرّامة *'float'*, and *ġaṭṭāsah 'diver'*.

Compatibility with the qualificative adjective

There is no compatibility between the semantic features of *Verbs of swimming* and those of *the qualificative adjective*, where the features (situation entity) of the verbs involve event entities, and those of the qualificative adjective involves state entities. Therefore, it is not valid semantically to generate a qualificative adjective from *Verbs of swimming*.

Compatibility with the locative noun

There is compatibility between the semantic features of *Verbs of swimming* (that involve concrete and event entities) and those of *the locative noun* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a locative noun from *Verbs of swimming*. For example, *masbaḥ* معنون 'swimming pool', *maġṭas* 'bath ; bathtub', and *maġraq* `a place of sinking'.

Class 10: Verbs of lo	cation/place
The object entity	The situation entity
These verbs refer to concrete entities, expressing	These verbs refer to state entities. They
tangible meanings (of an action, event, or state	describe a state which is permanent or
which is accessible to one or more of the five	will last for a significant time.
senses) that occupy physical space, such as 'to	
be or become narrow, tight, close', 'to be or	
become far, faraway, far-off, distant, remote'	
and 'be or become deep(er)'.	

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of location/place* (that involve concrete and state entities) and those of *the active participle* (that may also involve concrete and state entities). Therefore, it is valid semantically to generate an active participle from *Verbs of location/place*. For example, $s\bar{a}hiq$ ساحق 'deep, $c\bar{a}lin$ 'high', $w\bar{a}si^{c}$ واسع 'wide'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of location/place* (that involve concrete and state entities) and those of *the passive participle* (that may also involve concrete and state entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of location/place*. For example, $ma \, ^c \, luw$ معموق 'elevated', $ma \, ^c \, m\bar{u}q$ معموق 'deep', maq siy مغموق 'remote'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of location/place* (that involve concrete and state entities) and those of *the form of exaggeration* (that may also involve concrete and state entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of location/place*. For example, *mišrāq* فيشراق 'directed toward the east', *cammāq* عمّاق 'very deeper', *wasū c* وسوع 'dilatable'.

Compatibility with the instrumental noun

There is no compatibility between the semantic features of *Verbs of location/place* and those of *the instrumental noun*, where the features of the verbs involve concrete and state entities, and those of the instrumental noun involves concrete and event entities. Therefore, it is not valid semantically to generate an instrumental noun from *Verbs of location/place*.

Compatibility with the qualificative adjective

qarīb عميق 'nearby; close', *camīq* قريب 'deep'.

Compatibility with the locative noun

There is compatibility between the semantic features of *Verbs of location/place* (that involve concrete and event entities) and those of *the locative noun* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a locative noun from *Verbs of location/place*. For example, *mašriq* مشرق 'orient, east; the eastern part of a country (place)', *maġrib* 'ast; the western part of a country (place)', *maġrib* 'ast; the western part of a country (place)', and *maḍiq* 'strait'.

Class 11: Verbs of violence and abuse			
The object entity	The situation entity		
These verbs refer to concrete entities, expressing	These verbs refer to event entities.		
tangible meanings (of an action, event, or state	They involve a change from one state		
which is accessible to one or more of the five	to another.		
senses) that occupy physical space, such as 'to			
break, smash, crush', 'to hit', and 'to throw'.			
Compatibility with the active participle			

There is compatibility between the semantic features of *Verbs of violence and abuse* (that involve concrete and event entities) and those of *the active participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate an active participle from *Verbs of violence and abuse*. For example, *qātil* نال 'killer', *kāsir* 'breaker', *dārib* 'intter'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of violence and abuse* (that involve concrete and event entities) and those of *the passive participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of violence and abuse*. For example, *majrū*, majrū, majr

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of violence and abuse* (that involve concrete and event entities) and those of *the form of exaggeration* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of violence and abuse*. For example, *qattāl* 'mortal', *jallād* 'executioner', *darrāb* 'ivel'.

Compatibility with the instrumental noun

There is compatibility between the semantic features of *Verbs of violence and abuse* and those of *the instrumental noun*, where both of them involves concrete and event entities. Therefore, it is valid semantically to generate an instrumental noun from *Verbs of violence and abuse*. For example, *mirjam مِرْجَم 'launcher'*, *mišnaqah مِسْنَفَة' 'gibbet'*, and *miș c āq pour electric stick'*.

Compatibility with the qualificative adjective

There is no compatibility between the semantic features of *Verbs of violence and abuse* and those of *the qualificative adjective*, where the features (situation entity) of the verbs involve event entities, and those of the qualificative adjective adjective involves state entities. Therefore, it is not valid semantically to generate a qualificative adjective from *Verbs of violence and abuse*.

Compatibility with the locative noun

There is compatibility between the semantic features of *Verbs of violence and abuse* (that involve concrete and event entities) and those of *the locative noun* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a locative noun from *Verbs of violence and abuse*. For example, *maḥraqah* محرقة 'a place of burning (holocaust)', *mašnaqah* مشنقة 'a place of gibbet', and *mat can* 'a place of stabbing'.

Class 12: Verbs of mental process						
		The obj	iect e	ntity		The situation entity
These	verbs	refer	to	abstract	entities,	These verbs refer to state entities. They

expressing non-physical, unseen and intangible	describe a state which is permanent or
meanings (which are not accessible to any of	will last for a significant time.
the five senses) that do not occupy physical	
space, such as 'to calculate', 'to look into' and	
'to solve'.	

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of mental process* (that involve abstract and state entities) and those of *the active participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate an active participle from *Verbs of mental process*. For example, $b\bar{a}hi\underline{i}$ 'researcher', $h\bar{a}sib$ 'counter', $z\bar{a}n$ 'difference'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of mental process* (that involve abstract and state entities) and those of *the passive participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of mental process*. For example, *mabhūth* مجدوث 'studied', *mafrūd* مفروض 'supposed', *mahsūb* محسوب 'calculated'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of mental process* (that involve abstract and state entities) and those of *the form of exaggeration* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of mental process*. For example, $bah\bar{a}t$: superior researcher', *haffāz* : one who has strong memory', *hallāl* : c (problem) solver'.

Compatibility with the instrumental noun

There is no compatibility between the semantic features of *Verbs of mental process* and those of *the instrumental noun*, where the features (object entity) of the verbs involve abstract entities, and those of the instrumental noun involves concrete entities. Therefore, it is not valid semantically to generate an instrumental noun from *Verbs of mental process*.

Compatibility with the qualificative adjective

There is compatibility between the semantic features of *Verbs of mental process* (that involve abstract and state entities) and those of *the qualificative adjective* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a qualificative adjective from *Verbs of mental process*. For example, *fațīn* fatin fatin

Compatibility with the locative noun

There is no compatibility between the semantic features of *Verbs of mental process* and those of *the locative noun*, where the features (object entity) of the verbs involve abstract entities, and those of the locative noun involves concrete entities. Therefore, it is not valid semantically to generate a locative noun from *Verbs of mental process*.

Class 13: Verbs of financial transaction		
The object entity	The situation entity	
These verbs refer to abstract entities,	These verbs refer to state entities. They	
expressing non-physical, unseen and intangible	describe a state which is permanent or	
meanings (which are not accessible to any of	will last for a significant time.	
the five senses) that do not occupy physical		
space, such as 'to discount', ' to borrow', and		
'to earn'.		
Compatibility with the estive next sinle		

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of financial transaction* (that involve abstract and state entities) and those of *the active participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate an active participle from *Verbs of financial transaction*. For example, $b\bar{a}^{\circ}i^{\circ}c^{\circ}$ بائع 'seller', $d\bar{a}^{\circ}in$ (that involve abstract involve abstract and state entities).

Compatibility with the passive participle

There is compatibility between the semantic features of Verbs of financial transaction (that

involve abstract and state entities) and those of *the passive participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of financial transaction*. For example, *mabhūth مجرود (supposed', mahsūb محسوب 'calculated'.*

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of financial transaction* (that involve abstract and state entities) and those of *the form of exaggeration* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of financial transaction*. For example, $bayy\bar{a}^\circ$ بيّاع 'salesclerk', *qarrād* فَرَّاض 'person who lends money', *ġaššāš* نُقْرَّاض 'deceitful person in business'.

Compatibility with the instrumental noun

There is no compatibility between the semantic features of *Verbs of financial transaction* and those of *the instrumental noun*, where the features (object entity) of the verbs involve abstract entities, and those of the instrumental noun involves concrete entities. Therefore, it is not valid semantically to generate an instrumental noun from *Verbs of financial transaction*.

Compatibility with the qualificative adjective

There is compatibility between the semantic features of *Verbs of financial transaction* (that involve abstract and state entities) and those of *the qualificative adjective* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a qualificative adjective from *Verbs of financial transaction*. For example, *xasīr* 'loser', $rafī^c$ thigh price', and raxīs 'cheap; inexpensive'.

Compatibility with the locative noun

There is no compatibility between the semantic features of *Verbs of financial transaction* and those of *the locative noun*, where the features (object entity) of the verbs involve abstract entities, and those of the locative noun involves concrete entities. Therefore, it is not valid semantically to generate a locative noun from *Verbs of financial transaction*.

Class 14: Verbs of agriculture	
The object entity	The situation entity
These verbs refer to concrete entities, expressing	These verbs refer to event entities.
tangible meanings (of an action, event, or state	They involve a change from one state
which is accessible to one or more of the five	to another.
senses) that occupy physical space, such as 'to	
sow', 'to irrigate, water', and 'to harvest'.	

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of agriculture* (that involve concrete and event entities) and those of *the active participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate an active participle from *Verbs of agriculture*. For example, $h\bar{a}ri\underline{t}$ - L -

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of agriculture* (that involve concrete and event entities) and those of *the passive participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of agriculture*. For example, $mazr\bar{u}^c$ مزروع 'planted', $mahs\bar{u}d$ محصود 'harvested', marwiy مروي 'watered'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of agriculture* (that involve concrete and event entities) and those of *the form of exaggeration* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of agriculture*. For example, *baddār* بذار 'breeders seed', *haffār* خفّار 'inscriber', *zarrā*^c زرّاع 'cultivator'.

Compatibility with the instrumental noun

There is compatibility between the semantic features of *Verbs of agriculture* and those of *the instrumental noun*, where both of them involves concrete and event entities. Therefore,

it is valid semantically to generate an instrumental noun from *Verbs of agriculture*. For example, *miḥrāṯ* مِحْرَات 'plough', *ḥaṣṣādah*' حصّادة 'mower', and *ḥaffārah*' مِحْرَات 'digger'.

Compatibility with the qualificative adjective

There is no compatibility between the semantic features of *Verbs of agriculture* and those of *the qualificative adjective*, where the features (situation entity) of the verbs involve event entities, and those of the qualificative adjective involves state entities. Therefore, it is not valid semantically to generate a qualificative adjective from *Verbs of agriculture*.

Compatibility with the locative noun

There is compatibility between the semantic features of *Verbs of agriculture* (that involve concrete and event entities) and those of *the locative noun* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a locative noun from *Verbs of agriculture*. For example, *maštal* مترب 'plant nursery', *maġras* 'a place of planting', and *mazra cah*' مزرعة 'farm'.

Class 15: Verbs of desire and request		
The object entity	The situation entity	
These verbs refer to abstract entities,	These verbs refer to state entities. They	
expressing non-physical, unseen and intangible	describe a state which is permanent or	
meanings (which are not accessible to any of	will last for a significant time.	
the five senses) that do not occupy physical		
space, such as 'to desire, wish', 'to beg', and		
'to need'.		
Compatibility with the active participle		
There is compatibility between the semantic features of Verbs of desire and request (that		
involve abstract and state entities) and those of the <i>active participle</i> (that may also involve		

There is compatibility between the semantic features of *Verbs of desire and request* (that involve abstract and state entities) and those of the *active participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate an active participle from *Verbs of desire and request*. For example, $c\bar{a}mil$ fightharpoondown between the semantic features of*Verbs of desire and request* $. For example, <math>c\bar{a}mil$ fightharpoondown between the semantic features of*Verbs of desire and request* $. For example, <math>c\bar{a}mil$ fightharpoondown between the semantic features of*Verbs of desire and request* $. For example, <math>c\bar{a}mil$ fightharpoondown between the semantic features of*Verbs of desire and request*.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of desire and request* (that involve abstract and state entities) and those of *the passive participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of desire and request*. For example, *ma^omūl* مأمول 'hoped', *marģūb* 'cequired'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of desire and request* (that involve abstract and state entities) and those of *the form of exaggeration* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of desire and request*. For example, *tawwāq* توّاق 'agog', *tamūḥ* 'agog', *tamūḥ* 'demanding'.

Compatibility with the instrumental noun

There is no compatibility between the semantic features of *Verbs of desire and request* and those of *the instrumental noun*, where the features of the verbs involve abstract and state entities, and those of the instrumental noun involves concrete and event entities. Therefore, it is not valid semantically to generate an instrumental noun from *Verbs of desire and request*.

Compatibility with the qualificative adjective

There is compatibility between the semantic features of *Verbs of desire and request* (that involve abstract and state entities) and those of *the qualificative adjective* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a qualificative adjective from *Verbs of desire and request*. For example, *jaši^c* جثيع 'greedy', *harīş* حريص 'eager', *tami^c* طمع 'avid'.

Compatibility with the locative noun

There is no compatibility between the semantic features of *Verbs of desire and request* and those of *the locative noun*, where the features (object entity) of the verbs involve abstract entities, and those of the locative noun involves concrete entities. Therefore, it is not valid
semantically to generate a locative noun from Verbs of desire and request.

Class 16: Verbs of intention	
The object entity	The situation entity
These verbs refer to abstract entities, expressing	These verbs refer to state entities.
non-physical, unseen and intangible meanings	They describe a state which is
(which are not accessible to any of the five senses)	permanent or will last for a
that do not occupy physical space, such as 'to seek	significant time.
(to) ', 'to intend', 'to aim', and 'to intend'.	

Compatibility with the active participle

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of intention* (that involve abstract and state entities) and those of *the passive participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of intention*. For example, *maqṣūd* مقصود 'intended', *ma c niy* منزوي 'meant', *manwiy* منزوي 'proposed'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of intention* (that involve abstract and state entities) and those of *the form of exaggeration* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of intention*. For example, *tamūh* dat dat 'ambitious', *haddāf* هدّاف 'skilled marksman', *sa c cā* with the semantical of the semantic features o

Compatibility with the instrumental noun

There is no compatibility between the semantic features of *Verbs of intention* and those of *the instrumental noun*, where the features of the verbs involve abstract and state entities, and those of the instrumental noun involves concrete and event entities. Therefore, it is not valid semantically to generate an instrumental noun from *Verbs of intention*.

Compatibility with the qualificative adjective

There is compatibility between the semantic features of *Verbs of intention* (that involve abstract and state entities) and those of *the qualificative adjective* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a qualificative adjective from *Verbs of intention*. For example, *tamih* طمخ 'avaricious', *saciyy* 'striving', *camid* عبد 'determined'.

Compatibility with the locative noun

There is no compatibility between the semantic features of *Verbs of intention* and those of *the locative noun*, where the features (object entity) of the verbs involve abstract entities, and those of the locative noun involves concrete entities. Therefore, it is not valid semantically to generate a locative noun from *Verbs of intention*.

Class 17: Verbs of combining and constructing		
The object entity	The situation entity	
These verbs refer to concrete entities, expressing	These verbs refer to event entities.	
tangible meanings (of an action, event, or state	They involve a change from one	
which is accessible to one or more of the five	state to another.	
senses) that occupy physical space, such as 'to		
build, construct', 'to collect', 'to sew', and 'to mix'.		

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of combining and constructing* (that involve concrete and event entities) and those of *the active participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to

generate an active participle from *Verbs of combining and constructing*. For example, *bānin بان* 'builder', *xā^sitah* خائطة 'sewer', *jāmi* ^c جامع 'gatherer'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of combining and constructing* (that involve concrete and event entities) and those of *the passive participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of combining and constructing*. For example, *mabniy* of built', *majmū*^c of combining and constructing. For example, 'field, attached'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of combining and constructing* (that involve concrete and event entities) and those of *the form of exaggeration* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of combining and constructing*. For example, *jammā* ^c جمّاع 'collector', *laḥḥām* 'solderer', *nassāj* 'weaver'.

Compatibility with the instrumental noun

There is compatibility between the semantic features of *Verbs of combining and constructing* and those of *the instrumental noun*, where both of them involves concrete and event entities. Therefore, it is valid semantically to generate an instrumental noun from *Verbs of combining and constructing*. For example, *mixyatah* مِخْيَطَة 'sewing machine', *mixlat* مِخْيَطَة 'concrete mixer', and *milham* مِلْحَم 'welding apparatus'.

Compatibility with the qualificative adjective

There is no compatibility between the semantic features of *Verbs of combining and constructing* and those of *the qualificative adjective*, where the features (situation entity) of the verbs involve event entities, and those of the qualificative adjective adjective involves state entities. Therefore, it is not valid semantically to generate a qualificative adjective from *Verbs of combining and constructing*.

Compatibility with the locative noun

There is compatibility between the semantic features of *Verbs of combining and constructing* (that involve concrete and event entities) and those of *the locative noun* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a locative noun from *Verbs of combining and constructing*. For example, *mabnā* generate a locative noun from *Verbs of combining and constructing*. For example, *mabnā* of 'a place of meeting', and *maḥar* of 'insectary'.

Class 18: Verbs of sending and carrying	
The object entity	The situation entity
These verbs refer to concrete entities, expressing	These verbs refer to event entities.
tangible meanings (of an action, event, or state	They involve a change from one
which is accessible to one or more of the five	state to another.
senses) that occupy physical space, such as 'to	
send, dispatch, forward', 'to carry, bear', and 'to	
ship'.	

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of sending and carrying* (that involve concrete and event entities) and those of *the active participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate an active participle from *Verbs of combining and constructing*. For example, $\bar{s}\bar{a}hin$ \hat{u} 'shipper', $h\bar{a}mil$ \hat{v} bearer', $s\bar{a}hib$ \hat{u} - \hat{v} puller'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of sending and carrying* (that involve concrete and event entities) and those of *the passive participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of sending and carrying*. For example, $mab \, {}^c \bar{u} t$ 'sent', *maḥmūl* 'carried', *mašḥūn* 'shipped'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of Verbs of sending and carrying

(that involve concrete and event entities) and those of *the form of exaggeration* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of combining and constructing*. For example, *ḥammāl* form of exaggeration from *Verbs of combining and constructing*. For example, *ḥammāl* carrier', *cattāl* عتّال 'porter, bearer', *sawwāq* حمّال

Compatibility with the instrumental noun

There is compatibility between the semantic features of *Verbs of sending and carrying* and those of *the instrumental noun*, where both of them involves concrete and event entities. Therefore, it is valid semantically to generate an instrumental noun from *Verbs of sending and carrying*. For example, *ḥammālah* حصّالة 'hand barrow', *misḥab* مِسْحَب 'drag', and *midfā* ^c مِدْفَاع 'impeller'.

Compatibility with the qualificative adjective

There is no compatibility between the semantic features of *Verbs of sending and carrying*. and those of *the qualificative adjective*, where the features (situation entity) of the verbs involve event entities, and those of the qualificative adjective involves state entities. Therefore, it is not valid semantically to generate a qualificative adjective from *Verbs of sending and carrying*.

Compatibility with the locative noun

There is compatibility between the semantic features of *Verbs of combining and constructing* (that involve concrete and event entities) and those of *the locative noun* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a locative noun from *Verbs of combining and constructing*. For example, *mašḥan* of a place of shipping', *majraf* معبر 'a place of drifting (torrent)', and *ma c bar* of passageway'.

Class 19: Verbs of Separating and Disassembling	
The object entity	The situation entity
These verbs refer to concrete entities, expressing	These verbs refer to event entities.
tangible meanings (of an action, event, or state	They involve a change from one state

which is accessible to one or more of the five	to another.
senses) that occupy physical space, such as 'to	
open', 'to separate', 'to unscrew', and 'to divide'.	

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of separating and disassembling* (that involve concrete and event entities) and those of *the active participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate an active participle from *Verbs of separating and disassembling*. For example, *c* $\bar{a}zil$ (insulator', $s\bar{a}di^{c}$ and c crushing', $s\bar{a}hiq$ and c destroyer'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of separating and disassembling* (that involve concrete and event entities) and those of *the passive participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of separating and disassembling*. For example, *mabtūr* معزول off, $ma^c z \bar{u}l$ (the vertice of the passive participle) and the vert entities of the passive participle (the vertice) and the vertice of the passive participle (the vertice) and the vert entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of separating and disassembling*. For example, *mabtūr* (the vert off), $ma^c z \bar{u}l$ (the vert off), $mac^c z \bar{u}l$ (the vert of the vert

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of separating and disassembling* (that involve concrete and event entities) and those of *the form of exaggeration* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of separating and disassembling*. For example, *haddām* هدّام 'demolisher; destroyer', *šaṭār شطّاح*' divider', *qaṭā c* tutter'.

Compatibility with the instrumental noun

There is compatibility between the semantic features of *Verbs of separating and disassembling* and those of *the instrumental noun*, where both of them involves concrete and event entities. Therefore, it is valid semantically to generate an instrumental noun from *Verbs of separating and disassembling*. For example, *miftā*h مِقْتَاح 'key', *mifakk مِقْتَاح 'screw-driver*', and *minšār مِنْتَنَار 'saw'*.

Compatibility with the qualificative adjective

There is no compatibility between the semantic features of *Verbs of separating and disassembling* and those of *the qualificative adjective*, where the features (situation entity) of the verbs involve event entities, and those of the qualificative adjective adjective involves state entities. Therefore, it is not valid semantically to generate a qualificative adjective from *Verbs of separating and disassembling*.

Compatibility with the locative noun

There is compatibility between the semantic features of *Verbs of separating and disassembling* (that involve concrete and event entities) and those of *the locative noun* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a locative noun from *Verbs of separating and disassembling*. For example, *ma*^c *zal* راحد 'a place that is isolated', *mahdam مهذم* 'a place that is destroyed', and *mafṣal* 'a place that is divided'.

Class 20: Verbs of removing	
	-
The object entity	The situation entity
These verbs refer to concrete entities, expressing	These verbs refer to event entities.
tangible meanings (of an action, event, or state	They involve a change from one
which is accessible to one or more of the five	state to another.
senses) that occupy physical space, such as 'to	
extract', 'to remove, take away, take off', and 'to	
erase'.	

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of removing* (that involve concrete and event entities) and those of *the active participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate an active participle from *Verbs of removing*. For example, $h\bar{a}dif$ climinator', $q\bar{a}li$ climinator', $q\bar{a}li$ climinator', $n\bar{a}tif$ if 'extractor', $n\bar{a}tif$ if 'plucker'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of removing* (that involve concrete and event entities) and those of *the passive participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of removing*. For example, $mahd\bar{u}f$ محذوف 'deleted', $mahl\bar{u}q$ محلوق 'shaved', $manz\bar{u}^{c}$ منزوع 'extracted'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of removing* (that involve concrete and event entities) and those of *the form of exaggeration* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of removing*. For example, *hallāq* ألم 'barber', *mihdāf* محذاف 'extractor'.

Compatibility with the instrumental noun

There is compatibility between the semantic features of *Verbs of removing* and those of *the instrumental noun*, where both of them involves concrete and event entities. Therefore, it is valid semantically to generate an instrumental noun from *Verbs of removing*. For example, *maḥhāyah* محّاية 'eraser', *miqšaṭah* مِقْتَسَطَة 'scraper', and *miknasah* محّاية 'broom'.

Compatibility with the qualificative adjective

There is no compatibility between the semantic features of *Verbs of removing* and those of *the qualificative adjective*, where the features (situation entity) of the verbs involve event entities, and those of the qualificative adjective involves state entities. Therefore, it is not valid semantically to generate a qualificative adjective from *Verbs of removing*.

Compatibility with the locative noun

There is compatibility between the semantic features of *Verbs of removing* (that involve concrete and event entities) and those of *the locative noun* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a locative noun from *Verbs of removing*. For example, *maqla* ^c مقلع 'quarry', *makšat* 'a place that is scratched', and *mamsah* 'analyce' 'a place that is wiped off'.

Class 21: Verbs of bending	
The object entity	The situation entity
These verbs refer to concrete entities,	These verbs refer to event entities. They
expressing tangible meanings (of an action,	involve a change from one state to
event, or state which is accessible to one or	another.
more of the five senses) that occupy physical	
space, such as 'to fold', 'to wrap up' and 'to	
turn'.	
Compatibility with the active participle	

There is compatibility between the semantic features of *Verbs of bending* (that involve concrete and event entities) and those of the *active participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate an active participle from *Verbs of bending*. For example, $\underline{t}anin$ ثانِ 'someone/something that folds', *harif* خارف' someone/something that deviates', $l\bar{a}ff$ someone/something that turns'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of bending* (that involve concrete and event entities) and those of *the passive participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of bending*. For example, *matwiy* مطوي 'folded', *maqlūb* 'determined', *malwiy* 'wrench'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of bending* (that involve concrete and event entities) and those of *the form of exaggeration* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of bending*. For example, *qallāb* نقلاب 'reversible; tiltable; tippable', *fattāl* نقاف 'twister', *laffāf* نفاف 'wrapper'.

Compatibility with the instrumental noun

There is compatibility between the semantic features of *Verbs of bending* and those of *the instrumental noun*, where both of them involves concrete and event entities. Therefore, it is valid semantically to generate an instrumental noun from *Verbs of bending*. For example, *mitwā* مِطْوى 'bender', *laffāfah* لفافة 'roll', and *mitnā* مِطْوى 'wrap'.

Compatibility with the qualificative adjective

There is no compatibility between the semantic features of *Verbs of bending* and those of *the qualificative adjective*, where the features (situation entity) of the verbs involve event entities, and those of the qualificative adjective involves state entities. Therefore, it is not valid semantically to generate a qualificative adjective from *Verbs of bending*.

Compatibility with the locative noun

There is compatibility between the semantic features of *Verbs of bending* (that involve concrete and event entities) and those of *the locative noun* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a locative noun from *Verbs of bending*. For example, *malaff* مطوى 'a place of wrapping', *matwā* of a place of folding', and *maqlab* 'a place of turning'.

Class 22: Verbs of decorating and transcribing	
The object entity	The situation entity
The object entity	The situation entity
These verbs refer to concrete entities, expressing	These verbs refer to event entities.
tangible meanings (of an action, event, or state	They involve a change from one state
which is accessible to one or more of the five	to another.
senses) that occupy physical space, such as 'to	
stamp, imprint, print', 'to draw, trace 'to polish,	
burnish', 'to hew (out), and 'to tattoo'.	

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of decorating and transcribing* (that involve concrete and event entities) and those of *the active participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to

generate the active participle from Verbs of decorating and transcribing. For example, $t\bar{a}bi\ ^{\circ}$ ناحت 'someone/something that prints', $r\bar{a}sim$ راسم 'someone/something that draws', $n\bar{a}hit$ ناحت 'someone/something that sculptures'.

Compatibility with the passive participle

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of decorating and transcribing* (that involve concrete and event entities) and those of *the form of exaggeration* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of decorating and transcribing*. For example, *rassām* رستام 'artist; portraitist', *dahhān نحّان* 'someone who paints', *naḥāt* نحّات 'carver; chiseler'.

Compatibility with the instrumental noun

There is compatibility between the semantic features of *Verbs of decorating and transcribing* and those of *the instrumental noun*, where both of them involves concrete and event entities. Therefore, it is valid semantically to generate an instrumental noun from *Verbs of decorating and transcribing*. For example, *mixtam* مِخْتَم 'stamp', *mitba c a مَ*نْبَعة 'printer', and *naḥātah*' نحّاتة 'machine sculpture'.

Compatibility with the qualificative adjective

There is no compatibility between the semantic features of *Verbs of decorating and transcribing* and those of *the qualificative adjective*, where the features (situation entity) of the verbs involve event entities, and those of the qualificative adjective adjective involves state entities. Therefore, it is not valid semantically to generate a qualificative adjective from *Verbs of decorating and transcribing*.

Compatibility with the locative noun

There is compatibility between the semantic features of *Verbs of decorating and transcribing* (that involve concrete and event entities) and those of *the locative noun* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a locative noun from *Verbs of decorating and transcribing*. For example, *marsam* (applice of drawing (atelier)', *maktab* مكتب 'a place of writing (writing desk, office)', and *maşbaġah* 'acuse'.

Class 23: Verbs of measure		
The object entity	The situation entity	
These verbs refer to concrete entities,	These verbs refer to event entities. They	
expressing tangible meanings (of an action,	involve a change from one state to	
event, or state which is accessible to one or	another.	
more of the five senses) that occupy physical		
space, such as 'to measure', 'to gauge', and 'to		
weigh'.		

Compatibility with the active participle

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of measure* (that involve concrete and event entities) and those of *the passive participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of measure*. For example, *makıl* مَكِيل 'gauged', *maşı c*

'measured', mawzūn موزون 'weighted'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of measure* (that involve concrete and event entities) and those of *the form of exaggeration* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of measure*. For example, *wazzān* وَزَان the person who weights', *hāmil* فَيَاس the person who gauges', *qayyās* فَيَال

Compatibility with the instrumental noun

There is compatibility between the semantic features of *Verbs of measure* and those of *the instrumental noun*, where both of them involves concrete and event entities. Therefore, it is valid semantically to generate an instrumental noun from *Verbs of measure*. For example, *miqyās* مِذْرَان 'gauge', *mikyāl* مِكْتَال 'scoop', and *mizān* مِيزَان 'scale'.

Compatibility with the qualificative adjective

There is no compatibility between the semantic features of *Verbs of measure* and those of *the qualificative adjective*, where the features (situation entity) of the verbs involve event entities, and those of the qualificative adjective involves state entities. Therefore, it is not valid semantically to generate a qualificative adjective from *Verbs of measure*.

Compatibility with the locative noun

There is compatibility between the semantic features of *Verbs of measure* (that involve concrete and event entities) and those of *the locative noun* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a locative noun from *Verbs of measure*. For example, *makāl* مكال 'a place of gauging', *maqās* 'a place that is measured', and *mawzin* 'a place of weighing'.

Class 24: Verbs of quantity and size	
The object entity	The situation entity

These verbs refer to abstract entities, expressing
non-physical, unseen and intangible meanings
(which are not accessible to any of the five
senses) that do not occupy physical space, such as
such as 'to increase, grow', 'to decrease,
diminish', and 'to shrink, dwindle'.These verbs refer to state entities. They
describe a state which is permanent or
will last for a significant time.

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of quantity and size* (that involve abstract and state entities) and those of *the active participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate an active participle from *Verbs of quantity and size*. For example, $z\bar{a}^{2}id$ زائد 'something that increases', $n\bar{a}qis$ is which is the training that decreases', $n\bar{a}hit$ is which the training that raises'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of quantity and size* (that involve abstract and state entities) and those of *the passive participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of quantity and size*. For example, *masġūr* مصغور 'small', *mazıd*', 'increased', *manqūş* منقوص 'decreased'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of quantity and size* (that involve abstract and state entities) and those of *the form of exaggeration* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of quantity and size*. For example, *qallāl* is 'someone who decreases something', *katār* نَقُاص 'someone who increases something', *katār* is 'someone who diminishes something'.

Compatibility with the instrumental noun

There is no compatibility between the semantic features of *Verbs of quantity and size* and those of *the instrumental noun*, where the features of the verbs involve abstract and state entities, and those of the instrumental noun involves concrete and event entities. Therefore, it

is not valid semantically to generate an instrumental noun from Verbs of quantity and size.

Compatibility with the qualificative adjective

There is compatibility between the semantic features of *Verbs of quantity and size* (that involve abstract and state entities) and those of *the qualificative adjective* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a qualificative adjective from *Verbs of quantity and size*. For example, *šaḥiḥ* شحيح 'scarce', *ģazīr شحيح* 'copious; plenteous', *kabīr كبير* 'large'.

Compatibility with the locative noun

There is no compatibility between the semantic features of *Verbs of quantity and size* and those of *the locative noun*, where the features (object entity) of the verbs involve abstract entities, and those of the locative noun involves concrete entities. Therefore, it is not valid semantically to generate a locative noun from *Verbs of quantity and size*.

Class 25: Verbs of stability	
The object entity	The situation entity
These verbs refer to concrete entities, expressing	These verbs refer to state entities. They
tangible meanings (of an action, event, or state	describe a state which is permanent or
which is accessible to one or more of the five	will last for a significant time.
senses) that occupy physical space, such as 'to	
remain, stay', 'to live in', and 'to last, continue'.	
~	

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of stability* (that involve concrete and state entities) and those of *the active participle* (that may also involve concrete and state entities). Therefore, it is valid semantically to generate an active participle from *Verbs of stability*. For example, $b\bar{a}qin$ باق 'someone/something that stays', <u>t</u> $\bar{a}bit$ 'someone/something that is stable', $r\bar{a}kid \leq 1$ 'someone/something that is stagnant'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of stability* (that involve concrete and state entities) and those of *the passive participle* (that may also involve concrete and state entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of stability*. For example, *markūn* مركون 'stagnant', *maskūn*' مسكون 'inhabited', *marsūx*' 'solid'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of stability* (that involve concrete and state entities) and those of *the form of exaggeration* (that may also involve concrete and state entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of stability*. For example, jatun, jatun, 'someone who crouches in a place', *rakkād* ($\lambda = 1$), 'someone who stagnates in a place', *labāt*.

Compatibility with the instrumental noun

There is no compatibility between the semantic features of *Verbs of stability* and those of *the instrumental noun*, where the features of the verbs involve concrete and state entities, and those of the instrumental noun involves concrete and event entities. Therefore, it is not valid semantically to generate an instrumental noun from *Verbs of stability*.

Compatibility with the qualificative adjective

There is compatibility between the semantic features of *Verbs of stability* (that involve concrete and state entities) and those of *the qualificative adjective* (that may also involve concrete and state entities). Therefore, it is valid semantically to generate a qualificative adjective from *Verbs of stability*. For example, *rakīn ركين* 'solid; steady', *waţīd* 'firm', *şamid* 'perpetual'.

Compatibility with the locative noun

There is compatibility between the semantic features of *Verbs of stability* (that involve concrete and state entities) and those of *the locative noun* (that may also involve concrete and state entities). Therefore, it is valid semantically to generate a locative noun from *Verbs of stability*. For example, *maqarr* مقرّ 'domicile, flat', *maskan* 'house (residence)', and *mawțin* 'home) country; birthplace'.

Class 26: Verbs of a	creation	
The object entity	The situation entity	
These verbs refer to concrete entities, expressing	These verbs refer to event entities.	
tangible meanings (of an action, event, or state	They involve a change from one state	
which is accessible to one or more of the five	to another.	
senses) that occupy physical space, such as 'to		
form, shape', 'to make, do, perform; to		
manufacture', and 'to create, make, form,		
fashion'.		
Compatibility with the ac	tive participle	
There is compatibility between the semantic feature	ares of Verbs of creation (that involve	
concrete and event entities) and those of the ac	tive participle (that may also involve	
concrete and event entities). Therefore, it is val	id semantically to generate an active	
participle from <i>Verbs of creation</i> . For example, <i>xāliq</i> خالق 'creator', <i>ṣāni °</i> صانع 'maker',		
<i>sā°iġ</i> صائغ 'former'.		
Compatibility with the passive participle		
There is compatibility between the semantic features of Verbs of creation (that involve		
concrete and event entities) and those of the passive participle (that may also involve		
concrete and event entities). Therefore, it is valid semantically to generate a passive		
participle from Verbs of creation. For example, ma	مصنوع fabricated', maskūn معمول 'fabricated'	
'created', <i>maxlūq</i> مخلوق 'creature'.		
Compatibility with the form of exaggeration		
There is compatibility between the semantic features of Verbs of creation (that involve		
concrete and event entities) and those of the form of exaggeration (that may also involve		
concrete and event entities). Therefore, it is valid semantically to generate a form of		

Compatibility with the instrumental noun

There is compatibility between the semantic features of *Verbs of creation* and those of *the instrumental noun*, where both of them involves concrete and event entities. Therefore, it is valid semantically to generate an instrumental noun from *Verbs of creation*. For example, $mibn\bar{a}^{\circ}$ مِصْنَع 'builder', $mij^{\circ}al$ مِجْعَل 'maker', and $misna^{\circ}$ رُحْمَا.

Compatibility with the qualificative adjective

There is no compatibility between the semantic features of *Verbs of creation* and those of *the qualificative adjective*, where the features (situation entity) of the verbs involve event entities, and those of the qualificative adjective involves state entities. Therefore, it is not valid semantically to generate a qualificative adjective from *Verbs of creation*.

Compatibility with the locative noun

Class 27: Verbs of preparing	
The object entity	The situation entity
These verbs refer to concrete entities, expressing	These verbs refer to event entities.
tangible meanings (of an action, event, or state	They involve a change from one state to
which is accessible to one or more of the five	another.
senses) that occupy physical space, such as 'to rub,	
scrub', 'to rinse, wash', 'to cook', and 'to knead'.	

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of preparing* (that involve concrete and event entities) and those of *the active participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate an active participle from

washer'. ظامل cooker', *ġāsil* نطابخ 'baker', *ṭābix*' خابز 'washer'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of preparing* (that involve concrete and event entities) and those of *the passive participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of preparing*. For example, *maxlūt* مخلوط 'admixed', *maxbūz*', *baked'*, *matbūx*' ocoked'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of preparing* (that involve concrete and event entities) and those of *the form of exaggeration* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of preparing*. For example, *xabāz* خبّاز 'baker', *tabbāx*' خبّان 'chef; head cook in a hotel,restaurant,etc', *cajjān*' عجّان 'perineum; kneader'.

Compatibility with the instrumental noun

There is compatibility between the semantic features of *Verbs of preparing* and those of *the instrumental noun*, where both of them involves concrete and event entities. Therefore, it is valid semantically to generate an instrumental noun from *Verbs of preparing*. For example, *xallāṭah* مِعْجَن 'blender', *tabbāxah* طَبّاخة 'cooker', and *mi cjan*' خلاطة 'kneading machine'.

Compatibility with the qualificative adjective

There is no compatibility between the semantic features of *Verbs of preparing* and those of *the qualificative adjective*, where the features (situation entity) of the verbs involve event entities, and those of the qualificative adjective involves state entities. Therefore, it is not valid semantically to generate a qualificative adjective from *Verbs of preparing*.

Compatibility with the locative noun

There is compatibility between the semantic features of *Verbs of preparing* (that involve concrete and event entities) and those of *the locative noun* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a locative noun from *Verbs of preparing*. For example, *matbax* مطبخ 'kitchen', *maxbaz*' مخبز 'bakehouse; bakery', and *ma cjan*'

'a place of kneading'.

Class 28: Verbs of ingesting		
The object entity	The situation entity	
These verbs refer to concrete entities, expressing	These verbs refer to event entities.	
tangible meanings (of an action, event, or state	They involve a change from one state to	
which is accessible to one or more of the five	another.	
senses) that occupy physical space, such as 'to		
drink', 'to chew, masticate', and 'to digest (food)'.		

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of ingesting* (that involve concrete and event entities) and those of *the active participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate an active participle from *Verbs of ingesting*. For example, *akil* (at the semantically to generate an active participle from *Verbs of ingesting*. For example, *kill* (b) (at the semantically to generate an active participle from the semantic feature).

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of ingesting* (that involve concrete and event entities) and those of *the passive participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of ingesting*. For example, $ma^{2}k\bar{u}l$ $d^{2}c$ 'eaten', $mabl\bar{u}c$ 'swollen', $mamd\bar{u}d$ 'masticated'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of ingesting* (that involve concrete and event entities) and those of *the form of exaggeration* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of ingesting*. For example, *šāḥin* شاحن 'shipper', *ḥāmil* ألما 'bearer', $s\bar{a}hib$ ساحب 'puller'.

Compatibility with the instrumental noun

There is compatibility between the semantic features of *Verbs of ingesting* and those of *the instrumental noun*, where both of them involves concrete and event entities. Therefore, it is valid semantically to generate an instrumental noun from *Verbs of ingesting*. For example, raddacaba and raddacab and raddacaba and

Compatibility with the qualificative adjective

There is no compatibility between the semantic features of *Verbs of ingesting* and those of *the qualificative adjective*, where the features (situation entity) of the verbs involve event entities, and those of the qualificative adjective involves state entities. Therefore, it is not valid semantically to generate a qualificative adjective from *Verbs of ingesting*.

Compatibility with the locative noun

There is compatibility between the semantic features of *Verbs of ingesting* (that involve concrete and event entities) and those of *the locative noun* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a locative noun from *Verbs of ingesting*. For example, *mašrab* مشرب 'a place of drinking', *mahḍam* 'a place of digesting (food)', and *ma²kal* أكل 'a place of eating'.

Class 29: Verbs of the five senses		
The object entity	The situation entity	
These verbs refer to concrete entities, expressing	These verbs refer to event entities.	
tangible meanings (of an action, event, or state which	They involve a change from one state	
is accessible to one or more of the five senses) that	to another.	
occupy physical space. They may be technologically		
applied by electronic devices, such as 'to see', 'to		
hear', 'to touch', 'to taste' and 'to smell'.		

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of the five senses* (that involve concrete and event entities) and those of *the active participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate an active participle from

Verbs of the five senses. For example, *bāṣir* باصر 'seer', *sāmi °* سامع 'drinker', *dā^siq* ذائق 'taster'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of the five senses* (that involve concrete and event entities) and those of *the passive participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of the five senses*. For example, *malmūs* ملموس 'touchable', *mar^oiy* 'seen', *mašmūm* 'smelled'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of the five senses* (that involve concrete and event entities) and those of *the form of exaggeration* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of the five senses*. For example, baṣṣār بصّار 'augur; someone who lays eyes on', *lammās* لمّاس 'someone who touches a lot', *dawwāq* نوّاق 'a gourmand person; someone who enjoys good food and drink'.

Compatibility with the instrumental noun

There is partially compatibility between the semantic features of *Verbs of the five senses* and those of *the instrumental noun*, where both of them involves concrete and event entities. Therefore, it is valid semantically to generate some instrumental nouns from *Verbs of the five senses*. For example, مِجْسَ 'tachistoscope', *mijass' مِجْسَ* 'stethoscope', and *sammā c ah م*ِبْصَار 'speaker'.

Compatibility with the qualificative adjective

There is no compatibility between the semantic features of *Verbs of the five senses* and those of *the qualificative adjective*, where the features (situation entity) of the verbs involve event entities, and those of the qualificative adjective involves state entities. Therefore, it is not valid semantically to generate a qualificative adjective from *Verbs of the five senses*.

Compatibility with the locative noun

There is compatibility between the semantic features of Verbs of the five senses (that involve

concrete and event entities) and those of *the locative noun* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a locative noun from *Verbs of the five senses*. For example, $mar^2\bar{a}$ مرأى 'sight; view of landscape', malmas a place of touch or contact', and madaa a (in the mouth)'.

Class 30: Verbs of ruling and government		
The object entity	The situation entity	
These verbs refer to abstract entities, expressing non-	These verbs refer to state entities.	
physical, unseen and intangible meanings (which are	They describe a state which is	
not accessible to any of the five senses) that do not	permanent or will last for a	
occupy physical space, such as 'to order, command',	significant time.	
'to rule, reign, dominate', and 'to head, lead'.		

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of ruling and government* (that involve abstract and state entities) and those of *the active participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate an active participle from *Verbs of ruling and government*. For example, $h\bar{a}kim$ حاكم 'ruler', $q\bar{a}din$ 'judge', $q\bar{a}^{\circ}id \doteq$ 'leader'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of ruling* (that involve abstract and state entities) and those of *the passive participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of ruling*. For example, *ḥakkām* حكّام 'someone who rules strongly', *ammār أ*مّار 'someone who commands a lot', *caddāl* عدّال 'equitable; evenhanded; rightful'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of ruling* (that involve abstract and state entities) and those of *the form of exaggeration* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a form of exaggeration from

'puller'. ماحب bearer', sāḥib' حامل shipper', ḥāmil' شاحن bearer', sāḥib' ساحب 'puller'.

Compatibility with the instrumental noun

There is no compatibility between the semantic features of *Verbs of ruling and government* and those of *the instrumental noun*, where the features of the verbs involve abstract and state entities, and those of the instrumental noun involves concrete and event entities. Therefore, it is not valid semantically to generate an instrumental noun from *Verbs of ruling and government*.

Compatibility with the qualificative adjective

There is compatibility between the semantic features of *Verbs of ruling and government* (that involve abstract and state entities) and those of *the qualificative adjective* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a qualificative adjective from *Verbs of ruling and government*. For example, *camīr* أمير 'prince', $ra \bar{r} \bar{s}$ (the f', *hakīm* حكيم 'sage'.

Compatibility with the locative noun

There is no compatibility between the semantic features of *Verbs of ruling and government* and those of *the locative noun*, where the features (object entity) of the verbs involve abstract entities, and those of the locative noun involves concrete entities. Therefore, it is not valid semantically to generate a locative noun from *Verbs of ruling and government*.

Class 31: Verbs of the development of life	
The object entity	The situation entity
These verbs refer to abstract entities, expressing non-	These verbs refer to state entities.
physical, unseen and intangible meanings (which are	They describe a state which is
not accessible to any of the five senses) that do not	permanent or will last for a significant
occupy physical space, such as 'to give birth', 'to	time.
grow up', 'to age, grow old', and 'to be youthful'.	
Compatibility with the active participle	

There is compatibility between the semantic features of *Verbs of the development of life* (that involve abstract and state entities) and those of *the active participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate an active participle from *Verbs of the development of life*. For example, $r\bar{a}sid$ (adult', $s\bar{a}bb$) (young', $c\bar{a}sis$) 'young', $c\bar{a}sis$) 'liver'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of the development of life* (that involve abstract and state entities) and those of *the passive participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of the development of life*. For example, *mawlūd* مولود 'born', *mablūg*' 'mature', *mafniy* مفنى 'died'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of the development of life* (that involve abstract and state entities) and those of *the form of exaggeration* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of the development of life*. For example, *wallādah* $e^{\frac{1}{2}}$ 'fruitful; producing offspring prolifically', *mawwāt* $e^{\frac{1}{2}}$ 'someone who is nearly to die', *ballāg* 'mature person'.

Compatibility with the instrumental noun

There is no compatibility between the semantic features of *Verbs of the development of life* and those of *the instrumental noun*, where the features of the verbs involve abstract and state entities, and those of the instrumental noun involves concrete and event entities. Therefore, it is not valid semantically to generate an instrumental noun from *Verbs of the development of life*.

Compatibility with the qualificative adjective

There is compatibility between the semantic features of *Verbs of the development of life* (that involve abstract and state entities) and those of *the qualificative adjective* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a qualificative adjective from *Verbs of the development of life*. For example, *walīd velīd*, 'born',

harim هرم 'very aged', fatīs فطيس 'dead; carcass'.

Compatibility with the locative noun

There is no compatibility between the semantic features of *Verbs of ruling and government* and those of *the locative noun*, where the features (object entity) of the verbs involve abstract entities, and those of the locative noun involves concrete entities. Therefore, it is not valid semantically to generate a locative noun from *Verbs of ruling and government*.

Class 32: Verbs of uttering		
The object entity	The situation entity	
These verbs refer to abstract entities, expressing non-	These verbs refer to state entities.	
physical, unseen and intangible meanings (which are	They describe a state which is	
not accessible to the five senses except the hearing)	permanent or will last for a significant	
that do not occupy physical space, such as 'to read,	time.	
recite', 'to narrate, relate, tell', and 'to say, tell; to		
speak'.		

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of uttering* (that involve abstract and state entities) and those of *the active participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate an active participle from *Verbs of uttering*. For example, $r\bar{a}win$ راو 'reciter', $q\bar{a}ri^{2}$ فارئ' speaker', $q\bar{a}ss$ 'narrator'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of uttering* (that involve abstract and state entities) and those of *the passive participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of uttering*. For example, *matluw* متأل 'recited', *mahkiy* 'cited', *maqūl*' 'said'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of Verbs of uttering (that involve abstract

and state entities) and those of *the form of exaggeration* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of uttering*. For example, *qaṣṣāṣ قصّاص* 'storyteller', *xaṭṭāb* 'orator; a person making a speech', *maddāḥ مدّاح* 'commender; person who praises a lot'.

Compatibility with the instrumental noun

There is no compatibility between the semantic features of *Verbs of uttering* and those of *the instrumental noun*, where the features of the verbs involve abstract and state entities, and those of the instrumental noun involves abstract and event entities. Therefore, it is not valid semantically to generate an instrumental noun from *Verbs of uttering*.

Compatibility with the qualificative adjective

There is compatibility between the semantic features of *Verbs of uttering* (that involve concrete and state entities) and those of *the qualificative adjective* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a qualificative adjective from *Verbs of uttering*. For example, *madīḥ* Δc_{LT} 'panegyric', *qarīz* 'eulogy', *sajic*' assonant; rhymed'.

Compatibility with the locative noun

There is no compatibility between the semantic features of *Verbs of uttering* and those of *the locative noun*, where the features (object entity) of the verbs involve abstract entities, and those of the locative noun involves concrete entities. Therefore, it is not valid semantically to generate a locative noun from *Verbs of uttering*.

Class 33: Verbs of accepting		
The object entity	The situation entity	
These verbs refer to abstract entities,	These verbs refer to state entities. They	
expressing non-physical, unseen and intangible	describe a state which is permanent or will	
meanings (which are not accessible to any of	last for a significant time.	
the five senses) that do not occupy physical		
space, such as 'to accept', 'to obey, follow',		

and 'to be or become satisfied (with)'.	

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of accepting* (that involve abstract and state entities) and those of *the active participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate an active participle from *Verbs of accepting*. For example, $q\bar{a}bil$ فابل 'acceptant', $r\bar{a}din$ راض 'satisfier', $t\bar{a}bi$ 'follower'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of accepting* (that involve abstract and state entities) and those of *the passive participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of accepting*. For example, *masmūh* مسموح 'allowed', *maqbūl* accepted', *ma²dūn* 'accepted', *ma²dūn* alicute'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of accepting* (that involve abstract and state entities) and those of *the form of exaggeration* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of accepting*. For example, $qan\bar{u}^{c}$ قنوع 'contented; satisfied', *hāmil* سموح 'permissive', qabbāl شموح 'acceptor'.

Compatibility with the instrumental noun

There is no compatibility between the semantic features of *Verbs of accepting* and those of *the instrumental noun*, where the features of the verbs involve abstract and state entities, and those of the instrumental noun involves concrete and event entities. Therefore, it is not valid semantically to generate an instrumental noun from *Verbs of accepting*.

Compatibility with the qualificative adjective

There is compatibility between the semantic features of *Verbs of accepting* (that involve abstract and state entities) and those of *the qualificative adjective* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a qualificative

adjective from *Verbs of accepting*. For example, *qani^c* قنع 'contented', *samīḥ* سميح 'tolerant, good-hearted', *taba^c* نبغ 'follower'.

Compatibility with the locative noun

There is no compatibility between the semantic features of *Verbs of accepting* and those of *the locative noun*, where the features (object entity) of the verbs involve abstract entities, and those of the locative noun involves concrete entities. Therefore, it is not valid semantically to generate a locative noun from *Verbs of accepting*.

Class 34: Verbs of refusing and disobedience		
The object entity	The situation entity	
These verbs refer to abstract entities,	These verbs refer to state entities. They	
expressing non-physical, unseen and intangible	describe a state which is permanent or will	
meanings (which are not accessible to any of	last for a significant time.	
the five senses) that do not occupy physical		
space, such as 'to accept', 'to obey, follow',		
and 'to be or become satisfied (with)'.		
Compatibility with the active participle		

There is compatibility between the semantic features of *Verbs of refusing and disobedience* (that involve abstract and state entities) and those of *the active participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate an active participle from *Verbs of refusing and disobedience*. For example, $r\bar{a}fid$, رافض 'refuser', $\sigma\bar{a}bin$ 'decliner', $n\bar{a}kir$, ω 'denier'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of refusing* (that involve abstract and state entities) and those of *the passive participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of refusing*. For example, *majhūd* مجود 'denied', *marfūd*', *irefused'*, *mardūd*' مرفوض' 'refused'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of refusing* (that involve abstract and state entities) and those of *the form of exaggeration active participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of refusing*. For example, *jaḥūd* جود 'denier', *raffāḍ* نَبُاذ discarder'.

Compatibility with the instrumental noun

There is no compatibility between the semantic features of *Verbs of refusing and disobedience* and those of *the instrumental noun*, where the features of the verbs involve abstract and state entities, and those of the instrumental noun involves concrete and event entities. Therefore, it is not valid semantically to generate an instrumental noun from *Verbs of refusing and disobedience*.

Compatibility with the qualificative adjective

There is compatibility between the semantic features of *Verbs of refusing and disobedience* (that involve abstract and state entities) and those of *the qualificative adjective* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a qualificative adjective from *Verbs of refusing and disobedience*. For example, *°abiyy* أبي 'disdainful; haughty', *nakīt* نكيت' 'infringer ; renege', *°aṣiyy* 'revolutionary, rebellious'.

Compatibility with the locative noun

There is no compatibility between the semantic features of *Verbs of refusing and disobedience* and those of *the locative noun*, where the features (object entity) of the verbs involve abstract entities, and those of the locative noun involves concrete entities. Therefore, it is not valid semantically to generate a locative noun from *Verbs of refusing and disobedience*.

Class 35: Verbs of preventing and prohibition	
The object entity	The situation entity
These verbs refer to concrete entities, expressing	These verbs refer to event entities.
tangible meanings (of an action, event, or state which	They involve a change from one state

is accessible to one or more of the five senses) that	to another.
occupy physical space, such as 'to veil, cover', 'to	
prevent, hinder, stop', and 'to jail, lock up'.	

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of preventing and prohibition* (that involve concrete and event entities) and those of *the active participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate an active participle from *Verbs of preventing and prohibition*. For example, $r\bar{a}di^{c}$ دادع 'deterrent', $c\bar{a}^{o}iq$ 'barrier', $m\bar{a}ni^{c}$ مانع 'prohibiter'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of preventing and prohibition* (that involve concrete and event entities) and those of *the passive participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of preventing and prohibition*. For example, *ma^osūr* مضور 'inmate', *maḥdūr* محضور 'prohibited', *makbūḥ* محضور 'controlled'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of preventing and prohibition* (that involve concrete and event entities) and those of *the form of exaggeration* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of preventing and prohibition*. For example, *sajjān* 'jailer', *nahhā*' نهاء 'forbidder', and *mannā*' منّاع 'obstacle'.

Compatibility with the instrumental noun

There is compatibility between the semantic features of *Verbs of preventing and prohibition* and those of *the instrumental noun*, where both of them involves concrete and event entities. Therefore, it is valid semantically to generate an instrumental noun from *Verbs of preventing and prohibition*. For example, *mirbat* مِرْبَط 'rope', *mi cyaq' مِعْيَق 'barrier'*, and *mikbah* 'brake'.

Compatibility with the qualificative adjective

There is no compatibility between the semantic features of Verbs of preventing and

prohibition and those of *the qualificative adjective*, where the features (situation entity) of the verbs involve event entities, and those of the qualificative adjective involves state entities. Therefore, it is not valid semantically to generate a qualificative adjective from *Verbs of preventing and prohibition*.

Compatibility with the locative noun

There is compatibility between the semantic features of *Verbs of preventing and prohibition* (that involve concrete and event entities) and those of *the locative noun* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a locative noun from *Verbs of ingesting*. For example, *masjan محبت* 'jail', *mahjar محبت* 'a place of quarantine', and *mahbas محبت* 'concentration camp'.

Class 36: Verbs of occurrence and progressing	
	100
The object entity	The situation entity
These verbs refer to abstract entities, expressing non-	These verbs refer to state entities.
physical unseen and intensible meanings (which are	They describe a state which is
physical, unseen and intangible meanings (which are	They describe a state which is
not accessible to any of the five senses) that do not	permanent or will last for a significant
•	-
occupy physical space, such as 'to become', 'to	time.
nappen, take place, occur, and to be complete.	

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of occurrence and progressing* (that involve abstract and state entities) and those of *the active participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate an active participle from *Verbs of occurrence and progressing*. For example, $h\bar{a}di\underline{t}$ -incident', $b\bar{a}di^{\circ}$ · incident', $b\bar{a}di^{\circ}$ · use 'starter', $w\bar{a}qi^{\circ} \in I^{\circ}$ · occurring'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of occurrence and progressing* (that involve abstract and state entities) and those of *the passive participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a passive

participle from *Verbs of occurrence and progressing*. For example, *mabdū*^o مبدوء 'started', *matmūm* متموم 'precedented'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of occurrence* (that involve abstract and state entities) and those of *the form of exaggeration* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of occurrence*. For example, *sabbāq* سبّاق 'earliest', *šarrā* ' شرّاع 'initiator', *baddā*' بدّاء 'originator'.

Compatibility with the instrumental noun

There is no compatibility between the semantic features of *Verbs of occurrence and progressing* and those of *the instrumental noun*, where the features of the verbs involve abstract and state entities, and those of the instrumental noun involves concrete and event entities. Therefore, it is not valid semantically to generate an instrumental noun from *Verbs of occurrence and progressing*.

Compatibility with the qualificative adjective

There is compatibility between the semantic features of *Verbs of occurrence and progressing* (that involve abstract and state entities) and those of *the qualificative adjective* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a qualificative adjective from *Verbs of occurrence and progressing*. For example, *hadīt* حديث 'modern', *atīl* أثيل 'deep-rooted; firmly established', *qadīm* فديم 'ancient; old'.

Compatibility with the locative noun

There is no compatibility between the semantic features of *Verbs of occurrence and progressing* and those of *the locative noun*, where the features (object entity) of the verbs involve abstract entities, and those of the locative noun involves concrete entities. Therefore, it is not valid semantically to generate a locative noun from *Verbs of occurrence and progressing*.

Class 37: Verb of appearance

The object entity	The situation entity
These verbs refer to abstract entities, expressing non-	These verbs refer to state entities.
physical, unseen and intangible meanings (which are	They describe a state which is
not accessible to any of the five senses) that do not	permanent or will last for a
occupy physical space, such as 'to appear', 'to	significant time.
emerge, rise, show', 'to well, well up', and 'well out'.	

Compatibility with the active participle

There is compatibility between the semantic features of *Verb of appearance* (that involve abstract and state entities) and those of *the active participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate an active participle from *Verb of appearance*. For example, $b\bar{a}^{2}in$ بانن 'clear', $z\bar{a}hir$ 'apparent', $w\bar{a}dih$ واضح 'obvious'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verb of appearance and progressing* (that involve abstract and state entities) and those of *the passive participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a passive participle from *Verb of appearance*. For example, *makšūf* مظهور 'exposed', *mazhūr* مظهور 'appeared', *mašhūd* 'observed'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verb of appearance* (that involve abstract and state entities) and those of *the form of exaggeration* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verb of appearance*. For example, waddah وضماع 'transparent; obvious; evident', sattace 'manifest', nabbace 'itemming'.

Compatibility with the instrumental noun

There is no compatibility between the semantic features of *Verb of appearance* and those of *the instrumental noun*, where the features of the verbs involve abstract and state entities, and those of the instrumental noun involves concrete and event entities. Therefore, it is not valid semantically to generate an instrumental noun from *Verb of appearance*.

Compatibility with the qualificative adjective

There is compatibility between the semantic features of *Verb of appearance* (that involve abstract and state entities) and those of *the qualificative adjective* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a qualificative adjective from *Verb of appearance*. For example, *jahīr* جهير 'stentorian', *zahīr* ظهير 'seeming', *jaliyy* 'clear'.

Compatibility with the locative noun

There is no compatibility between the semantic features of *Verb of appearance and progressing* and those of *the locative noun*, where the features (object entity) of the verbs involve abstract entities, and those of the locative noun involves concrete entities. Therefore, it is not valid semantically to generate a locative noun from *Verb of appearance*.

Class 38: Verbs of disappearing and ending		
The object entity	The situation entity	
These verbs refer to abstract entities, expressing	These verbs refer to state entities. They	
non-physical, unseen and intangible meanings	describe a state which is permanent or will	
(which are not accessible to any of the five	last for a significant time.	
senses) that do not occupy physical space, such		
as 'to disappear, vanish', 'to sink down, fall		
down, collapse', and 'to hide; to be hidden'.		
Connectibility with the active nexticials		

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of disappearing and ending* (that involve abstract and state entities) and those of *the active participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate an active participle from *Verbs of disappearing and ending*. For example, $b\bar{a}^{\circ}id$ بالد 'extinct', $z\bar{a}^{\circ}il$ زائل 'evanescent', $f\bar{a}rig$ فارغ 'empty'.

Compatibility with the passive participle

There is compatibility between the semantic features of Verbs of disappearing and ending

(that involve abstract and state entities) and those of *the passive participle* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of disappearing and ending*. For example, *maxfiy* منفوق 'disappeared', *manfūq* منفوق 'exhausted', *mandūb* منضوب 'depleted'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of disappearing and ending* (that involve abstract and state entities) and those of *the form of exaggeration* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of disappearing and ending*. For example, *affāl أَقَال* 'declining', *xammād أَقَال* 'inactive', *naddāb* 'iexhaustible'.

Compatibility with the instrumental noun

There is no compatibility between the semantic features of *Verbs of disappearing and ending* and those of *the instrumental noun*, where the features of the verbs involve abstract and state entities, and those of the instrumental noun involves concrete and event entities. Therefore, it is not valid semantically to generate an instrumental noun from *Verbs of disappearing and ending*.

Compatibility with the qualificative adjective

There is compatibility between the semantic features of *Verbs of disappearing and ending* (that involve abstract and state entities) and those of *the qualificative adjective* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a qualificative adjective from *Verbs of disappearing and ending*. For example, *kamīn* كمين 'covered', *xafiyy*' فِي 'hidden', *afijl*' evanescent'.

Compatibility with the locative noun

There is no compatibility between the semantic features of *Verbs of disappearing and ending* and those of *the locative noun*, where the features (object entity) of the verbs involve abstract entities, and those of the locative noun involves concrete entities. Therefore, it is not valid semantically to generate a locative noun from *Verbs of disappearing and ending*.
Class 39: Verbs of bodily movements							
The object entity	The situation entity						
These verbs refer to concrete entities, expressing	These verbs refer to event entities. They						
tangible meanings (of an action, event, or state	involve a change from one state to						
which is accessible to one or more of the five	another.						
senses) that occupy physical space, such as 'to							
kneel down', 'to lie down', and 'to sit down'.							

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of bodily movements* (that involve concrete and event entities) and those of *the active participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate an active participle from *Verbs of bodily movements*. For example, *jālis* filts, *rāki*, *rāki*, *vāqif*, *vāqif*, *vāqif*, *vaqif*, *vaqif*, *vaqif*.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of bodily movements* (that *Concrete* and *Event* entities) and those of *the passive participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of bodily movements*. For example, *majlūs* مجلوس 'sit down', *mawqūf* مركوع 'standee', *markū* مركوع 'kneeled (down)'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of bodily movements* (that involve concrete and event entities) and those of *the form of exaggeration* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of bodily movements*. For example, $qa \, c \, c \, \bar{a}d$ is someone who sits down a lot', $waqq\bar{a}f$ وقاف 'someone who stands up a lot', $barr\bar{a}k$ وزاف 'someone who kneels down a lot'.

Compatibility with the instrumental noun

There is compatibility between the semantic features of Verbs of bodily movements and those

of *the instrumental noun*, where both of them involves concrete and event entities. Therefore, it is valid semantically to generate an instrumental noun from *Verbs of bodily movements*. For example, *mijlasah* مِدْهَات 'seat', *minhad* مِذْهَض 'crutch', and *miwța³ah*' footrest'.

Compatibility with the qualificative adjective

There is no compatibility between the semantic features of *Verbs of bodily movements* and those of *the qualificative adjective*, where the features (situation entity) of the verbs involve event entities, and those of the qualificative adjective adjective involves state entities. Therefore, it is not valid semantically to generate a qualificative adjective from *Verbs of bodily movements*.

Compatibility with the locative noun

There is compatibility between the semantic features of *Verbs of bodily movements* (that involve concrete and event entities) and those of *the locative noun* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a locative noun from *Verbs of bodily movements*. For example, *masjid* مسجد 'mosque; a place of kneeling in worship (God)', *majlis* مجلس 'council, seat', and *marqas* مرقص 'ballroom'.

Class 40: Verbs of taking and giving							
The object entity	The situation entity						
These verbs refer to concrete entities, expressing	These verbs refer to event entities. They						
tangible meanings (of an action, event, or state	involve a change from one state to						
which is accessible to one or more of the five	another.						
senses) that occupy physical space, such as 'to							
take, pick up', to bring', and 'to obtain, get'.							

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of taking and giving* (that involve concrete and event entities) and those of *the active participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate an active participle from *Verbs of taking and giving*. For example, ${}^{2}\bar{a}xi\underline{d}$ ideal 'taker', $j\bar{a}lib$ 'bringer', $n\bar{a}{}^{2}il$ 'getter'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of taking and giving* (that *Concrete* and *Event* entities) and those of *the passive participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of taking and giving*. For example, *majlūb* مجلوب 'brought', *maksūb* مكسوب 'gained', *mamnūh* ممنوح 'granted'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of taking and giving* (that involve concrete and event entities) and those of *the form of exaggeration* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of taking and giving*. For example, *wahhāb* وهُاب 'donor; grantor', *axxād* أخّاذ 'taker', *sarrāq* سرّاق 'thief'.

Compatibility with the instrumental noun

There is compatibility between the semantic features of *Verbs of taking and giving* and those of *the instrumental noun*, where both of them involves concrete and event entities. Therefore, it is valid semantically to generate an instrumental noun from *Verbs of taking and giving*. For example, *mirja^c* مِزْجَع 'retrieval', *hassālah*' حصّالة 'moneybox', *mijlab*'.

Compatibility with the qualificative adjective

There is no compatibility between the semantic features of *Verbs of taking and giving* and those of *the qualificative adjective*, where the features (situation entity) of the verbs involve event entities, and those of the qualificative adjective adjective involves state entities. Therefore, it is not valid semantically to generate a qualificative adjective from *Verbs of taking and giving*.

Compatibility with the locative noun

There is compatibility between the semantic features of *Verbs of taking and giving* (that involve concrete and event entities) and those of *the locative noun* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a locative noun from *Verbs of taking and giving*. For example, *marji*^c $\sim \sim \sim$ 'a place of returning', *majlab* 'a place of taking'.

Class 41: Verbs of human sounds						
The object entity	The situation entity					
These verbs refer to concrete entities in terms of	These verbs refer to state entities. They					
they are accessible to one of the five senses where	describe a state which is permanent or					
they can be heard, but they express non-physical,	will last for a significant time.					
unseen and intangible meanings that do not occupy						
physical space, such as 'to cry, weep', 'to snort'						
and 'to sing, chant, warble'.						

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of human sounds* (that involve concrete and state entities) and those of *the active participle* (that may also involve concrete and state entities). Therefore, it is valid semantically to generate an active participle from *Verbs of human sounds*. For example, $b\bar{a}kin$ باك 'crier', $s\bar{a}^{2}ih$ 'yeller', $d\bar{a}hik$ 'aughter'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of human sounds* (that *Concrete* and *State* entities) and those of *the passive participle* (that may also involve concrete and state entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of human sounds*. For example, *maṣrūx* مصروخ 'yelled', *mašdiy* 'sung', *mahtūf mahtūf* 'shouted'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of human sounds* (that involve concrete and state entities) and those of *the form of exaggeration* (that may also involve concrete and state entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of human sounds*. For example, *nawwāh* نوّاح 'someone who wails loudly', *hāmil* 'someone who shouts loudly', *šaxxār* 'عدّار 'someone who snorts loudly'.

Compatibility with the instrumental noun

There is no compatibility between the semantic features of *Verbs of human sounds* and those of *the instrumental noun*, where the features of the verbs involve concrete and state entities, and those of the instrumental noun involves concrete and event entities. Therefore, it is not valid semantically to generate an instrumental noun from *Verbs of human sounds*.

Compatibility with the qualificative adjective

There is compatibility between the semantic features of *Verbs of human sounds* (that involve concrete and state entities) and those of *the qualificative adjective* (that may also involve concrete and state entities). Therefore, it is valid semantically to generate a qualificative adjective from *Verbs of human sounds*. For example, *anīn أنين* 'moan', *safīr صفير* 'whistling', *sarīx صريخ* 'shrieking; yelling'.

Compatibility with the locative noun

There is no compatibility between the semantic features of *Verbs of human sounds* and those of *the locative noun*, where the features (object entity) of the verbs involve entities that can only be heard, but cannot be seen or touched, and those of the locative noun involves concrete entities that can be seen or touched. Therefore, it is not valid semantically to generate a locative noun from *Verbs of human sounds*.

Class 42: Verbs of sounds made by animals						
The object entity	The situation entity					
These verbs refer to concrete entities in terms of	These verbs refer to state entities. They					
they are accessible to one of the five senses	describe a state which is permanent or will					
where they can be heard, but they express non-	last for a significant time.					
physical, unseen and intangible meanings that						
do not occupy physical space, such as 'to roar',						
'to howl, yelp, yowl', and 'to coo'.						

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of sounds made by animals* (that involve concrete and state entities) and those of *the active participle* (that may also involve

concrete and state entities). Therefore, it is valid semantically to generate an active participle from *Verbs of sounds made by animals*. For example, $z\bar{a}^{\circ}ir$ زائر 'lion which roars', $n\bar{a}bih$ نابح 'dog which barks', $n\bar{a}hiq$ ناهق 'donkey which brays'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of sounds made by animals* (that *Concrete* and *State* entities) and those of *the passive participle* (that may also involve concrete and state entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of sounds made by animals*. For example, *maz^oūr* مزؤور 'roared (at)', *manbūh* منبوح 'barked (at)', *manhūq* منبوق 'brayed (at)'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of sounds made by animals* (that involve concrete and state entities) and those of *the form of exaggeration* (that may also involve concrete and state entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of combining and constructing*. For example, $za^{2}\bar{a}r$ زَرَّ a roaring (lion)', *şahhāl* موّاء 'a neighing (horse)', $maww\bar{a}^{2} \circ \bar{a}$ ' a meowing (cat)'.

Compatibility with the instrumental noun

There is no compatibility between the semantic features of *Verbs of sounds made by animals* and those of *the instrumental noun*, where the features of the verbs involve concrete and state entities, and those of the instrumental noun involves concrete and event entities. Therefore, it is not valid semantically to generate an instrumental noun from *Verbs of sounds made by animals*.

Compatibility with the qualificative adjective

There is compatibility between the semantic features of *Verbs of sounds made by animals* (that involve concrete and state entities) and those of *the qualificative adjective* (that may also involve concrete and state entities). Therefore, it is valid semantically to generate a qualificative adjective from *Verbs of sounds made by animals*. For example, *hadīl* هديل 'whoop', *nahīq* نعيق 'braying', *nacīq* نعيق 'cooing'.

Compatibility with the locative noun

There is no compatibility between the semantic features of *Verbs of sounds made by animals* and those of *the locative noun*, where the features (object entity) of the verbs involve entities that can only be heard, but cannot be seen or touched, and those of the locative noun involves concrete entities that can be seen or touched. Therefore, it is not valid semantically to generate a locative noun from *Verbs of sounds made by animals*.

Class 43: Verbs of l	bodily care
The object entity	The situation entity
These verbs refer to concrete entities, expressing	These verbs refer to event entities. They
tangible meanings (of an action, event, or state	involve a change from one state to
which is accessible to one or more of the five	another.
senses) that occupy physical space, such as 'to	
shave, shave off', 'to comb (the hair)', 'to darken	
the (edges of the) eyelids with kohl'.	

Compatibility with the active participle

There is compatibility between the semantic features of *Verbs of bodily care* (that involve concrete and event entities) and those of *the active participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate an active participle from *Verbs of bodily care*. For example, *hāliq* حالق 'shaver', *lābis* 'wearer', *sābiġ* 'dyer'.

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of bodily care by animals* (that *Concrete* and *Event* entities) and those of *the passive participle* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of bodily care*. For example, *maḥlūq* محلوق 'shaved', *maṣbūġ* 'coloured (hair)', *malbūs* ملبوس 'weared'.

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of bodily care* (that involve concrete and event entities) and those of *the form of exaggeration* (that may also involve

concrete and event entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of bodily care*. For example, $hal\bar{u}q$ طوق 'someone who shaves a lot', sabūg خلوق 'someone who dyes (her/his hair) a lot', *labbās* صبوغ 'someone who wears a lot'.

Compatibility with the instrumental noun

There is compatibility between the semantic features of *Verbs of bodily care* and those of *the instrumental noun*, where both of them involves concrete and event entities. Therefore, it is valid semantically to generate an instrumental noun from *Verbs of bodily care*. For example, *mişbag* 'dyer', *mikhalah* مِنْزَع 'arabic kohl bottle', and *minza*' مِنْزَع 'tweezer'.

Compatibility with the qualificative adjective

There is no compatibility between the semantic features of *Verbs of bodily care* and those of *the qualificative adjective*, where the features (situation entity) of the verbs involve event entities, and those of the qualificative adjective involves state entities. Therefore, it is not valid semantically to generate a qualificative adjective from *Verbs of bodily care*.

Compatibility with the locative noun

There is compatibility between the semantic features of *Verbs of bodily care* (that involve concrete and event entities) and those of *the locative noun* (that may also involve concrete and event entities). Therefore, it is valid semantically to generate a locative noun from *Verbs of bodily care*. For example, *malbas* مطبق 'a place of wearing (changing) clothes', *maḥlaq* 'a place of shaving; barbershop', and *maṣbaġ* مصبغ 'a place of dyeing hair'.

Class 44: Verbs of wa	inning
The object entity	The situation entity
These verbs refer to abstract entities, expressing non-	These verbs refer to state entities.
physical, unseen and intangible meanings (which are	They describe a state which is
not accessible to any of the five senses) that do not	permanent or will last for a significant
occupy physical space, such as 'to earn, gain, win,	time.
profit', 'to excel, overtop', and 'to overcome,	
defeat'.	

Compatibility with the active participle

Compatibility with the passive participle

There is compatibility between the semantic features of *Verbs of winning* (that *Concrete* and *State* entities) and those of *the passive participle* (that may also involve concrete and state entities). Therefore, it is valid semantically to generate a passive participle from *Verbs of winning*. For example, maġlūq aśleq assive for the semantic features of th

Compatibility with the form of exaggeration

There is compatibility between the semantic features of *Verbs of winning* (that involve abstract and state entities) and those of *the form of exaggeration* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a form of exaggeration from *Verbs of winning*. For example, $rajj\bar{a}h$ رجّاح 'preponderant; superior', *qahhār* غنّام 'conqueror; victor', *ġannām* غنّام 'gainer'.

Compatibility with the instrumental noun

There is no compatibility between the semantic features of *Verbs of winning* and those of *the instrumental noun*, where the features of the verbs involve abstract and state entities, and those of the instrumental noun involves concrete and event entities. Therefore, it is not valid semantically to generate an instrumental noun from *Verbs of winning*.

Compatibility with the qualificative adjective

There is compatibility between the semantic features of *Verbs of winning* (that involve abstract and state entities) and those of *the qualificative adjective* (that may also involve abstract and state entities). Therefore, it is valid semantically to generate a qualificative adjective from *Verbs of winning*. For example, $fad\bar{l}l$ فنيل 'favorite', $rab\bar{l}h$ ربيح 'winner', $haz\bar{l}m$ هزيم 'defeated'.

Compatibility with the locative noun

There is no compatibility between the semantic features of *Verbs of winning* and those of *the locative noun*, where the features (object entity) of the verbs involve abstract entities, and those of the locative noun involves concrete entities. Therefore, it is not valid semantically to generate a locative noun from *Verbs of winning*.

Chapter Six

Computational Morphology

6.1 Introduction

This chapter concerns the formation of nominal derivatives from a computational linguistic perspective. It discusses the importance of computational morphology and its branches, including morphological analysis and morphological generation. Morphological generation problems are taken into consideration, specifically the overgeneration problem. The chapter then deals with overgeneration in the formation of nominal derivatives, as well as our solution to the overgeneration problem. The computational application of generating nominal derivatives is presented. The chapter ends with an evaluation of the performance and accuracy of our model (see section 2.8 that discusses literature of computational linguistics on Arabic morphology).

6.2 Computational morphology

One subfield of computational linguistics ¹⁵ is computational morphology. This is concerned with the development of a computational application (based on theoretical frameworks) to analyse a word within a given text, such as determining the given word's grammatical category (part of speech, whether it is a verb, noun or adjective), and determining its morphological features (e.g. voice, mood, case, gender, number, person) (Sawalha 2011: 4).

Computational morphology lies at the heart of computational linguistics due to its significant interactions with syntax and semantics. Therefore, applications of computational morphology contribute to the work of other more advanced computational applications. For example, computational morphological applications such as root extraction, stem extraction, and part-of-speech tagging are the basis of higher applications involved in information retrieval, machine translation and speech recognition. In addition,

^{15.} The terms "Computational Linguistics", "Natural Language Processing", and "Language Engineering" are usually used as synonyms (cf. Kiraz 2001: 15). However, "Computational Linguistics" refers to a branch of linguistics that deals with the theories and computational techniques applied to solve linguistic problems from a computational perspective, whereas "Natural Language Processing" and "Language Engineering" refer to a research area with computational linguistics that is limited to the computational implementing (programming) of natural languages by pure computational processes.

computational morphology is affected by orthographic issues due to the variation in the shapes of several letters in Arabic depending on the position in the word (cf. Habash 2010: 65).

Due to the complexity of the nonconcatenative structure of Arabic morphology, computational handling of Arabic morphology has been a challenge to researchers. Arabic morphological analysis and generation have been the focus of attention of researchers over the last two decades. Research on computational morphology has focused in general on morphological analysis rather than morphological generation. This is partly due to the fact that the stage of analyzing and understanding morphological structure comes before the stage of generating further morphological structures, thus morphological generation can be described as a reverse process in terms of the outcome of morphological analysis and can be used as an input for generating another linguistic form. In other words, "the analyzer reads the inflected surface form of each word in a text and provides its lexical form while Generation is the inverse process" (Jayan, Rajeev & Rajendran, 2011: 15). For example, the morphological analysis of the (input) word *kātib* 'Writer' is the active participle of the verb *katab* 'Zir 'to write'. In a reversed way, morphological generation creates the active participle *kātib* 'Zir 'writer' from the verb *katab* 'Cir write'.

Theoretical morphological approaches are reflected in computational works. For example, the autosegmental approach (McCarthy 1979, 1981) has been used in many computational attempts to handle Arabic morphology, specifically in the models written by finite state technology (Beesley 1990, 1996, 1998; Kiraz 1994, 2001). An example of the largest system of Arabic morphology built by finite-state technology is the Xerox Morphological Analysis and Generation System (Beesley 1998). Within this system, using a lexicon and rule compilers, automatic intersection processes are performed by interdigitating of triliteral roots and patterns to generate new forms.

Computational systems for nonconcatenative morphologies have mainly been implemented based on finite-state technology since the 1980s. This is defined as a mathematical input-output model by which computational programs accompanied with sequential logic circuits are designed. Furthermore, the advantage of implementing twolevel morphology (such as root and pattern) within a finite-state transducer makes the morphological system bidirectional; thus it can be used for analysis and generation. The computing of the morphological analysis and generation faces challenges that stem from the nonconcatenative (nonlinear) nature of Arabic morphology. Unlike many European languages that can construct a word by linearly concatenating morphemes (such as *un-break-able*), Arabic relies on mainly nonconcatenative morphological processes in word formation. A stem such as $k\bar{a}tib$ كاتب 'writer' is constructed by interdigitating the three root consonants *k-t-b* with the pattern $C\bar{a}CiC$. After this, prefixes and suffixes can be attached to the stem $k\bar{a}tib$ writer' to become *al-katib-un* 'Liew 'the writers'. Another challenge is the orthographic structure of the Arabic script where Arabic words are typically written without diacritics (short vowels). This lack of diacritics frequently causes morphological and syntactic ambiguity; for example the non-vowelled word z can refer to the nouns *calam* and syntactic ambiguity; for example the non-vowelled word, allow 'to teach', allow 'culim 'it is known', and *calim* 'to know'.

Habash (2010: 67) argues that a number of requirements should be considered in building a computational morphological system (analyser and generator). These are "(1) coverage of the language of interest in terms of both lexical coverage (large scale) and coverage of morphological and orthographic phenomena (robustness); (2) the surface forms are mapped to/from a deep level of representation that abstracts as much as possible over language-specific morphological and orthographic features; (3) full reversibility of the system so it can be used as an analyzer or a generator; (4) usability in a wide range of natural language processing applications such as MT or IR; and finally, (5) availability for the research community".

6.2.1 Morphological analysis

Morphological analysis is concerned with analyzing the components of a given word that include prefixes, stems and suffixes. The analysis includes providing all possible (valid and invalid) morphological solutions of the grammatical categories, as well as providing linguistic features such as person, gender, number and voice (Habash 2004). The morphological analysis applications are largely dedicated to the inflectional level of morphology rather than its derivational level where information related to tense, aspect, person, and number are provided. For example, the morphological analysis of the Arabic word $k\bar{a}tib\bar{u}n$ Δc writers' is the active participle *writer* plus the *-un* plural morpheme ACTIVE PARTICIPLE-writer + Plural- $\bar{u}n$.

The morphological analysis applications involve processes that handle both the function and the form of a given word. The function-based approach provides the morphosyntactic information of the input words, while the form-based process analyzes the input words in terms of their morphemic components (prefix, stem and suffix).

The computational morphological analysis techniques are inspired by accepted theoretical approaches to Arabic morphology, including the root-based and stem-based approaches. Within the root-based approach, a given word is analyzed as the integration of a sequence of root consonants and a pattern in addition to prefixes and suffixes. Within this integration, (usually) three root consonants are inserted in certain slots in a pattern. By contrast, the stem-based approach analyzes a given word as composed of prefix, stem and suffix. These components are recognized by checking them against lists of the prefixes, stems and suffixes stored in lexicons (cf. Attia 2008: 26-27).

The morphological analysis applications perform many tasks. These applications include morphological analyzers, lemmatizers, pattern matching algorithms and stemmers (cf. Sawalha 2011: 6). The morphological analyzers produce all possible analyses of a given word out of context (Beesley 1998; Buckwalter 2002, 2004; Boudlal et al. 2010). The stemmers and root-extractors generally extract the root/stem of a given word by removing the suffixes and prefixes, and then by matching the rest of the word against a list of verbal and nominal patterns (Khoja 1999; Al-Shalabi et al. 2003). The lemmatisers determine the lemma of a given word by considering its context and by recognizing its part of speech,

basic form and dictionary entry (Dichy 2001; Al-Shammari and Lin 2008). The patternmatching algorithms identify the templatic pattern and vocalic melody of a given word (Dichy and Farghaly 2003; Al-Shalabi 2005).

6.2.2 Morphological generation

Morphological generation has received less attention than morphological analysis despite the fact that both experience the same challenges that stem from the rich and complex morphology of Arabic. Morphological generation is concerned with generating all possible word-forms from a given root/stem. It also provides grammatical information regarding their grammatical category and morphological features.

Morphological generation serves at two levels of Arabic morphology. First is generation at the inflectional level, which aims to generate inflected forms of a stem/word. At this level, prefixes and/or suffixes are attached to a stem/word without changing its grammatical category, such as the definite article prefix ${}^{\circ}al$ - ${}^{\circ}l$, the plural form suffix $-\bar{u}n \downarrow_{\mathcal{F}}$, and the feminine form suffix $-ah \leftarrow$ as in ${}^{\circ}al$ - $r\bar{a}bih$, the winner', $r\bar{a}bih\bar{u}n$ $\downarrow_{\mathcal{F}}$, winners', and $r\bar{a}bih$ - $ah \leftarrow_{\mathcal{F}}$ is generate inflected for morphological generation at the inflectional level of morphology. It can generate inflected word forms by concatenating prefixes and suffixes to a stem which is taken as the base form. This system includes a prefix database (299 entries), a suffix database and a stem database (82,158 entries). The concatenation of the prefixes and suffixes to a given stem is governed by morphological compatibility tables.

Cavalli-Sforza, Soudi & Mitamura (2000) have presented a computational model for Arabic morphological generation based on a concatenative strategy. Their model is restricted to building a morphological generator for verbal morphology, based on morphological transformational rules that handle the stem change in different prefix/suffix contexts; thus, the generator is simpler and smaller. They exclude infixation and pattern change from the generation of the inflectional level of morphology as they rely only on the concatenation of morphemes to generate inflected word forms by prefixing and suffixing affixes to a stem. Secondly is generation at the derivational level of morphology, which aims to generate new words from a root/stem. At this level, the new word that is the output of the generation may have a different grammatical category from the input of the generation, which here can be either a root or a stem. For example, $d\bar{a}ris$ clcu, clcu,

6.2.2.1 Undergeneration problem

Undergeneration is a shortcoming that occurs when the morphological generation applications fail to generate a valid word form. Sawalha (2011: 77) describes undergeneration as a "problem [that] happens when the generation cannot generate all possible vocabulary of the language". The undergeneration problem can occur at both derivational and inflectional levels of morphology. The occurrence of this problem at the derivational level is more critical than its occurrence at the inflectional level, as the former leads to a lack of generation of a stem that may represent a whole grammatical category, while the latter leads to a lack of generation of inflected word forms of the same stem. The undergeneration problem may also involve non-derived words that are not governed or controlled by morphological patterns.

The solution to the undergeneration problem must be to increase the coverage of the morphological generation applications. Sawalha (2011: 77) argues that "in theory, any morphological generation program for Arabic will suffer from both over-generation and under-generation problems unless it has been provided with a comprehensive database that contains all the non-generated vocabulary". To enhance this coverage, Sawalha (2011: 307) relies on "using the traditional Arabic lexicons text as a corpus … the corpus contains a large number of words (14,369,570) and word types (2,184,315), and the possibility of finding the different forms of the derived words of a given root".

6.2.2.2 Overgeneration problem

Overgeneration occurs when morphological generation applications generate invalid word forms. These forms are correctly generated according to the word-formation rules of the language, but they are invalid semantically and are not a part of real language. Sawalha (2011: 77) argues that "[t]he over-generation problem results in many lexical entries which are correctly structured but are not part of the real language vocabulary".

The problem of overgeneration may occur at both derivational and inflectional levels of morphology. At the derivational level, the generation application generates an invalid stem that represents a grammatical category, such as an instrumental noun from the verb *karih* that represents a grammatical category, such as an instrumental noun from the verb *karih* 'to hate'. At the inflectional level, the generation applications produce invalid forms of a valid/invalid stem, such as adding the sound plural suffix $-\bar{u}n$ ' $_{\mathcal{L}}$ ' to the stem *şabūr* 'patient'.

A clear example of the overgeneration problem is found in the Xerox Morphological Analysis and Generation System (Beesley 1996, 1998). The system includes some 5,000 roots and 400 patterns, and a word stem is generated by interdigitating a root with a specific pattern. Here, the question is: what criteria have been adopted to apply a pattern to a root? According to Beesley (1996: 89), "[t]here are perhaps 5000 Arabic roots in common usage, and about 400 phonologically distinct patterns, most of which are ambiguous. Each root can legally combine with only a small subset of the phonologically distinct patterns, an average of about seventeen or eighteen, and this decidedly derivational process must be controlled by old-fashioned lexicography".

Beesley clarifies that it is not valid to interdigitate all 5,000 roots with all 400 patterns. He states that the criteria for applying a pattern to a root rely on "old-fashioned lexicography". This requires manual or automatic checking of the validity of all the generated forms to determine if the word is in actual use (newspapers, magazines, all print media and, published dictionaries). However, there are no details provided by Beesley regarding that.

Although Beesley states that only an average of "seventeen or eighteen" patterns can be applied to each root, this does not prevent overgeneration. He also refers (1996: 90, 91) to "the lexicon codings" where "each root [is] coded to restrict the patterns with which it can

in fact co-occur". In fact, what Beesly refers to is not applicable as he has not given any details about it.

In this regard, Attia (2008: 39) argues that there is "overgeneration in word derivation. The distribution of patterns for roots is not even, and although each root was hand-coded in the system to select from among the 400 patterns, the task is understandably tedious and prone to mistakes". In addition Sawalha (2011: 32) points out that "[t]he major disadvantages of Xerox are: overgeneration in word derivation due to uneven distribution of patterns for roots...". It is probable that overgeneration occurs more commonly where computational morphological applications adopt a root-based approach; generation by the stem-based approach is, as I argue later (in the conclusion), theoretically more controlled and targeted.

The overgeneration problem is clearly found in two computational applications for Arabic verb generation, namely Qutrub: Arabic verbs conjugator¹⁶ and Verbix verb conjugator¹⁷. Both generate invalid inflected forms for any input verb/root. Qutrub takes various forms of verbs as the input, and then provides vowelled inflected forms that involve the tense/aspect, person, voice, mood, gender and number. Verbix takes triliteral roots only as the input, and then provides perfect and imperfect forms in the singular, dual and plural cases.

6.3 Overgeneration in the formation of nominal derivatives

In theory, the overgeneration problem occurs in the generation of three types of nominal derivative: the instrumental noun, the qualificative adjective, and the locative noun (see sections 5.2.4, 5.2.5, 5.2.6, 5.3.4, 5.3.5, and 5.3.6). By contrast, the overgeneration problem does not occur in generating the active participle (agent noun), the passive participle, or the form of exaggeration (see sections 5.2.1, 5.2.2, 5.2.3, 5.3.1, 5.3.2, and 5.3.3), because the semantic features of these nominal derivatives allow for broader application.

The generation of the instrumental noun from the following verb classes leads to overgeneration (the next section 6.4 provides the reason):

^{16.} Available online at: http://qutrub.arabeyes.org/.

^{17.} Available online at: http://www.verbix.com/languages/arabic.shtml/.

- 1. Verbs of the state of the body
- 2. Verbs of body parts
- 3. Verbs of diseases
- 4. Verbs of social interaction
- 5. Verbs of emotions
- 6. Verbs of colouring
- 7. Verbs of bodily qualities
- 8. Verbs of locations/places
- 9. Verbs of mental process
- 10. Verbs of financial transactions
- 11. Verbs of desire and request
- 12. Verbs of intention

- 13. Verbs of quantity and size
- 14. Verbs of stability
- 15. Verbs of ruling and government
- 16. Verbs of the development of the life
- 17. Verbs of uttering
- 18. Verbs of accepting
- 19. Verbs of refusing and disobedience
- 20. Verbs of occurrence and progressing
- 21. Verb of appearance
- 22. Verbs of disappearing and ending
- 23. Verbs of human sounds
- 24. Verbs of sounds made by animals Verbs of winning

The generation of the qualificative adjective from the following verb classes leads to overgeneration (the next section 6.4 provides the reason):

- 1. Verbs of motion
- 2. Verbs of swimming
- 3. Verbs of violence and abuse
- 4. Verbs of agriculture
- 5. Verbs of combining and constructing
- 6. Verbs of sending and carrying
- 7. Verbs of separating and disassembling
- 8. Verbs of removing
- 9. Verbs of bending

- 10. Verbs of decorating and transcribing
- 11. Verbs of measure
- 12. Verbs of creation
- 13. Verbs of preparing
- 14. Verbs of ingesting
- 15. Verbs of the five senses
- 16. Verbs of preventing and prohibition
- 17. Verbs of bodily movements
- 18. Verbs of taking and giving
- 19. Verbs of bodily care

The generation of the locative noun from the following verb classes leads to overgeneration (the next section 6.4 provides the reason):

- 1. Verbs of the state of the body
- 2. Verbs of diseases
- 3. Verbs of social interaction
- 4. Verbs of emotions
- 5. Verbs of bodily qualities
- 6. Verbs of mental process
- 7. Verbs of financial transactions
- 8. Verbs of desire and request
- 9. Verbs of intention

- 11. Verbs of ruling and government
- 12. Verbs of the development of the life
- 13. Verbs of uttering
- 14. Verbs of accepting
- 15. Verbs of refusing and disobedience
- 16. Verbs of occurrence and progressing
- 17. Verb of appearance
- 18. Verbs of disappearing and ending
- 19. Verbs of human sounds

10. Verbs of quantity and size

20. Verbs of sounds made by animals21. Verbs of winning

6.4 Our solution to the overgeneration problem of nominal derivatives

This study provides a model to control the word formation of nominal derivatives. Our model handles the components of word formation of the nominal derivatives as inputs and outputs: the inputs are Form I verbs and the outputs are six types of nominal derivative. Form I verbs (the inputs) are classified semantically and syntactically into 44 verb classes (see chapter 4). This model uses the notions of the object entity (concrete and abstract entities) and situation entity (state and event entities) (see section 5.3) in order to tag and describe both inputs and outputs, i.e. the verb classes are tagged to determine the validity or invalidity of deriving the nominal derivatives.

Both the 44 classes of verbs (the inputs) and the nominal derivatives (the outputs) are tagged and described in terms of the object entity as concrete or abstract entities; and in terms of the situation entity as state or event entities (see figure 11 in the conclusion). As stated earlier, this study has determined that the overgeneration problem is found only in three types of nominal derivative: the instrumental noun, the qualificative adjective and the locative noun. The constraints on deriving them are expressed as follows:

- The instrumental noun refers to concrete and event entities, given that its formation requires it to be only generated from a verb that is similarly tagged as concrete and event, such as the verb *ġasal* غسل 'to wash', whose instrumental noun is *ġassālah* 'washing machine'. Therefore, if the instrumental noun is generated from verbs that are not tagged only as concrete and event, it has been overgenerated.
- The qualificative adjective refers to concrete/abstract and state entities, given that its formation requires to be generated from a verb that is similarly tagged as concrete/abstract and state, such as the verb *kabur* کبر 'to be big' whose qualificative adjective is *kabur* کبیر 'big'. Therefore, if a qualificative adjective is generated from verbs that are not tagged as concrete/abstract and state, it has been overgenerated.

The locative noun refers to concrete and state/event entities, given that its formation requires to be generated from a verb that is similarly tagged as concrete and state/event, such as the verb *lacib* (see to play) to play whose locative noun is *malcab* (playground). Therefore, if the qualificative adjective is generated from verbs that are not tagged only as concrete/abstract and state, it has been overgenerated.

There is no overgeneration problem in generating the active participle, the passive participle, and the form of exaggeration, due to they may be tagged for both concrete and abstract entities in terms of the object entity, and may tagged for both state and event in terms of the situation entity, thus, the compatibility between the semantic features of these three nominal derivatives and the semantic features of all the verb classes is achieved. i.e. with some nominals within this class taking concrete entities, and some taking abstract entities: for example, the active participle $d\bar{a}rib$ $\dot{d}rib$ $\dot{d}reamer$ ' refers to a concrete object entity, while the active participle $h\bar{a}lim$ $\dot{d}reamer'$ refers to an abstract object entity, although certain types of nominals within this class taking a state situation entity, and some taking an event situation entity: for example, the active participle $d\bar{a}rib$ $\dot{d}rib$ $\dot{d}reamer'$ refers to a state situation entity.

6.5 The computational application of nominal derivatives¹⁸

This study presents a computational application for retrieving the nominal derivatives of Form I verbs. It includes 980 Form I verbs and 7,374 stems of nominal derivatives. The application was built using the Microsoft Access 2007/2010 database. The application consists of two parts, a retriever and an applicability checker. By inputting a triliteral verb (Form I), the first part executes the querying process to retrieve the required data from the database. The output data can be classified into four types as shown in table 28:

General	1. transliteration			
information	2. the class of verb			
Syntactic	3. imperfective form			
information	4. transitive/intransitive			
Semantic	5. concrete/abstract and state/event			
information	6. English meaning			
Nominal	7. the active participle			
derivatives	8. the passive participle			
	9. the form of exaggeration			
	10. the instrumental noun			
	11. the qualificative adjective			
	12. the locative noun			
	13. the noun of instance			
	14. the <i>mīmī maṣdar</i>			
	15. the noun of manner			

Table 28: The output data of the computational application

The second part of the application involves a programming code to provide further information based on specific data in the database. This information indicates whether or not the instrumental noun, the locative noun and the qualificative adjective outputs are applicable where the semantic compatibility criteria is satisfied. This part of the application seeks to answer three questions:

• Is the instrumental noun applicable?

^{18.} The application was designed by collaboration with Abdullah Alfaifi, PhD reasearcher at the School of Computing / University of Leeds.

- Is the locative noun applicable?
- Is the qualificative noun applicable?

The application answers these questions by conducting an *if* statement which checks four features of the verb query, *concrete*, *abstract*, *state* and *event*. Based on some specific cases of these features, the applicability is shown as follows:

- 1. If the verb has the features *concrete* and *event* then the instrumental noun is applicable.
- 2. If the verb has the features *concrete* and *state/event* then the locative noun is applicable.
- 3. If the verb has the features *concrete/abstract* and *state* then the qualificative noun is applicable.

For instance, the verb *karih* $\geq \leq$ 'hate' is tagged as abstract and state, therefore, the generation of its instrumental noun and locative noun is not applicable, while its qualificative noun is applicable, as shown in the following figure:

Search form							
Search l	DOX Ente	r a verb -or part of it- v	with the di	acritics ^{کرہ}		Search	
#	192	Verb	کرہ	Class	Verbs of	emotions	
Transcript	kariha	Active P	کارہ	Instrumental		instance	ػٙۯۿ؋
Im	يكرّه	Passive P	مكروه	Locative		manner	<u>ڮ</u> ۯۿة
т	i	Exaggeration	کرّاہ	Qualificative	کرهان	mīmī maşdar	مَكره
Concreate	no	Abstract	yes	State	yes	Event	no
Instrumental	is not applica	able					

Figure 9: The interface of the computational application

Instrumental is not applicable Locative is not applicable

Qualificative is applicable

The application and its databases will be particularly beneficial in many respects. The database can be reused in building or improving other computational morphological applications, as it includes classified (tagged) data at various levels of analysis. Morphologically, the data can be reused as morphological lexicons for a morphological analyzer and generator including 980 Form I verbs and 7374 stems representing six types of nominal derivative. For a further level of generation, the application can be expanded to

involve inflectional morphology to generate all inflicted words of nominal derivative stems. By adding valid morphological affixes to each stem, inflected plural forms and feminine forms can be generated.

Semantically, wordNet applications can benefit from the application's data of 980 verbs classified into 44 semantic classes, the tagging of the verbs as concrete/abstract and state/event, an English glossary of the 980 verbs, and the lists of semantically related stems. Syntactically, the application's data classifies verbs into transitive and intransitive. The semantic information can be useful in practice for computational linguistic analysis purposes in future research. Furthermore, the application can be used for teaching and learning the Arabic word formation system.

6.6 Evaluation

This section aims to evaluate the performance of the model that we have developed to generate valid nominal derivatives from Form I verbs. Specifically, the evaluation is dedicated to three types of nominal derivative: the instrumental noun, the qualificative adjective, and the locative noun. These three types suffer from the overgeneration problem, so evaluation aims to measure the accuracy of predicting valid nominal derivatives in order to determine the extent to which overgeneration has been avoided.

The sample that has undergone evaluation comprises 150 verbs and three types of their nominal derivatives (potentially 450): the instrumental noun, the qualificative adjective, and the locative noun. These 150 verbs were randomly selected from all 44 verb classes. Only one or two of the three nominal derivatives are predicted to be generated from each verb. If the generation of a nominal derivative is predicted, then it is represented as YES, while if it is not predicated it is represented as NO. For each verb's derivative, YES or NO is predicted according to our model's criteria, as follows:

- The prediction of the instrumental noun is represented as YES if the verb is concrete and event. Otherwise, its prediction is represented as NO.
- The prediction of the qualificative adjective is represented as YES if the verb is concrete/abstract and state. Otherwise, its prediction is represented as NO.
- The prediction of the locative noun is represented as YES if the verb i concrete and state/event. Otherwise, its prediction is represented as NO.

The prediction of 450 nominal derivatives (YES or NO) has been checked manually by searching three resources:

- Sawalha's Arabic root-meaning search¹⁹, representing traditional dictionaries.
- Arabdict²⁰, representing modern dictionaries.
- arTenTen12 corpus²¹ (Sketchengine), representing the largest Arabic corpus.

^{19.} Available online at: http://www.comp.leeds.ac.uk/cgi-bin/scmss/arabic_roots.py

^{20.} Available online at: http://www.arabdict.com/ar/

^{21.} It contains 5,794,161,583 words, and is available online at:

 $https://the.sketchengine.co.uk/bonito/run.cgi/first_form?corpname=preloaded/artenten12;$

At this stage, binary data is held in two lists. The first list refers to whether the nominal derivative is generated or not; YES means that generation of the predicted nominal derivative is valid, and NO means that its generation is not valid. The second list refers to whether the nominal derivative is found in one of the three mentioned resources: YES means that the nominal derivative is real and found, while NO means that it is not real and not found. Table 29 shows the evaluation data:

#	Verb	Concreate	Abstract	State	Event	Derivitives	Generated	Found	Cases
1	أرق	NO	YES	YES	NO	Instrumental	NO	NO	TN
	°ariq					Locative	NO	NO	TN
						Qualificative	YES	YES	ТР
2	أنَّ	NO	YES	YES	NO	Instrumental	NO	NO	TN
	°ann					Locative	NO	NO	TN
						Qualificative	YES	YES	ТР
3	بتر	YES	NO	NO	YES	Instrumental	YES	NO	FP
	batara					Locative	YES	NO	FP
						Qualificative	NO	NO	TN
4	حكم	NO	YES	YES	NO	Instrumental	NO	NO	TN
	<u></u> hakam					Locative	NO	NO	TN
						Qualificative	YES	YES	ТР
5	بخل	NO	YES	YES	NO	Instrumental	NO	NO	TN
	baxula					Locative	NO	NO	TN
						Qualificative	YES	YES	TP

Table 29: The evaluation data

Confusion matrix formulas are used to measure quantitatively the accuracy of predicting derivatives correctly as well as the degree of overgeneration. Confusion matrix elements are represented in table 30:

Table 30: Confusion matrix elements

	PREDICTED CLASS					
		Class=Yes	Class=No			
ACTUAL	Class=Yes	True Positive (TP)	False Negative (FN)			
CLASS	Class=No	False Positive (FP)	True Negative (TN)			

• True Positive (TP): represents the case in which the generated nominal derivative is predicted correctly, where it is found and used in real language.

- False Positive (FP): represents the case in which the generated nominal derivative is predicted correctly, but is not found and used in real language.
- True Negative (TN): represents the case in which the generated nominal derivative is predicted to be not generated, and is not found and used in real language.
 The values of the evaluation data are 213 cases of true positive, 16 cases of false positive, and 221 cases of the true negative. These values are represented in Table 31:

	PREDICTED CLASS					
		Class=Yes	Class=No			
ACTUAL	Class=Yes	213	0			
CLASS	Class=No	16	221			

Table 31: The values of the confusion matrix elements

Three confusion matrix formulas can be applied to analyze the result of the evaluation:

(1) Precision criteria formula, defined as follows:

$$Precision = \frac{\text{TP}}{\text{TP} + \text{FP}}$$

Precision: 213 / (213 + 16) = 93%

(2) Specificity criteria formula defined as follows:

Specificity =
$$\frac{\text{TN}}{\text{TN} + \text{FP}}$$

Specificity: 221 / (221 + 16) = 93.2%

(3) Accuracy criteria formula defined as follows:

$$Accuracy = \frac{TP + TN}{TP + TN + FP + FN}$$

Accuracy: (213+221) / (213+221+16+0) = 96.4%

According to the evaluation results, the following facts are inferred:

• The percentage of overgeneration in our model is 3.55%. In contrast, the percentage of the potential overgeneration, in the case of generating all the nominal derivatives from all the verbs, is 52.6%, as follows:

(TN+FP) * 100 / (TP+TN+FP) (221+16)=237 * 100 / (213+221+16) = 52.6%

• The percentage of our total accuracy is 96.4%. This percentage includes two cases: first, the nominal derivatives that are correctly predicted to be generated, and are found and used in real language; secondly, the nominal derivatives that are correctly predicted to be not generated, and are not found and used in real language.

With respect to the errors analysis, 3.55%, these errors represent the overgenerated cases. The nominal derivatives in these cases are generated according to semantic and morphological rules, but they are not found or used in real language. The absence of a word in the real language does not mean that it is necessarily invalid. This study considers the nominal derivatives theoretically valid by analogy with other semantically similar forms that share the same semantic features in terms of concrete, abstract, state and event.

Here are some of the overgenerated nominal derivatives and justification for their validity:

- The instrumental noun *mibtar* مبتر generated from the verb *batar* بتر 'to cut off, sever, lop off'. This verb is tagged as concrete and event, thus its instrumental noun is valid by analogy with other instrumental nouns such as *miqaṣṣ* 'scissors' and *qaṭṭācah* فطّاعة 'cutter'. These instrumental nouns share a similar meaning and are generated from verbs that are also tagged as concrete and event.
- The instrumental noun *mimzij* معزج generated from the verb *mazaj* vito mix, mingle, blend'. This verb is tagged as concrete and event, thus its instrumental noun is valid by analogy with other instrumental nouns such as *xallāțah* vitide vitide
- The locative noun *mabdar* مبذر generated from the verb *badar* بذر *to sow, seed, strew*'. This verb is tagged as concrete and event, generating a valid locative noun by analogy with other locative nouns such as *mazra*^cah مزرعة farm' and *maštal*

'plantation' that are generated from semantically related verbs, and are tagged as concrete and event.

Conclusion

This study consisted of six chapters and a conclusion. The first chapter described the linguistic tradition of the Arabic language and introducing the research topic, discussing the research problem, questions, motivation and contribution, and the thesis structure. Chapter two presented the theoretical background and research context, and Chapter three the research methodology. Chapter four provided a classification of Arabic verbs, while Chapter five presented the morphological, prosodic and semantic analysis of the nominal derivatives. Finally, Chapter six dealt with the formation of the nominal derivatives from a computational linguistic perspective.

This study has three main novel contributions to knowledge that would be particularly beneficial to morphologists, prosodic phonologists, and computational linguists:

- The study has provided an unprecedented Arabic verb classification on semantic and syntactic basis. The significance of this classification originates from: benefiting from the semantic features and the syntactic behaviour of a given verb class in order to control the formation of its valid nominal derivatives; providing tagged linguistic data, semantically and syntactically, that can be used in other future research from different perspectives.
- The study has presented a prosodic morphological account of analyzing and describing the word formation processes and prosodic structure of nominal derivatives, within which the input of word formation is Form I verb while the output stem is a nominal derivative. The study has provided three pieces of practical evidence in favour of the stem-based approach.
- The study has developed a semantic model to avoid the overgeneration problem that is found in computational linguistic applications. This model has determined the compatibilities (harmony) between the semantic features of verb classes and the semantic features of nominal derivatives in order to determine the valid forms.

The analysis began with the establishment of an Arabic verb classification in which 980 Form I verbs were divided into 44 classes (see Chapter four). Verbs were allocated a class according to their semantic features and syntactic behaviour. Semantically, the verbs in each class share related meanings and semantic functions. In addition, semantic relations such as synonyms, antonyms, polysemy, and hyponym were taken into consideration. Syntactically, the verbs in each class share syntactic behaviour in terms of their transitivity and syntactic frames.

At the morphological and prosodic level, the study has given an account of analyzing and describing the formation of nominal derivatives that are categorized as: the active participle (agent noun), the passive participle, the form of exaggeration, the instrumental noun, the qualificative adjective, and the locative noun. Morphologically, the word formation of nominal derivatives was analyzed and described using the stem-based approach where the input stem is a Form I verb and the output stem is a nominal derivative. Prosodically, the analysis describes the word formation processes and prosodic structures of the nominal derivatives. The word formation processes in the formation of the nominal derivatives involve a series of templatic changes, including: reduplication of initial/final moras, (mora) μ -prefixation, (mora) μ -suffixation, disassociation of vowels, mora association, and finally melodic overwriting.

The prosodic representation of templatic changes that occur on the input stem, were exemplified by diagrams (prosodic trees) showing the stages of derivation of the nominal derivatives from their Form I verb stems. In addition, the prosodic structure of the nominal derivatives were analyzed within these levels: the syllabic, moraic and, foot levels. The final consonant in the prosodic templates of the nominal derivatives can be either extrametrical or extrasyllabic. It is extrametrical if the final syllable of the nominal derivative is *CVC*, while it is considered extrasyllabic if the final syllable is *CVVC*.

This study concludes that it is both unnecessary and insufficient to consider the root(-based approach) in the formation of nominal derivatives. The formation of nominal derivatives depends completely on the verbal stem rather than the root. The instrumental noun, the locative noun, and the qualificative adjective cannot be derived from the root, since stem information is required in their derivation. Here, the study provides three pieces of evidence in favour of the stem-based approach:

First, transitivity (transitive and intransitive) features are essential to determine whether a nominal derivative is valid to be derived or not in two cases: (i) The instrumental noun is only derived from a transitive verb (Form I stem), such as the verb *kanas* كنس 'to sweep, broom' from which the instrumental noun *miknasah* مكنسة' broom' is derived. (ii) The qualificative adjective is only derived from an intransitive verb (Form I stem), such as the verb *kanas* 'broom' is derived. (ii) The qualificative adjective is only derived from an intransitive verb (Form I stem), such as the verb *sacid* 'use 'to be happy' from which the qualificative adjective *sacīd*' 'use 'happy, glad' is derived.

Second, the vocalic melody of the stem (Form I verb) is vital in deriving the locative noun. The default template of the locative noun is $maC_1C_2aC_3$, such as $mal^cab \rightarrow playground'$ that is derived from the verb $la^cib \rightarrow play'$. However, the locative noun takes the template $maC_1C_2iC_3$ in case the imperfective stem has the vowel -i- (after the second consonant), as in *jalas – yajlis* ($C_1aC_2aC_3 - yaC_1C_2iC_3$) 'to sit' that from which the locative noun *majlis* $\rightarrow production (C_1aC_2aC_3 - yaC_1C_2iC_3)$ of the transitivity of a Form I verb may be reflected in the vocalic melody. Where the vocalic melody of the Form I verb is /u/ in the perfective and imperfective, then the verb is predicted to be intransitive. This is useful in the formation of the instrumental noun where its verb should be transitive.

Third, In Arabic, there are fifteen verb forms, of which ten are common in Modern Standard Arabic. Each of these verb forms has its specific nominal derivatives from the other verb forms. For instance, the root *k*-*t*-*b* is associated with eight verbal stems that have eight different active and passive participles as shown in Table 32, and since no single root is associated with all verbal forms, the root \underline{d} -*k*-*r* for Form V and *s*-*w*-*d* for Form IX. Therefore, the root cannot be taken into consideration in deriving the nominal derivatives.

Root	Verbal stem		Active	Template	Passive	Template
			Participle	Participle		_
k-t-b	Ι	katab	kātib	$C_1 \bar{a} C_2 i C_3$	maktūb	$maC_1C_2\bar{u}C_3$
k-t-b	II	kattab	mukattib	$muC_1aC_2C_2iC_3$	mukattab	$muC_1aC_2C_2aC_3$
k-t-b	III	kātab	mukātib	$muC_1\bar{a}C_2iC_3$	mukātab	$muC_1\bar{a}C_2aC_3$
k-t-b	IV	°aktab	mu²aktib	$mu^{\circ}aC_{1}C_{2}iC_{3}$	mu²aktab	$mu^{\circ}aC_{1}C_{2}aC_{3}$
₫-k-r	V	ta <u>d</u> akkar	muta <u>d</u> akkir	$mutaC_1aC_2C_2aC_3$	muta <u>d</u> akkar	$mutaC_1aC_2C_2aC_3$
k-t-b	VI	takātab	mutakātib	$mutaC_1\bar{a}C_2iC_3$	mutakātab	$mutaC_1\bar{a}C_2a C_3$
k-t-b	VII	°inkatab	munkatib	mun $C_1 a C_2 i C_3$	munkatab	$munC_1aC_2aC_3$
k-t-b	VIII	°iktatab	muktatib	$muC_1C_2aC_2iC_3$	muktatab	$muC_1C_2aC_2aC_3$
s-w-d	IX	°iswadd	muswadid	$muC_1C_2aC_3iC_3$	muswadad	$muC_1C_2aC_3aC_3$
k-t-b	Χ	°istaktab	mustaktib	$mustaC_1C_2iC_3$	mustaktab	$mustaC_1C_2aC_3$

Table 32: The nominal derivatives of the verbal forms

As shown in the above table, the active participle and passive participle differ according to their verbal form stems. Each of the ten mentioned verbal forms has its unique active/passive participle. The Form I stem is the input from which Forms II, III, IV, VII, VIII, IX, and X are derived, while Form V is derived from Form II, and Form VI from Form III, as shown in Figure 10.



Figure 10: Two stages of derivation

This study reaches the conclusion that there are two stages of derivation, as illustrated in Figure 10. First is the derivation from the root, i.e. deriving the Form I stem from the root. At this stage, the root-based approach can be utilized. Second is the derivation from the Form I stem, which is considered the base form, from which verbal derivatives and nominal derivatives are derived, with emphasis on the fact that each verbal derivative has its unique nominal derivatives.

The Form I stem has six different templates (see Table 33), therefore there three Form I stems can be derived from the same root. For example, from the root *d-r-b* three Form I verb stems are derived: *darab* ن منرب یضرب 'to hit', *darib* ' *فنرب* یضرب 'to be corrupted', and *darub* ' *ضرب* یضرب 'to be strong'. The verb *darab* ' *ضرب* یضرب 'to hit' is a transitive verb from which an instrumental noun can be derived, while the verbs *darib* ' *darub*

approach causes an overgeneration problem if there are two or more Form I stems of the same root.

Form I	Perfect template	Imperfect template
'to go' ذهَب <u>d</u> ahab	$C_1 a C_2 a C_3$	$yaC_1C_2aC_3$
to hit' ضرَب <i>darab</i>	$C_1 a C_2 a C_3$	$yaC_1C_2iC_3$
'to write' کتّب katab	$C_1 a C_2 a C_3$	$yaC_1C_2uC_3$
fariḥ فرح 'to be happy'	$C_1 a C_2 i C_3$	$yaC_1C_2aC_3$
<i>hasib حسِب</i> 'to think'	$C_1 a C_2 i C_3$	$yaC_1C_2iC_3$
'to be noble' شرُف <i>šaruf</i>	$C_1 a C_2 u C_3$	$yaC_1C_2uC_3$

Table 33: Form I stem templates

The current study concludes that only four or five valid nominal derivatives can be derived from a given verb, not all six types. The derivation of an invalid nominal derivative causes the overgeneration problem that is involved in computational linguistic applications. The current study determines that overgeneration occurs in three types of nominal derivative: the instrumental noun, the qualificative adjective and the locative noun.

There is no overgeneration problem in deriving the active participle, the passive participle or the form of exaggeration, as they may be tagged for either concrete or abstract entities in terms of the object entity, and may be tagged for either state or event in terms of the situation entity; thus, compatibility between the semantic features of these three nominal derivatives and the semantic features of all the verb classes is achieved. That is, some nominals within this class take concrete entities, and some take abstract entities: for example, the active participle $d\bar{a}rib$ -1 (dreamer' refers to a concrete object entity, while the active participle $h\bar{a}lim$ -1 (dreamer' refers to an abstract object entity. Similarly, certain types of nominal derivative may be tagged for state and event in terms of the situation entity, with some nominals within this class taking a state situation entity, and some taking an event situation entity. For example, the active participle $d\bar{a}rib$ -1-1 (dreamer' refers to a state situation entity, while the active participle to a state situation entity. The study considers that reliance on the root-based approach instead of on the stem-based approach may cause overgeneration if there are two or more Form I stems of the same root. For example, the root *b-t-r* ب ت ر has two Form I verb stems²²: *batar* $C_1aC_2aC_3$ / $yaC_1C_2aC_3$ vito cut off' (transitive) and *batir* $C_1aC_2iC_3$ / $yaC_1C_2iC_3$ vito be amputated' (intransitive). From the former stem only the instrumental noun and the locative noun can be derived, while from the latter only the qualificative adjective can be derived. Therefore, reliance on the stem-based approach allows for syntactic features (transitive and intransitive) that govern the formation of nominal derivatives, namely the instrumental noun and qualificative adjective.

To avoid the overgeneration problem, the current study developed its own model, based on: classifying 980 Form I verbs semantically and syntactically into 44 classes; and determining the compatibility between the semantic features of the verbs of each of the 44 classes and the semantic features of each type of nominal derivative. For each class of verbs, the compatibility with each type of the six nominal derivatives is determined in terms of object entities (concrete and abstract) and situation entities (state and event). There is compatibility between the semantic features of the active participle, the passive participle, and the form of exaggeration on one hand, and the semantic features of the 44 verb classes (comprising all 980 verbs) on the other. Therefore, no overgeneration occurs in these three nominal derivatives.

By contrast, overgeneration may occur in deriving the instrumental noun, the qualificative adjective, and the locative noun from some verb classes. That is because there is no compatibility between the semantic features of these three nominal derivatives and the semantic features of some verb classes. In Figure 11, the 44 verb classes are categorized into three groups according to their semantic features. Figure 11 clarifies and determines the compatibilities between the verb classes on one hand, and the instrumental noun, qualificative adjective and locative noun on the other.

^{22.} See Table 33 that icludes Form I stem templates.


Figure 11: Nominal derivatives compatibilities

The current study characterizes: the instrumental noun as having the semantic features concrete and event, which requires derivation only from verbs that have the same features (concrete and event); the locative noun as having the semantic features concrete and state/event, which requires derivation only from verbs that have the same features (concrete and state/event); and the qualificative adjective as having the semantic features concrete/abstract and state, which requires derivation only from verbs that have the same features (concrete/abstract and state, which requires derivation only from verbs that have the same features (concrete/abstract and state).

Figure 12 shows the proportions of the total number of valid nominal derivatives that can be derived from the 980 verbs in the 44 classes: 409 valid derived instrumental nouns, 492 valid derived locative nouns, and 571 valid derived qualificative adjectives.



Figure 12: The numbers of valid nominal derivatives

The current study concludes that three kinds of nominal derivatives can be found in Arabic: (i) nominal derivatives that are semantically and grammatically valid and in common use, such as *kātib* 'writer' (active participle), *maktūb* 'message' (passive participle), *maktab* 'office' (noun of place); (ii) nominal derivatives that are semantically and grammatically valid but not in common use, such as the locative noun, and instrumental nouns derived from verbs of combining and constructing; and (iii) nominal derivatives that are grammatically valid, but are not valid semantically, and are not in common use, such as the locative noun, and request and verbs of intention.

Recommendations

Recommendations for future study include improving the rules for deriving nominal derivatives that have been examined and implemented in the form of generation algorithms for building a morphological generator. The database of that generator can be expanded to include a larger number of verbs as inputs and other derivatives such as the verbal forms as outputs. In addition to generation at the derivational level, the inflectional level could be taken into consideration to generate inflected words from the output stem, such as a plural form, a feminine form, and a definite form.

In view of the shortcomings in the content of traditional and modern dictionaries, where not all valid words are found, the study suggests development of an online (updatable) dictionary of nominal derivatives. In addition, most Arabic dictionaries do not normally indicate the grammatical category of their lexical entries. An online dictionary could provide default definitions of the nominal derivatives, combining two parts: first, a functional semantic meaning of the nominal derivative, and secondly, a basic meaning of (the base form) Form I verb presented in dictionaries. The derivatives could be linked with the Modern Standard Arabic corpus to indicate their usage.

References

- Al-Anbāri, A. (1961). Al-³inṣāf fī masā³il al-xilāf. Cairo: Al-Maktaba at-Tijāriyyah al-Kubrā.
- Al-Ba^calbaki, R. (1995). *Al-Mawrid: English-Arabic, Arabic-English dictionary*. Beirut: Dar al^cilm li-l-Malāyīn.
- Al-Halawāni, M. (1978). Al-Wādih fī n-nahw wa-ṣ-ṣarf. Damascus: Dar al-Maºmūn li-t-Turāt.
- Al-Hamalāwi, A. (1957). Šadā al-^carf fī fann aṣ-ṣarf. Cairo: Mustafa al-Bābī.
- Al-Istarabādi, R. (1983). Al-Wāfiyah fī šarķ al-kāfiyah. Masqat: Wizārat al-Turāt al-Qawmı wa-t-Taqāfah.
- Al-Jurjāni, A. (1984). Dalā^oil Al-^oi^cjāz. ed. Shakir, M. Cairo: Maktabat al-Khanji.
- Al-Khalīl, A. (1967). *Mu ^cjam al- ^cayn*. ed. Abdallah Darwīsh, A. Baghdad: Maṭba^cat al-^c ānī.
- Al-Rājhi, A. (1973). At-Tatbīq as-sarfi. Beirut: Dār al-Nahdah al- ^c Arabiyyah.
- Al-Sayyid, A. (1998). Fi- n-naḥw al- carabi. Dubai: Dār al-Qalam.
- Al-Shalabi, R., & Evens, M. (1998). A computational morphology system for Arabic. In Proceedings of the Workshop on Computational Approaches to Semitic Languages COLING-ACL 98. 66-72.
- Al-Shalabi, R., Kanaan, G. & Al-Serhan, H. (2003). New approach for extracting Arabic roots. In ACIT '2003: Proceedings of the 2003 Arab Conference on Information Technology, Alexandria. 42-59.
- Al-Shammari, E. and Lin, J. (2008). A novel Arabic lemmatization algorithm. AND '08: Proceedings of the Second Workshop on Analytics for Noisy Unstructured Text Data, Singapore. 113-118.

- Altantawy, M., Habash, N., Rambow, O., & Saleh, I. (2010). Morphological analysis and generation of Arabic nouns: A morphemic functional approach. In *Proceedings of the International Conference on Language Resources and Evaluation, LREC.* Valletta, Malta. 851-858.
- Attia, M. (2008). Handling Arabic morphological and syntactic ambiguity within the LFG framework with a view to machine translation. PhD Thesis. Manchester: University of Manchester.
- Az-Zajjāji, A. (1984). Kitāb al-jumal fi-n-naļw. Beirut: Mu°assasat al-Risālah.
- Bat-El, O. (1994). Stem modification and cluster transfer in Mmodern Hebrew. *Natural Language & Linguistic Theory*, 12(4), 571-596.
- Bat-El, O. (2001). In search for the roots of the C-root: The essence of Semitic morphology. Workshop on Root and Template Morphology. Los Angeles: University of South California.
- Beesley, K. (1990). Finite-state descriptions of Arabic morphology. In *Proceedings of the* 2nd Cambridge Conference on Bilingual Computing in Arabic and English.
 Cambridge University: Literary and Linguistic Computing Centre. No pagination.
- Beesley, K. (1996). Arabic finite-state morphological analysis and generation. Proceedings of the 16th Conference on Computational Linguistics, Vol 1. Copenhagen: Association for Computational Linguistics, 89-94.
- Beesley, K. (2001). Finite-state morphological analysis and generation of Arabic at Xerox research: status and plans in 2001. In *The ACL 2001 Workshop on Arabic Language Processing: Status and Prospects*. Toulouse: Association for Computational Linguistics, 1-8.
- Beesley, K. R. (1998). Arabic morphology using only finite-state operations. In Proceedings of the Workshop on Computational Approaches to Semitic Languages. Montreal: Association for Computational Linguistics. 50-57.
- Benmamoun, E. (1999). Arabic morphology: The central role of the imperfective. *Lingua* 108(2-3), 175-201.

- Boudlal, A., Lakhouaja, A., Mazroui, A., Meziane, A., Bebah, M. O. A. O. & M.Shoul. (2010). Alkhalil morpho sys: A morphosyntactic analysis system for Arabic texts. In *Proceedings of the International Arab Conference on Information Technology*, ACIT, Benghazi, Libya, December 14-15.
- Buckwalter, T. (2002). Buckwalter Arabic morphological analyzer version 1.0. Linguistic Data Consortium, catalog number LDC2002L49 and ISBN 1-58563-257-0.
- Buckwalter, T. (2004). Buckwalter Arabic morphological analyzer version 2.0. Linguistic Data Consortium, catalog number LDC2004L02 and ISBN 1-58563-324-0.
- Cavalli-Sforza, V., Soudi, A., & Mitamura, T. (2000). Arabic morphology generation using a concatenative strategy. In *Proceedings of the 1st North American chapter* of the Association for Computational Linguistics conference. Stroudsburg: Association for Computational Linguistics, 86-93.
- Chomsky, N., & Halle, M. (1968). *The sound pattern of English*. New York: Harper & Row.
- Chon, M., & Arzt, D. E. (2005). Walking while Muslim. Law and Contemporary Problems, 68(2), 215-254.
- Crystal, D. (2008). Dictionary of linguistics and phonetics. Oxford: Wiley-Blackwell.
- Čulo, O., Erk, K., Padó, S., & Im Walde, S. S. (2008). Comparing and combining semantic verb classifications. *Language Resources and Evaluation*, 42(3), 265-291.
- Daif, S. (1983). Al-Madāris al-Nahwiyya, Cairo: Dār al-Ma cārif.
- Darden, B. (1992). The Cairene Arabic verb without form classes. The Joy of Grammar: A Festschrift in Honor of James D. McCawley. ed. Bretari, D., et.al. Amsterdam: John Benjamins. 11-24.
- Darwish, K. (2002). Building a shallow Arabic morphological analyzer in one day. In Proceedings of the ACL-02 Workshop on Computational Approaches to Semitic Languages. Stroudsburg: Association for Computational Linguistics, 1-8.

- Davis, S. & B. A. Zawaydeh (2001). Arabic hypocoristics and the status of the consonantal root. *Linguistic Inquiry*, 32(3): 512-520.
- Dichy, J. & A. Farghaly (2007). Grammar-lexis relations in the computational morphology of Arabic. Arabic Computational Morphology. ed. Soudi, A., Bosch, A., & Neumann, G. Amsterdam: Springer, 115-140.
- Dichy, J. (2001). On lemmatization in Arabic, A formal definition of the Arabic entries of multilingual lexical databases. ACL/EACL 2001 Workshop on Arabic NLP, Toulouse, France.
- Dichy, J. and Farghaly, A. (2003). Roots & patterns vs. stems plus grammar-lexis specifications: on what basis should a multilingual database centred on Arabic be built? In *Proceedings of the MT Summit IX Workshop: Machine Translation for Semitic Languages*. New Orleans. 1-8.
- Dowty, D. R. (1979). Word meaning and Montague grammar: The semantics of verbs and times in generative semantics and in Montague's PTQ. Amsterdam: Springer.
- El-Affendi, M. (1991). An algebraic algorithm for Arabic morphological analysis. *Arabian journal for science and engineering*, 16(4), 605-611.
- El-Affendi, M.A. (1999). Building an Arabic distributed collaboration environment. The final report. *Research project: AR-16-094*, King Abdulaziz City for Science and Technology.

Fillmore, C. J. (1968). The case for case. In *Universals in Linguistic Theory*. ed. Bach, E. & Harms, R. New York: Holt, Rinehart and Winston. 1-88.

- Fillmore, C. J. (1971). Some problems for case grammar. Report of the Twenty-Second Annual Round Table Meeting on Linguistics and Language Studies. ed. Richard J. O'Brien, S.J. Washington: Georgetown University Press. 35-56.
- Finegan, E. (2008). Language: Its structure and use. Boston: Thompson & Heinle.
- Fromkin, V., Rodman, R., & Hyams, N. (2011). *An introduction to language* (9th ed.). Wadsworth: Cengage Learning.

- Gafos, A. I. (1999). The articulatory basis of locality in phonology. New York: Garland.
- Goldsmith, J. (1976). An overview of autosegmental phonology. *Linguistic analysis*, 2, 23-68.
- Gridach, M., & Chenfour, N. (2011). Developing a new system for Arabic morphological analysis and generation. In Proceedings of the 2nd Workshop on South and Southeast Asian Natural Language Processing (WSSANLP). 52-57.
- Habash, N. (2004). Large scale lexeme based Arabic morphological generation. *JEPTALN* 2004, Session Traitement Automatique de l'Arabe, Fès. 271-276.
- Habash, N. (2010). Introduction to Arabic natural language processing. Morgan & Claypool Publishers. www.morganclaypool.com.
- Hasan, A. (1969). An-Nahw Al-Wāfī. Cairo: Dar al-Ma^cārif.
- Hayes, B. (1995). *Metrical stress theory: principles and case studies*. Chicago: University of Chicago Press.
- Heath, J. (1987). *Ablaut and ambiguity: Phonology of a Moroccan Arabic dialect*. Albany, New York: State University of New York.
- Helbig, H. (2006). *Knowledge representation and the semantics of natural language*. New York: Springer.
- Hetzron, R. (1992). Semitic languages. ed. Bright, W. International Encyclopedia of Linguistics, Vol 3. Oxford: Oxford University Press. 412-417.
- Ibn Jinnī, A. (1956). Al-Xaşā³iş. Cairo: Dār al-Kutub al-Maşriyyah.
- Ibn Jinnī, A. (1960). Al-Munșif. ed. by Ibarhem Mustafa. Cairo: Mustafa al-Bābī al-Halabī.
- Jayan, J. P., Rajeev, R. R., & Rajendran, D. S. (2011). Morphological analyser and morphological generator for Malayalam-Tamil machine translation. *International Journal of Compter Applications*, 13(8), 15-18.
- Kager, R., van der Hulst, H., & Zonneveld, W. (Eds.). (1999). *The prosody-morphology interface*, Vol. 79. Cambridge: Cambridge University Press.

Katamba, F. (1993). Morphology. New York: St Martin's Press.

- Kenny, A. (1963). Action, emotion and will. London: Routledge & Kegan Paul.
- Khoja, S. (1999). Stemming Arabic text. Computing Department, Lancaster University. http://www.comp.lancs.ac.uk/computing/users/khoja/stemmer.ps
- Kiraz, G. A. (1994). Multi-tape two-level morphology: A case study in Semitic non-linear morphology. In *Proceedings of the 15th Conference on Computational Linguistics*, Vol 1. Stroudsburg: Association for Computational Linguistics, 180-186.
- Kiraz, G. A. (2001). Computational nonlinear morphology with emphasis on Semitic languages. Cambridge: Cambridge University Press.
- Koskenniemi, K. (1983). Two-level morphology: A general computational model for word-form recognition and production. *Publication 11*, Department of General Linguistics, University of Helsinki.
- Levin, B. (1993). English verb classes and alternations: A preliminary investigation. Chicago: University of Chicago Press.
- McCarthy, J. J. (1979). Formal problems in Semitic phonology and morphology. Doctoral dissertation, MIT.
- McCarthy, J. J. (1981). A prosodic theory of nonconcatenative morphology. *Linguistic Inquiry*, 12, 373-418.
- McCarthy, J. J. (1992). Template form in prosodic morphology. In Papers from the Third Annual Meeting of the Formal Linguistics Society of Midamerica. ed. Stvan, L. S. et al. Bloomington: Indiana University Linguistics Club. 187-218.
- McCarthy, J. J. (1993). A case of surface constraint violation. *Canadian Journal of linguistics*, 38(2), 169-195.
- McCarthy, J. J., & Prince, A. (1990a). Foot and word in prosodic morphology: The Arabic broken plural. *Natural Language & Linguistic Theory*, 8(2), 209-283.

- McCarthy, J. J., & Prince, A. (1990b). Prosodic morphology and templatic morphology. In Perspectives on Arabic linguistics II: Papers from the second annual symposium on Arabic linguistics. Amsterdam: John Benjamins. 1-54.
- McCarthy, J. J., & Prince, A. (1993). *Prosodic morphology I: Constraint interaction and satisfaction*. Ms., University of Massachusetts, Amherst, and Rutgers University.
- McCarthy, J. J., & Prince, A. (1995). Faithfulness and reduplicative identity. In *Papers in Optimality Theory*. ed. Beckman, J. Urbanczyk, S. & Walsh, L. Amherst MA: University of Massachusetts. 249-384.
- McCarthy, J. J., & Prince, A. (1996). *Prosodic morphology*. Ms., University of Massachusetts, Amherst, and Brandeis University, Waltham, Mass.
- McCarthy, J. J., & Prince, A. (1999). Faithfulness and identity in prosodic morphology. *The prosody-morphology interface*, ed. Kager, R., Hulst, H. & Zonneveld, W. Cambridge: Cambridge University Press. 218-309.
- McOmber, M. L. (1995). Morpheme edges and Arabic infixation. In Perspectives on Arabic Linguistics VII, ed. Eid, M. Amsterdam/Philadelphia: John Benjamins. 173-189.
- Pawlak, M., & Bielak, J. (2011). New perspectives in language, discourse and translation studies. Amsterdam: Springer.
- Piñeros, C. E. (1998). Prosodic morphology in Spanish: Constraint interaction in word formation. Doctoral dissertation. Columbus: Ohio State University.
- Plag, I. (2002). The role of selectional restrictions, phonotactics and parsing in constraining suffix ordering in English. In *Yearbook of Morphology 2001*, Amsterdam: Springer. 285-314
- Prunet, J. F., Béland, R., & Idrissi, A. (2000). The mental representation of Semitic words. Linguistic Inquiry, 31(4). 609-648.
- Ratcliffe, R. R. (1998). *The broken plural problem in Arabic and comparative Semitic: allomorphy and analogy in non-concatenative morphology*. Amsterdam/Philadelphia: John Benjamins.

- Ryding, K. C. (2005). *A reference grammar of Modern Standard Arabic*. Cambridge: Cambridge University Press.
- Ryle, G. (1949). The concept of mind. Chicago: University of Chicago Press.
- Sawalha, M. (2011). Open-source resources and standards for Arabic word structure analysis. PhD Thesis. Leeds: University of Leeds.
- Selkirk, E. (1980). Prosodic domains in phonology: Sanskrit revisited. *Juncture*, 7, 107-129.
- Sībawayhi, A. (1966). Al-Kitab. ed. Abd al-Salām Hārūn. Cairo: Dār al-Qalam.
- Smith, C. S. (1991). The parameter of aspect. Boston: Kluwer Academic.
- Smrž, O. (2007). Functional Arabic morphology: Formal system and implementation. PhD thesis. Prague: Charles University in Prague.
- Spencer, A. (1991). *Morphological theory: An introduction to word structure in generative grammar*. Oxford: Basil Blackwell.
- Thalouth, B., & Al-Dannan, A. (1990). A comprehensive Arabic morphological analyser generator. In *Proceedings Computers and the Arabic Language*. ed. MacKay, P. A. New York: Hemisphere Publishing Corporation. 208-217.
- Ussishkin, A. (1999). The inadequacy of the consonantal root: Modern Hebrew denominal verbs and output-output correspondence. *Phonology*, 16(3), 401-442.
- Ussishkin, A. (2003). Templatic effects as fixed prosody: The verbal system in Semitic. *Amsterdam Studies in the Theory and History of Linguistic Science Series*, 4, 511-530.
- Ussishkin, A. (2005). A Fixed prosodic theory of nonconcatenative templatic morphology. *Natural Language & Linguistic Theory*, 23(1), 169-218.
- Ussishkin, A. P. (2000). The emergence of fixed prosody. Doctoral dissertation. Santa Cruz: University of California.

Vendler, Z. (1968). Adjectives and nominalizations. Berlin: Walter De Gruyter.

- Versteegh, K. and Versteegh, C. H. M. (1997). *The Arabic language*. New York: Columbia University Press.
- Watson, J. C. E. (2002). *The phonology and morphology of Arabic*. Oxford: Oxford University Press.
- Watson, J. C. E. (2006a). Arabic morphology: diminutive verbs and diminutive nouns in San'āni Arabic. *Morphology*, 16(2), 189-204.
- Watson, J. C. E. (2006b). Arabic as an introflecting language. ed. Keith Brown. In *The Encyclopaedia of Language and Linguistics*. Amesterdam: Elsevier, 2nd edition. 431-434.
- Yagi, S & Harous, S. (2003). Arabic morphology: an algorithm and statistics. In Proceedings of the 2003 International Conference on Artificial Intelligence (IC-AI 2003), Las Vegas, Nevada.